DYSPNEA AND THE MECHANICS OF BREATHING DURING PROGRESSIVE EXERCISE

SUSAN P. BURKE

A thesis submitted to the Faculty of Graduate Studies and Research in partial fulfilment of the requirements for the degree of Master of Science

> Department of Physiology and Meakins-Christie Laboratories McGill University Montreal, Quebec, Canada February, 1993

© Susan P. Burke 1993

ABSTRACT

This study investigates dyspnea and the mechanics of breathing during progressive exercise. Three subject groups, athletes, normal sedentary subjects and chronic obstructive diseased patients were studied during progressive exercise testing to exhaustion on a cycle ergometer. Subjects rated dyspnea on a Borg Scale. Inspiratory flow, esophageal/gastric pressures and rib cage/abdominal displacements were measured.

Subjects demonstrated two patterns of dyspnearesponse to changes in esophageal (pleural) pressure All athletes, two normals and five patients were termed "low dyspnea responders", (LDR), whereas the remaining subjects were termed "high dyspnea responders", (HDR)

LDR demonstrated large, rapid negative gastric pressure swings, coupled with outward abdominal displacement during early inspiration when compared to HDR, suggesting that LDR utilized abdominal muscle relaxation at the onset of inspiration. This mechanism appears to provide an extra inspiratory force contributing to the increasing pleural pressures required. This breathing pattern appears to diminish the sensation of dyspnea at a given pleural pressure.

RESUME

Cette étude examine la dyspana et les mécaniques de la respiration durant l'exercice progressif. Trois grouper et lont les athlètes, les normaux sédentaires, ainsi que des patients avec mana a puis de le atructive chronique ont été etudies durant l'exercice progressif maximation durant de la dysphée. Débit respiration durant la cage thoracique/abdominale ont ausside et surés

Les sujets ont démontré deux types de réponses dysprieiques aux changements de pression pescohagienne. Tous les athlètes ainsi que deux normaux et cinq patients étaient classés comme repondants avec tres peu de dyspriée (RPD), tandis que les sujets restants étaient classés comme répondants avec dyspriée importante (RDI)

RPD ont némontré de grands et rapides changements d'amplitude en direction négative de pression gastrique ainsi qu'une augmentation de la circonférence abdominale durant la première partie de l'inspiration comparé aux RDI, suggérant un relâchement des muscles abdominaux dans le groupe RPD au début de l'inspiration. Ce mecanisme semble apporter un ajout à la force inspiratoire contribuant ainsi à augmenter la pression pleurale requise. Ce patron respiratoire semble diminuer la sensation de dyspriée a tous les niveaux de la pression pleurale

ACKNOWLEDGEMENTS

There are several people I wish to thank, without whose help this thesis would not have been possible. First, I would like to thank my supervisor, Dr. Alex Grassino for his support and guidance throughout the winding road I took in order to complete this study. His patience and help during my full-time Master of Business Administration studies ultimately resulted in the completion of this thesis. Also I would like to thank Dr. Grassino for "encouraging" me to speak at conferences and beer seminars. It was great experience for my present career in the pharmaceutical industry.

Lam indebted to Dr. Paul Begin and Christy Wilson, whose help and attendance during the exercise tests was invaluable. Lalso must thank all of the researchers in our lab at Notre Dame hospital, whose support allowed me to maintain my sanity. Norm Comtois, Christer Sinderby, Adriana Spahija and Doug Wight. Liwould also like to thank Jennifer Beck for her running around and help getting this thesis finished while Liwas working full time.

A special thanks also to Adriana, Jennifer and Diane for the translation of the abstract

I also would like to thank my roommate Penny for putting up with me during the last several horrendous months

Thanks to everyone at the Meakins-Christie Laboratories and Notre Dame Hospital who made the last many years both memorable and rewarding

I also want to thank my friends who volunteered as subjects and who endured the dreaded esophageal and gastric balloon catheters – Grant Gush, Richard Young, Steve Gregory and Dave Wilson

Finally, I would like to thank three people, without whose support and shoulders to lean on this work would not have been possible: my parents and Dave.

TABLE OF CONTENTS

ABSTRACT	I
RESUME	11
ACKNOWLEDGEMENTS	\$11
TABLE OF CONTENTS	IV
LIST OF FIGURES	XI
ABBREVIATIONS	XIV
CHAPTER 1 INTRODUCTION	1
CHAPTER 2 FUNCTIONAL ANATOMY OF THE RESPIRATORY MUSCLES 2.1 The Rib Cage	3 3 6 6 7 8
CHAPTER 3 PATHOPHYSIOLOGY OF CHRONIC OBSTRUCTIVE PULMONARY	

DISEASE		• •	 		11
3.1 Symptoms and Etiolog	y of COPD				11

	v
3.2 Mechanics of Breathing in COPD	13
CHAPTER 4 VENTILATION DURING EXERCISE	16
4.1 Ventilatory Responses to Exercise in Normal Subjects	16
4.2 Ventilatory Responses to Exercise in COPD Patients	18
4.3 Limitations of Exercise	19
CHAPTER 5 MECHANICS OF BREATHING AND RESPIRATORY MUSCLE	
COORDINATION	21
51 Historical Development	21
5.2 Analysis of Rib Cage and Abdominal Displacement	22
5 2.1 Respiratory Inductive Plethysmography (RIP)	24
5.3 Methods to Infer Respiratory Muscle Activity-The Analysis of Pressure	
Changes	25
531 The Macklem Diagram	26
CHAPTER 6 RESPIRATORY PSYCHOPHYSICS	29
6 1 Historical Development	29
6 2 Scaling Methods	30
6.2.1 Indirect Scaling	30
6 2 2 Direct Scaling	30
6 2 2 1 Ratio Scaling	30
6 2 2 1 1 Magnitude Estimation	31
6 2 2.1 2 Magnitude Production	31
6 2 2.1 3 Cross-modality Matching	31
6 2 2.2 Category Scaling	31
6 2 2.3 Visual Analogue Scale (VAS)	32
6 2 2.4 Category-Ratio Scaling-The BORG Scale	32
6.3 Respiratory Psychophysics	34
6 3.1 Threshold Load Detection in Normal Subjects	34

	COPD	
6.3.3	Mechanisms of Threshold Load Detection	
6.3.4	Threshold Detection of Ventilation and Pressure	
6.3 5	Direct Scaling of Perceptions of Volume, Ventilation,	
	Frequency and Pressure	
6.3.6	The Perception of Achieved Force (Tension) and Pressure	
6.37	Magnitude Estimation of Loads in COPD Patients	
6.3 8	Mechanisms of the Perceived Magnitude of Volume and	
	Added Loads	
639	Effects of Background Loading on Magnitude Estimation of	
	Resistive Loads	
6.3 1	0 Perception of Force Versus Perception of Effort .	
7 1 Methods	s Used to Quantify Dyspnea	
7.2 Psychor	physical Testing and Dyspnea	
7.2 Psychor 7.2 1	bhysical Testing and Dysphea	
7.2 Psychol 7.2 1	Dysphea During Hypercaphic and Hypoxic Ventilatory Stimulation	
7.2 Psychol 7.2 1 7.2 2	Dyspical Testing and Dyspica Dyspical During Hypercaphic and Hypoxic Ventilatory Stimulation The Role of Vagal Afferents in the Perception of	
7.2 Psychoj 7.2 1 7.2.2	Dyspical Testing and Dyspica Dyspical During Hypercaphic and Hypoxic Ventilatory Stimulation The Role of Vagal Afferents in the Perception of Breathlessness	
7.2 Psychoj 7.2 1 7.2.2 7.2.3	Dyspnea Dyspnea During Hypercaphic and Hypoxic Ventilatory Stimulation Stimulation The Role of Vagal Afferents in the Perception of Breathlessness Exercise Testing and Dyspnea	
7.2 Psychoj 7.2 1 7.2.2 7.2.2	Dyspnea During Hypercaphic and Hypoxic Ventilatory Stimulation The Role of Vagal Afferents in the Perception of Breathlessness Exercise Testing and Dyspnea 7 2 3 1 Dyspnea as a Euroction of Workload	
7.2 Psychoj 7.2 1 7.2.2 7.2.3	Dyspnea During Hypercaphic and Hypoxic Ventilatory Stimulation The Role of Vagal Afferents in the Perception of Breathlessness Exercise Testing and Dyspnea 7 2 3 1 Dyspnea as a Function of Workload 7.2.3 2 Dyspnea as a Function of Ventilation	
7.2 Psychoj 7.2 1 7.2.2 7.2.3	Dyspnea During Hypercaphic and Hypoxic Ventilatory Stimulation The Role of Vagal Afferents in the Perception of Breathlessness Exercise Testing and Dyspnea 7 2 3 1 Dyspnea as a Function of Workload 7.2.3 2 Dyspnea as a Function of Ventilation 7 2 3.3 The Effects of External Loads on Dyspnea	
7.2 Psychoj 7.2 1 7.2.2 7.2.3	 bysical Testing and Dyspnea Dyspnea During Hypercaphic and Hypoxic Ventilatory Stimulation The Role of Vagal Afferents in the Perception of Breathlessness Exercise Testing and Dyspnea 7 2 3 1 Dyspnea as a Function of Workload 7.2.3 2 Dyspnea as a Function of Ventilation 7 2 3.3 The Effects of External Loads on Dyspnea Perception 	
7.2 Psychoj 7.2 1 7.2.2 7.2.3	 Dyspnea During Hypercaphic and Hypoxic Ventilatory Stimulation The Role of Vagal Afferents in the Perception of Breathlessness Exercise Testing and Dyspnea 7 2 3 1 Dyspnea as a Function of Workload 7.2.3 2 Dyspnea as a Function of Ventilation 7 2 3.3 The Effects of External Loads on Dyspnea Perception 7 2 3.4 Dyspnea as a Function of Pressure Development 	
7.2 Psychoj 7.2 1 7.2.2 7.2.3	 byspical Testing and Dyspica Dyspical During Hypercaphic and Hypoxic Ventilatory Stimulation The Role of Vagal Afferents in the Perception of Breathlessness Exercise Testing and Dyspica 7 2 3 1 Dyspica as a Function of Workload 7.2.3 2 Dyspica as a Function of Ventilation 7 2 3.3 The Effects of External Loads on Dyspica Perception 7 2 3.4 Dyspica as a Function of Pressure Development 7.2.3.5 Dyspica as a Function of other Physiological 	
7.2 Psychoj 7.2 1 7.2.2 7.2.3	 bysical Testing and Dyspnea Dyspnea During Hypercaphic and Hypoxic Ventilatory Stimulation The Role of Vagal Afferents in the Perception of Breathlessness Exercise Testing and Dyspnea 7 2 3 1 Dyspnea as a Function of Workload 7.2.3 2 Dyspnea as a Function of Ventilation 7 2 3.3 The Effects of External Loads on Dyspnea Perception 7 2 3.4 Dyspnea as a Function of Pressure Development 7.2.3.5 Dyspnea as a Function of other Physiological Parameters 	

VI

7.4 Mechanisms of Augmentation and Relief of Dysphea In COPD	
Subjects	51
CHAPTER 8 METHODOLOGY	52
8 1 Computer Analysis	57
CHAPTER9 RESULTS	62
91 Anthropometrics	62
9 1 1 Pulmonary Function	62
9 1 2 Respiratory Muscle Strength	62
9.2 The Relationship of Dyspnea to Work	65
9 2 1 Dysphealas a Function of Workload (Watts)	65
9 2 2 Dysphea as a Function of Workload (% Maximum)	68
93 The Relationship of Minute Ventilation and Work	68
9.3.1 Minute Ventilation as a Function of Work	68
9 3 2 Normalized Minute Ventilation as a Function of Work	69
9.3.3 Normalized Minute Ventilation as a Function of Workload (%	
Maximum)	69
94 Dysphea as a Function of Minute Ventilation	72
9.5 Minute Ventilation Normalized as a Percentage of Reserve Capacity	7 3
9 5 1 VE/MVV as a Function of Work	73
9.5.2 VE/MVV as a Function of Workload (% maximum)	73
9.6 Dyspnea as a Function of VE/MVV	76
9.7 The Relationship of Dyspnea to Pleural Pressure Changes	76
9 7.1 ΔPes Normalized by Maximal Inspiratory Pressure (MIP)	76
9711 Dyspnea Response to $\Delta Pes/MIP$ of Athletes	77
9712 Dyspnea Response to $\Delta Pes/MIP$ of Normals	77
9713 Dyspnea Response to APes/MIP of Patients	78
97.2 ΔPes Normalized by the Maximum Pressure Generating	
Capacity of the Inspiratory Muscles during Exercise .	78

			VIII
	9 7.2.1	Dyspnea Response to APes/Peap of Athletes	79
	972.2	Dyspnea Response to APes/Pcap of Normals	79
	972.3	Dyspnea Response to APes/Pcap of Patients	81
973	Validity o	f Regression Analysis in Predicted Maximum	
	Achieved	d ΔPes/Pcap Values	81
98 Analysis	of "Pre-In	spiratory Flow" Pressure Generation	83
9 8.1	Dyspnea	a versus EEP	83
	9 8.1 1	Dyspnea Response to EEP of Athletes	86
	9812	Dyspnea Response to EEP in Normals	86
	9813	Dyspnea Response to EEP in Patients	86
9 8.2	EEP vers	sus (ΔPes/Δt) _{PRt}	87
	9 8.2.1	The Relationship of EEP to $\Delta Pes/\Delta t_{PRI}$	87
9.8.3	Dyspnea	versus $\Delta Pga/\Delta t_{PRI}$	88
	9.8 3.1	Dyspheal Response to $\Delta Pga/\Delta t_{PRF}$ of Athletes	88
	9 8.3.2	Dyspnea Response to $\Delta Pga/\Delta t_{PRE}$ of Normals	88
	9 8.3.3	Dyspnea Response to $\Delta Pga/\Delta t_{PRE}$ of Patients	89
9 8.4	The Rela	ationship of $\Delta Pes/\Delta t_{pRF}$ to $\Delta Pga/\Delta t_{pRF}$	89
	9.841	$\Delta Pes/\Delta t_{PR}$ to $\Delta Pga/\Delta t_{PR}$ of Athletes	89
	9842	$\Delta Pes/\Delta t_{PRE}$ to $\Delta Pga/\Delta t_{PRE}$ of Normals	89
	9843	$\Delta Pes/\Delta t_{PRT}$ to $\Delta Pga/\Delta t_{PPT}$ of Patients	90
99 Analysis	of the Ind	tial Phase of Pressure Generation (from 0 to Peak	
Inspirate	o ry Flow A	cceleration)	90
9 9.1	Dyspnea	versus $\Delta Pes/P_{cap}$	90
992	$\Delta \text{Pes}/\Delta t$	versus ∆ Pes/Pcap	90
	9.92.1	The $\Delta Pes/\Delta t$ to $\Delta Pes/Pcap Relationship in Athletes$	93
	9922	The $\Delta Pes/\Delta t$ to $\Delta Pes/Pcap Relationship in Normals$	93
	9 9.2 3	The $\Delta Pes/\Delta t$ to $\Delta Pes/Pcap Relationship in Patients$	93
9.9.3	Dyspnea	$\Delta \text{Pga}/\Delta t$.	93
	9 9.3.1	Dyspnea Response to ∆Pga/∆t of Athletes	94
	9.9.3 2	Dyspnea Response to ΔPga/ΔI of Normals	94

٦

	ix
9933 Dyspnea Response to $\Delta Pga/\Delta t$ of Patients	94
9.9.4 The Relationship of $\Delta Pes/\Delta t$ to $\Delta Pga/\Delta t$	95
9.9.4.1 $\Delta Pes/\Delta t$ to $\Delta Pga/\Delta t$ Relationship of Athletes	95
9942 $\triangle Pes/\Delta t$ to $\triangle Pga/\Delta t$ Relationship of Normals	95
9943 $\Delta Pes/\Delta t$ to $\Delta Pga/\Delta t$ Relationship of Patients	95
9 10 The Pressure-Time Index of the Inspiratory Muscles	96
9 10 1 Pressure Duty Cycle (Ti/Ttot)	96
9102 ΔPes/Pcap	96
9 10.3 TTIM	99
9 11 The Pressure-Time Index of the Diaphragm	99
9 11 1 Pressure Duty Cycle (Ti/Ttot)	99
9112 ΔPdi/Pdi _{max}	100
9 11 3 TTdi	100
9 12 The Relationship of Dyspnea to TTim and TTdi	101
9 12 1 Dyspnea versus TTim	101
9 12 1 1 Δ Dyspnea Response to Δ TTim of Athletes	101
9 12 1 2 Δ Dyspnea Response to Δ TTim of Normals	101
9 12 1 3 Δ Dyspnea Response to Δ TTim of Patienus	101
9 12 2 Dyspnea versus TTdı	103
9 12 2 1 Δ Dyspnea Response to Δ TTdi of Athletes	103
9 12 2 2 Δ Dyspnea Response to Δ TTdi of Normals	103
9.12.2.3 Δ Dyspnea Response to Δ TTdi of Patients	103
9.13 Comparison of the Δ Dyspnea Response to Δ TTim versus Δ TTdi	103
9 14 The Mechanics of Breathing A Qualitative Analysis	104
9 14 1 Mechanics of Breathing of Athletes versus Normal	
Subjects	106
9 14.1 1 Low Responding Athletes versus High Responding	
Normals	106
9.14.1.2 Low Responding Athlete versus Low Responding	
Normal	109

9 14.2 Mechanics of Breathing in COPD Patients	112
9.14 2 1 Low Responding Patients versus High Responding	
Patients	112
CHAPTER 10 DISCUSSION	116
10.1 Limitations of Study Design	116
10.2 The Perception of Dyspnea during Exercise	116
10.2 1 Dyspnea versus Work .	118
10.2 2 Dyspnea versus Ventilation	119
10 2.3 Dyspnea versus Pleural Pressure	120
10.3 Factors Limiting Exercise	121
10 3 1 Ventilation	121
10 3 2 The Inspiratory Muscles	122
10.4 Mechanisms of Dyspnea	123
10 4.1 The Relationship of Dyspnea to Pressure Changes	123
10 4 2 Dyspnea and the Mechanics of	
Breathing	125
10.7 CONCLUSIONS	135
SUGGESTIONS FOR FUTURE RESEARCH	136
REFERENCES	. 137

х

LIST OF FIGURES

Figure

Page

21	The respiratory muscles and their respective actions
31	Effects of hyperinflation on the respiratory system
51	The Konno-Mead diagram
52	The Macklem diagram
6 1	Modified borg Scale
81	The experimental setup
82a 82b	Method of calculating the ventilatory Ti,Te,Ttot 58 Method of calculating the pressure related Ti,Te,Ttot 58
83	The Pre-inspiratory and Initial Phases of pressure
	generation
91a 91b 91c	Dyspnea plotted as a function of workload66Dyspnea plotted as a function of normalized workload66Common logarithm of dyspnea plotted as a function of
	normalized workload
92a	Minute ventilation (VE) plotted as a function of workload
92b 92c 92d	Minute ventilation, normalized by Forced Vital Capacity (FVC)plotted as a function of workload
93a	VE normalized by the Maximum Voluntary Ventilation (MVV),
93b 93c	plotted as a function of workload 74 VE/MVV plotted as a function of normalized workload 74 Common logarithm of VE/MVV plotted as a function of 6
93d	normalized workload 75 Dyspnea plotted as a function of VE/MVV
94a	Dyspnea plotted as a function of pleural pressure (ΔPes), normalized by the maximum inspiratory pressure (MIP)
94b	Dyspnea plotted as a function of ΔPes , normalized by the maximum capacity of the inspiratory muscles to generate

pressure, corrected for inspiratory flow and volume (Pcap)	80
9.5 Validity of regression analysis in predicting the maximum achieved $\Delta Pes/P_{cap}$	82
9.6 Pre-inspiratory flow, four quadrant graph	84
9.7 Initial phase of pressure generation, four quadrant graph .	91
9.8.a Duty cycle (Ti/Ttot) plotted versus $\Delta Pes/P_{cap}$	98
9.8 b Duty cycle (Ti/Ttot) plotted versus $\Delta PdT/PdI_{max}$.	98
9.9 Dyspnea plotted as a function of the pressure-time index of the inspiratory muscles (TTim) and the diaphragm (TTdi)	102
9.10.a Macklem diagram for low responding athlete 9.10 b Inspiratory flow, Pes and Pga plotted versus time for low	107
responding athlete	107
9.10.c Konno-Mead diagram for low responding athlete	108
responding athlete	108
9.11.a Macklem diagram for high responding normal .	107
9.11.b Inspiratory flow, Pes and Pga plotted versus time for high	
responding normal	107
9.11.c Konno-Mead diagram for high responding normal	108
9 11.d Rib cage and abdominal displacement versus time for high	
responding normal	108
9.12.a Macklem diagram for low responding athlete	110
responding athlete	110
9.12.c Konno-Mead diagram for low responding athlete	111
responding athlete	111
9 13.a Macklem diagram for low responding normal	110
9.13 b Inspiratory flow, Pes and Pga plotted versus time for low	440
responding normal	110
9.13 d Rib cage and abdominal displacement versus time for low	
responding normal	111
9.14 a Macklem diagram for low responding patient	113
9.14.b Inspiratory flow, Pes and Pga plotted versus time for low	

XII

	responding patient	113
9 14 c	Konno-Mead diagram for low responding patient	114
9 14 d	Rib cage and abdominal displacement versus time for low	
	responding patient	114
9 15 a 9 15 b	Macklem diagram for high responding patient	113
	responding patient	113
9 15 c	Konno-Mead diagram for high responding patient	114
9 15 d	Rib cage and abdominal displacement versus time for high	
	responding patient	114
10 1 a	Inspiratory mechanics using intercostal &/or accessory/	
	scalene muscie	126
10 1 b	Inspiratory mechanics using passive relaxation of	
	triangularis sterni	126
10 1 c	Inspiratory mechanics during passive abdominal muscle	
	relaxation	126
10 1 d	Inspiratory mechanics using passive abdominal muscle	
	relaxation coupled with diaphragmatic contraction	126

Tables

71	Definitions of Dyspnea Proposed by Various Authors	42
91	Patient data	63
92	Normal and athletic subjects' data	64
93	Pre-inspiratory flow values at quiet breathing and maximum exercise of Dyspnea, EEP, $\Delta Pes/\Delta t_{PRE}$, $\Delta Pga/\Delta t_{PRE}$	85
94	Values during the initial phase of pressure generation at quiet breathing and maximum exercise, Dyspnea, $\Delta Pes/Pcap$, $\Delta Pes/\Delta t$, $\Delta Pga/\Delta t$	92
95	Symbols utilized for values at quiet breathing and	
	maximum exercise for individual subjects	97
96	Definitions of symbols used in the qualitative analysis 1	05

ABBREVIATIONS

AB	Abdomen
A-P	Anterior-posterior
cmH20	centimeters of water
COPD	Chronic obstructive pulmonary disease
EEP	End expiratory pleural pressure
fb	Frequency of breathing
FEV ₁	Forced expired volume in one second
FRC	Functional residual capacity
FVC	Forced vital capacity
HDR	High Dyspnea Responder
IMF	Inspiratory muscle fatigue
1	htre
l/min	litre/minute
l/sec	litre/sec
LDR	Low Dyspnea Responder
min	minute
MIP	Maximum inspiratory pressure
MSV	Maximum sustainable ventilation
MVV	Maximal voluntary ventilation
Pab	Abdominal pressure
Pcap	Maximum capacity for pleural pressure generation, adjusted for inspiratory flow and volume
Pdi	Transdiaphragmatic pressure
Pdi	Mean transdiaphragmatic pressure
Pdi _{max}	Maximum transdiaphragmatic pressure at functional residual capacity with airways closed
$\Delta Pdi/Pdi_{max}$	Mean Pdi swing as a fraction of maximum
Pes	Esophageal pressure
∆Pes	Peak to peak esophageal pressure swing
∆Pes/MIP	Peak to peak esophageal prossure swing as a fraction of the maximum inspiratory pressure generated at FRC
∆Pes/Pcap	Peak to peak esophageal pressure swing as a fraction of the maximum capacity to generate pleural pressure
Pes	Mean esophageal pressure swing
Pes _{max}	Maximum pleural pressure at functional residual capacity with airways closed

ΧIV

∆Pes/Pcap	Mean esophageal pressure expressed as a fraction of the maximum capacity to generate pleural pressure
PFT	Pulmonary function test
Pga	Gastric pressure
∆Pga	Peak to peak gastric pressure swing
ΡρΙ	Pteural pressure
PTI	Pressure-time index
RC	Rib cage
RMF	Respiratory muscle fatigue
RV	Residual volume
sec	second
Ті	Inspiratory time
Те	Expiratory time
T _{tot}	Total time of one breathing cycle
TLC	Total lung capacity
TTdi	Pressure-time index of the diaphragm
TTim	Pressure-time index of the inspiratory muscles
V	Inspiratory flow
VAS	Visual Analogue Scale
VC	Vital capacity
VE	Minute ventilation
VE/FVC	Minute ventilation normalized as a fraction of the Forced Vital Capacity
Ÿe/MVV	Minute ventilation normalized as a fraction of the Maximum Voluntary Ventilation
VAB	Abdominal volume
VRC	Rib cage volume
Vt	Tidal volume
Vt/Ti	Mean inspiratory flow
Vt/Te	Mean expiratory flow
W	Watt

CHAPTER 1 INTRODUCTION

Dyspnea may be defined as breathlessness, difficult or laboured breathing and is a psychophysical sensation which becomes a major debilitating factor for patients suffering from Chronic Obstructive Pulmonary Disease (COPD) Many proposals have been postulated concerning the physiological and psychological causes of dyspnea, however no precise mechanism of dyspnea has been identified to date

Presumably, dyspnea is a sensation resulting from a number of inter-related stimuli For example, the diseased state of the respiratory system in COPD patients may dramatically effect perceived dyspnea. The COPD patient may experience overwhelming and exercise limiting dyspnea upon even mild exertion, whereas in normal subjects, exercise is rarely limited solely by dyspnea. Only upon extreme, strenuous exercise may dyspnea become a limiting factor for normal subjects

The following presents a summary of the contents of this thesis. A relevant literature review is presented in Chapters 2 to 7 inclusively.

Chapter 2 will addr_ss the functional anatomy of the respiratory muscles

Chapter 3 discusses Chronic Obstructive Pulmonary Disease and the mechanical limitations it places upon the respiratory system

Chapter 4 will address ventilation during exercise in both normal/athletic subjects and COPD patients, and the mechanics of breathing during exercise

Chapter 5 examines respiratory muscle coordination and the mechanics of breathing and reviews the various methodologies existing for both the quantification and qualification of respiratory muscle activity

Chapter 6 will discuss the scientific field of psychophysics Following a brief historical review, the application of psychophysics to studies of respiration will be discussed

Chapter 7 addresses the specific respiratory perception of dyspnea and examines studies specifically addressing possible stimuli of this sensation

Chapter 8 will discuss the methodology of this study whose first objective was to investigate the relationship between dyspnea and work, ventilation and pressure changes in three distinct subject groups, elite athletes, normal subjects and COPD patients

The second objective of the present study was to examine muscular coordination and respiratory mechanics during progressive exercise testing to determine if a relationship exists to subject's perceived dyspnea

Chapter 9 presents the results of this study.

Chapter 10 will discuss and interpret the results and discuss future areas for investigation

This study attempts to both qualify and quantify parameters contributing to a subjects' subjective sensation of dyspnea. Clinically, a better understanding of possible mechanisms of dyspnea is essential for appropriate patient management.

CHAPTER 2 FUNCTIONAL ANATOMY OF THE RESPIRATORY MUSCLES

The thrust of this thesis is on respiratory muscle interaction in health and disease This description of the structure and function of the respiratory muscles provides the basis for future reference

All muscles comprising the respiratory system are skeletal (as any limb muscle), whose primary task is to displace the chest wall, resulting in ventilation of the lungs Functionally, the chest wall can be divided into two compartments mechanically arranged in parallel, the rib cage (RC) and the abdomen (AB), separated from each other by a thin musculotendinous structure, the diaphragm [1] Given their arrangement, an increase in lung volume can be achieved by either an increase in RC or AB volume [1,2] [fig 2.1]

2.1 The Rib Cage

Structurally, the RC consists of the thoracic vertebrae dorsally, upon which the bony ribs articulate with the transverse processes and vertebral bodies. The skeletal arches are completed via costal cartilage and articulations joining the bony ribs to the sternum ventrally. The ribs' movement is largely monoaxial (rotation about the axis defined by the ribs' articulations with the vertebral bodies and the transverse processes [3,4]), however, given their configuration, rib movement can also occur on a sagittal, frontal or axial plane. Mechanically, any muscles which elevate the ribs will provide an inspiratory force, while muscles that lower the ribs will have an expiratory effect [1]. The horizontal orientation of the upper ribs as well as the more restrictive sternal attachments result in a "pump handle" motion upon inspiration, and movement as a unit with the sternum [5,6]. In contrast, the lower ribs enjoy greater freedom of movement during inspiration, moving both cranially and faterally in the so called "bucket-handle" pattern.

2.2 The Diaphragm

The diaphragmatic fibers radiate from a central tendon to insert peripherally as either the crural (vertebral) or costal portions. The diaphragm has been described as an elliptical cylindroid, capped by a dome (the central tendon) **[1,2,7]**. The cylindrical portion of the



Figure 2.1 Diagrammatic representation of inspiratory and expiratory muscles and their respective actions (arrows)

diaphragm is opposed to the inner aspects of the lower RC, forming the "zone of apposition" **[8,9]**. The zone of apposition, thus, makes the lower RC in effect, a part of the AB container, whose movement is a direct result of changes in abdominal pressure (Pab). As the diaphragm shortens during inspiration, there is a piston-like displacement of the dome related to the axial shortening of the zone of apposition **[9,10]**, with a subsequent expansion of the thoracic cavity. During a quiet inspiration, contraction and descent of the diaphragm thus results in a fall in pleural pressure (Ppl) in the thoracic cavity, an increase in lung volume or decrease in alveolar pressure, depending on whether the airways are open or closed, and an increase in Pab. As the AB is mechanically considered a "liquid filled" container with the majority of its parts considered (ie, bony structures), its movement is largely limited to the ventral abdominal wall and to the diaphragm. Therefore, as Pab increases with diaphragmatic descent, the ventral abdominal wall moves outwards

The resultant fall in PpI has an expiratory action on the RC. This is counterbalanced by two inspiratory effects of diaphragmatic contraction upon the lower RC [8,11,12]. These are (i) the appositional component, which acts to expand the lower RC and is dependent on the amount of rise of Pab, AB wall compliance and the size of the zone of apposition, and (ii) the insertional component of the costal portion of the diaphragm, which also acts to lift and rotate outwards the lower RC mainly in the transverse direction, resulting directly from the action of the diaphragm at its insertion on the lower six ribs [7]. These two forces are active only if the contracting diaphragmatic muscle fibers remain oriented cranially, therefore the abdominal visceral mass acts as a brake to excessive diaphragmatic descent and also as the fulcrum against which the diaphragm pushes to lift the lower ribs [1,13]. It has also been demonstrated that the diaphragm has an expiratory action upon the upper RC, directly related to the fall in PpI incurred during diaphragmatic contraction [1,14,15].

Although authors disagree about the extent to which the diaphragm acts solely to cause the required inspiratory force [11,16], they all seem to agree that the inspiratory effect of the diaphragm is large at low lung volumes [7]. However, as lung volume increases, the zone of apposition diminishes, and there is less appositional force acting in the inspiratory direction via the increased Pab. As a consequence, the expiratory action

caused by the increase in the fraction of ribs exposed to PpI dominates [9,11,12] At extremely high lung volumes (approaching Total Lung Capacity (TLC)), as the zone of apposition disappears, the insertional forces of the diaphragm on the lower RC behave in an expiratory nature [9]

In the normal subject, utilizing the diaphragm as the sole muscle of inspiration would cause distortions of the RC at all lung volumes [1] However, at rest, a uniform expansion of both the upper and lower RC in the anterior-posterior (A-P) diameter and in the transverse diameter of the lower RC is observed. This indicates that even during resting breathing, normal humans contract other muscles of the RC to counter the distortional diaphragmatic forces, resulting in a uniform expansion and enhanced A-P diameter of the lower RC and expansion of the upper RC

2.3 The Neck Muscles

2.3.1 Sternocleidomastoid

The sternocleidomastoids originate at the mastoid processes and insert onto the ventral surface of the sternum and medial third of the clavicle. In normal subjects, during quiet breathing, the sternocleidomastoids are inactive and therefore not responsible for the upper RC expansion observed at rest. However, during high levels of ventilation (i.e. during exercise), or with inspiratory loading (i.e. experimental and diseased states), they actively help to cranially displace the sternum and expand the upper RC, predominately in the A-P direction. Their activity also results in some transverse expansion of the upper RC [1,17,18]

2 3.2 Scalenes

The scalenes consist of three bundles of muscles, running from the lower five cervical vertebrae to the upper surface of the first two ribs. It was previously thought that the scalenes, like the sternocleidomastoids, were only "accessory" muscles of inspiration, becoming actively recruited only as ventilatory demands were significantly increased above resting breathing [17]. Today, their role as a primary inspiratory muscle, even at quiet breathing has been recognized [18,19]. Their inspiratory activity is to expand the upper

RC mainly in the A-P direction [15], and to cranially displace the sternum [2] The scalenes shorten very little, even during a vital capacity manoeuvre, therefore it appears that the scalenes will maintain their contractile force even at very high lung volumes [18]

2.4 The Intercostal Muscles

There are two distinct muscle layers originating and inserting into the rib cage, the sternum and the spinal processes known as the external and internal intercostals respectively. The external intercostals originate from the tubercles of each upper rib dorsally to insert obliquely caudal and ventral to each respective rib below at the costochondral junctions and lay superficial and at right angles to the internal intercostals. In contrast, the internal intercostals extend from the sternocostal junctions ventrally to run obliquely caudal and dorsally from the rib above to the rib below to insert into the angles of the ribs [1].

The parasternal muscles are the interchondral portion of the internal intercostals and are the only muscles situated ventrally between the sternum and costochondral junctions [1,7].

Dorsally, the levator costae muscle runs from the transverse processes of the vertebrae cranially to the angle of the ribs caudally. After much controversy, it is now accepted that the external intercostal muscles, the levator costae and the parasternals are inspiratory in nature, acting to expand the RC, whereby the interosseous portion of the internal intercostals are expiratory in nature, decreasing RC diameter. Selective stimulation of the parasternals in dogs results in an elevation of the ribs, descent of the sternum and subsequently, an increase in lung volume [20]. Although the mechanisms of action in man of both the internal and external intercostals are still unclear [21,22], it is believed that through a coordinated mechanism, the internal and external intercostals are able to act in opposite directions on the RC [23,24]. Experimentation has also shown that the parasternals are rhythmically active even during quiet inspirations [23,25], while the sternum does move cranially in seated humans during quiet inspiration, indicating that the scalenes counteract the parasternal's action on the sternum.

It has been proposed that the intercostals (excluding the parasternals) may play more of a non-respiratory than true respiratory role, namely providing the muscular force for trunk rotation and postural maintenance [26,27]

2.5 Triangularis Sterni

This flat muscle is also known as "transversus thoracis" or "sternocostalis" It lies deep to both the sternum and the parasternal muscles. Its action is reciprocal to that of the parasternals, namely a caudal displacement of the ribs with a simultaneous cranial motion of the sternum [28]. Activity of the triangularis sterni is not seen in humans during quiet breathing in the supine posture [29], however activity has been observed in standing subjects at rest [30]. This muscle is continuously active during expirations below Functional Residual Capacity (FRC) as well as during spontaneous expiratory efforts (coughing, singing, speech) [23,29]. It can be assumed that during increased ventilatory demands, eg. exercise, the triangularis sterni will become actively recruited along with the internal interosseous intercostals to lower the ribs, increase pleural pressure and to actively force the expiration [23].

2.6 Abdominal Musculature

As previously mentioned, the abdomen is a virtually incompressible liquid-filled container, hence only the diaphragm or ventral abdominal wall can be displaced. The ventral lateral wall is comprised of four main abdominal muscles, whose functions play a major role in respiration [1]

The most ventral of these muscles is rectus abdominis, originating from the ventral aspect of the sternum, the fifth, sixth and seventh costal cartilages and runs along the entire length of the ventral abdominal wall to insert caudally into the publs. The other three muscles form the lateral abdominal wall and their aponeuroses form a sheath about rectus abdominis. These are 1) the external oblique which originates from the outer surface of the lower eight ribs (covering both the lower ribs and intercostal muscles) to insert into the iliac crest and inguinal ligament, 2) the internal oblique which lies deep to external oblique and originates from the iliac crest and inguinal ligament and runs cephalically to insert on

the costal margin and also forms an aponeurosis which comprises part of the rectus sheath, and 3) the transversus abdominis which lies the deepest and originates from the inner surface of the lower six ribs and interdigitates with the costal insertions of the diaphragm, and from the lumbar fascia, iliac crest and inguinal ligament. Its muscle bulk runs circumferentially about the abdominal viscera to insert ventrally into the rectus sheath [2]

Mechanically, these four muscles act in both non-respiratory functions as the prime flexors and rotators of the trunk and in postural stability as well demonstrating respiratory actions. Two principal actions are displayed by the abdominal muscles in respiration. First, their contractions result in an increase in Pab which in turn forces the diaphragm cranially into the thoracic cavity. This diaphragmatic ascent increases pleural pressure and (if the glottis is open) decreases lung volume Secordly, contraction of the abdominal muscles also has a direct effect upon the RC – Given their insertional attachments to the lower ribs and sternum, contraction of the abdominals should result in a caudal displacement of the lower ribs and subsequent deflation of the RC However, experimentation has demonstrated that abdominal action on the RC is more complex than just a pure insertional, expiratory effect. An opposing inspiratory action of the RC has been demonstrated in dogs [22] This "inflation" effect on the RC is directly due to the increase in Pab resulting from abdominal muscle contraction. The zone of apposition of the diaphragm permits the direct transmission of Pab to the lower RC - Also, when the increased Pab forces the diaphragm cranially, this muscle is stretched, and the resultant passive tension allows the diaphragm to act through its insertional forces to raise the lower ribs [2,22,30]

Although the shape of the RC differs in humans, and therefore insertional forces are different, it is still proposed that the action of the abdominals upon the RC is a balance between the insertional, expiratory force and the inspiratory force related to the rise in abdominal pressure [1]. Upon isolation, contraction of the rectis abdominis in humans results in a caudal displacement of the sternum, a significant decrease in the A-P diameter of the lower RC and a small increase in its transverse diameter [31]. Contraction of the sternum and a marked decrease, rather than an increase as seen in the dog, in the RC transverse diameter [31]. In humans, the isolated activity of internal oblique and transversus

abdominis have not been demonstrated

The role of the abdominals as expiratory muscles is not disputed, but their role as "accessory" muscles to inspiration is now acknowledged. By contracting during expiration, the abdominal muscles can reduce lung volume below the passively determined endexpiratory volume. Elastic energy may be stored in the chest wall during expiration, which subsequently could assist the proceeding inspiration [7]. Also, the resultant cranial displacement of the diaphragm increases its curvature, decreases its radius and thus allows for a greater generation of pressure for a given tension [7]. Abdominal muscle activity during expiration also improves the efficiency of the diaphragm as a pressure generator, improving its ability to expand the lower RC, due to an increase in the size of the zone of apposition [32]. The inspiratory muscles are also lengthened to a more favourable operating length, permitting them to generate a given amount of volume with much less activation [7].

It has been demonstrated in the horse [33] and the dog [34,35] that at the end of expiration, the respiratory system is below its resting position and at the onset of the next inspiration, abdominal muscle relaxation allows the diaphragm to descend passively prior to active inspiratory muscle activity. In the dog, this passive abdominal relaxation at the onset of inspiration, prior to the onset of inspiratory muscle activity, accounts for 50-60% of that animal's tidal volume (Vt) [36]

In humans, tonic abdominal muscle activity, assumed to be unrelated to respiration, is observed at rest in the standing position [30,37] However, during conditions of CO₂ breathing or inspiratory loading, rhythmic expiratory recruitment of the abdominal muscles is observed [38,39,40]

CHAPTER 3 PATHOPHYSIOLOGY OF CHRONIC OBSTRUCTIVE PULMONARY DISEASE

3.1 Symptoms and Etiology of COPD

The broad term "chronic obstructive pulmonary disease (COPD)" is traditionally used to describe three specific conditions: chronic bronchitis, asthma and emphysema, all of which are characterized by either chronic expiratory airflow limitations and/or increased airway resistance [41] Established risk factors for COPD include increased age, cigarette smoking, occupational exposures, air pollution and alpha, -antitryps in deficiency [42].

All patients participating in this study were clinically diagnosed as having chronic bronchitis, emphysema or a combination of both. Asthmatics were not studied therefore the following discussion will focus only on the related pathophysiologies, symptoms and limitations associated with chronic bronchitis &/or emphysema.

The major complaint of patients with COPD is dyspinea which limits or affects employment and recreation as well as activities of daily living [43]

Diagnosis of COPD is mainly achieved non-invasively by pulmonary function testing (PFT), including spirometry, evaluation of the flow-volume loop and lung volume measurements [44,45]. The forced expired volume in one second (FEV₁) is the volume of gas expired in one second following a maximum inspiration [46] and is expressed as the percentage of its predicted value which is based upon the subject's age, sex and height [47]. The lower the calculated ratio, usually the more severe the patient's symptoms are. The Forced Vital Capacity (FVC) is the maximum amount of air that can be exhaled from the lungs following a maximum inspiration. It is effort and time dependent [45]. Severity of an obstructive airway disease can be assessed by the FEV₁/FVC ratio. An FEV₁/FVC of 80% is normal, 65-79% represents mild impairment, 50-65% indicates moderate impairment and less than 50% represents a severe obstruction to airflow [48]. Causes of this airway obstruction include excess mucous in the airways, edema, active broncho-constriction, and loss of lung tissue supporting the airway [49].

Analysis of flow-volume loops demonstrates low flow rates for a given volume in

obstructed patients When maximum expiratory flow is severely reduced, expiratory flow during tidal breathing may reach its maximum level [50]

There are also observed changes in lung volumes. Functional Residual Capacity (FRC) and Residual Volume (RV) are consistently increased in patients with COPD [51,52] and in some patients Total Lung Capacity (TLC) is also increased (severe emphysema). This increase in TLC is mainly due to the large RV measured. Resting tidal volumes (Vt) are often reduced, whereas the breathing frequency (fb) is increased, thus maintaining a normal or even higher than average minute ventilation. VE, (greater increase in fb than decrease in Vt) during quiet breathing. This pattern of breathing is the result of major adjustments in respiratory muscle activation from the normal pattern [53].

In order to achieve a given volume, large swings in PpI are required in order to overcome increased airflow resistance and reduced dynamic compliance. In an attempt to minimize the negative effects of expiratory flow limitation, COPD patients often have reduced T_{I}/T_{IOI} (duty cycle), to allow more time for expiration [53]. Expiratory time (Te) is increased at the expense of inspiratory time (Ti). COPD patients also breathe from much higher resting volumes, due to the larger airway size and higher maximum expiratory flows that can be achieved from volumes above FRC [53].

Even though the observed airflow limitation is expiratory in nature, the COPD patient mainly compensates by increasing the work of the inspiratory muscles, observed through the larger inspiratory swings in PpI. This is a compensatory method to not only increase the mean inspiratory flow necessary to sustain a total ventilation when Ti is reduced but also, since breathing is occurring at high lung volumes, it is necessary to overcome the concurrent increase in the clastic work of breathing required to inflate the lungs and chest wall. Since more negative PpI is required to achieve a given tidal volume due to the combined effects of increased airflow resistance, decreased dynamic lung compliance and increased FRC, the inspiratory muscles are forced to chronically work at a higher than normal fraction of their maximum inspiratory force capacity [54]. Hyperinflation therefore increases the mechanical work of breathing through increases in both the frictional and elastic work required.

Severe COPD patients often exhibit marked positive end-expiratory alveolar

pressure since the energy potential developed during inflation is insufficient to return the system to a relaxed end-expiratory equilibrium point. This has been termed "auto-PEEP" and provides an added force the inspiratory muscles must overcome before the next inspiration can begin [55].

It has been demonstrated that during hyperinflation, the inspiratory muscles (intercostals and accessories) contract before expiration has proceeded long enough for the lung and chest wall to reach their respective relaxation volumes [56]. Therefore, the inspiratory muscles appear to contribute to the increase in FRC and keep the chest at a higher compensatory volume than would occur solely from the airway obstruction [57,58]. Also it was found that O_2 consumption of the respiratory muscles in patients of COPD is increased, due to the related distortions of the rib cage caused by hyperinflation, out of proportion to the actual energy required to perform the mechanical work of breathing [59]. The energy cost of breathing is greatly increased for the pressure actually being generated by the breathing effort [55].

3.2 Mechanics of Breathing in COPD

In patients suffering from COPD, accessory muscles of inspiration are often recruited even at quiet breathing [60], however, hypertrophy of sternocleidomastoid or the scalenes are rarely observed

As previously mentioned, the sternocleidomastoids actively contract to raise the sternum and to expand mainly the AP diameter of the upper rib cage [1,17,18]. Since the scalene muscles shorten very little, even during full Vital Capacity (VC) manoeuvres, their integrity as a primary inspiratory force is not severely compromised in states of hyperinflation in COPD patients.

With hyperinflation, where the patient is breathing at extremely high lung volumes (approaching TLC), the diaphragm becomes flattened, thus, the insertional forces of the diaphragm on the lower rib cage also behave in an expiratory nature [9], drawing the lower ribs inwards. Also, the zone of apposition disappears with hyperinflation, therefore, appositional forces in the inspiratory direction are compromised and replaced by expiratory forces due to the higher fraction of lower ribs now exposed to pleural rather than gastric

pressure [9,11,12] As a result, the inspiratory action of the diaphragm is compromised [fig 3 1]

Normally as a muscle shortens, its ability to generate force and tension decreases Therefore, increasing lung volume has a direct effect on the diaphragmatic muscle fibres, affecting their contractile force as the length-tension characteristics of the muscle predicts **[61]** Lower transdiaphragmatic pressure can be generated for a given tension **[55]**. During acute hyperinflation which may occur during an asthmatic attack or during high, strenuous exercise, the diaphragm will lose its ability to act as a prime inspiratory force Chronic hyperinflation, however, has been shown to result in chronically shortened diaphragm fibres in hamsters **[62,63]** Therefore, if an analogy can be drawn, the length-tension characteristics may not play as limiting a role to the ability of the diaphragm to generate tension in a chronically hyperinflated patient as previously thought. The insertional force changes will probably become much more significant

Similarly to, but to a lesser degree than the diaphragm [64], severe hyperinflation places the inspiratory intercostals (external, parasternals and levator costae) at a shortened length on their respective length-tension curves. Therefore, they become more horizontally oriented losing part of their 'bucket handle' action, resulting in a decrease in the inspiratory activity which the inspiratory intercostals are capable of generating.

The role of the abdominal muscles during hyperinflation has also been addressed It has been proposed that the increased abdominal muscle activity during hyperinflation attempts to lengthen the diaphragm and restore it to a more normal resting length. Also, upon inspiration, relaxation and therefore outward recoil of the abdominal wall has been demonstrated, acting to lower gastric pressure and facilitate diaphragmatic descent [57]. This role of the abdominal muscles as "accessory muscles" of inspiration may attempt to compensate for the added energy costs of the inspiratory intercostal and accessory muscle activity observed during hyperinflation in patients with airway obstruction [57].



Figure 3.1 The effects of hyperinflation on respiratory muscle function [65]

CHAPTER 4 VENTILATION DURING EXERCISE

The respiratory system can be divided into two major functional components, the lungs themselves, where gas-exchange occurs, and the ventilatory pump. This pump is comprised of the chest wall (rib cage and abdomen), the respiratory muscles, the related controlling centres of the central nervous system and the corresponding neural connections [66,67,68]

Exercise causes the metabolic demands of the exercising muscles to increase, therefore placing direct demands upon the respiratory system. This increase in metabolic rate results in increased CO_2 production and increased O_2 requirements, demands which are met by the respiratory system by moving larger volumes of air into and out of the lungs [68]

Exercise can be classified as static (isometric) or dynamic. It has been demonstrated that during static exercise, there is only a small increase in ventilatory demands [69,70], whereas, dynamic exercise imposes much greater demands upon the respiratory system. The proceedingstudy dealt with dynamic exercise testing, therefore it will be further detailed.

The increases in ventilatory demand are accomplished through increasing minute ventilation (VE),

	$VE = Vt \times fb$	(1)
whereby	Vt = tidal volume	
	fb = breathing frequency	

4.1 Ventilatory Responses to Exercise in Normal Subjects

Resting minute ventilation in normal subjects has been shown to be about 10 litres/minute, representing approximately 5% of a normal maximal ventilatory capacity and comprising only 1-2% of the total O_2 consumption of the body [71,72,73]. The maximum voluntary ventilation (MVV), represents the maximal ventilatory capacity of the respiratory system. It has been demonstrated that the MVV can only be sustained for 15-30 seconds, whereas 75% of the MVV can be sustained for about 4 minutes and 60% of the

MVV can be sustained for 15 minutes or longer. Thus 60% of the MVV is termed the maximum sustainable ventilation (MSV). Since it is inconceivable for exercise testing to last 30 seconds or less, it is more appropriate to consider the measured VE in relation to the MSV [56,74].

Ventilation has been shown to increase immediately at the onset of exercise [75,76], and during one minute incremental increases in exercise workload, the increase in ventilation is in direct proportion to the increases in CO2 output, up to the workload where metabolic acidosis is initiated [77] The increase in VE is achieved through a combination of increasing tidal volume (Vt) and breathing frequency (fb) [68] During light to moderate exercise, most of the increase in VE is achieved through increasing Vt, accompanied by a progressive increase in fb. However, it has been demonstrated that at higher workloads, the increase in Vt is constrained to about 50% of the individual's Forced Vital Capacity (FVC) [78,79,80] and increasing fb becomes the major contributor to the increasing VE with either no further increase or only slight increase in Vt observed [73,80,81] Breathing frequency during maximal exercise is normally about 50 breaths per minute, but this value may be higher in highly trained athletes [82] and at these maximal frequencies, a decrease in Vt may actually be observed prior to the cessation of exercise [75,83] It is believed that this pattern of a Vt of approximately 50% of FVC and an fb of 50 breaths/minute during maximal exercise minimizes the work of breathing, respects the mechanical limitations of the system and provides a sufficient alveolar ventilation to eliminate the CO, and maintain homeostasis of arterial blood gases and hydrogen ion concentration [83,84,85,86,87,88,89]

The pressure-volume curve of the respiratory system is such that by increasing Vi predominantly through increases in Vt, the flow-resistive work of breathing is minimized and only when Vt's exceed 75% of Vital Capacity (VC) do increases in the elastic work of breathing become significant [84]. The maximum flow volume (MEV) curve also provides constraints which are generally respected during exercise [86,88], however, it has been demonstrated that when some normal subjects reach maximum exercise, expiratory flows may exceed those predicted by the MEV curve [86]. It has been suggested that this may be due to a bronchodilation effect of exercise, a change in the elastic properties of the lung.

or merely by the incorrect placement of the tidal flow-volume curve during exercise on the volume axis of the resting maximum expiratory flow-volume (MEFV) curve [88].

4.2 Ventilatory Responses to Exercise in COPD Patients

Patients with COPD have similar VE at rest when compared to normal subjects [90], however, due to reduced airway conductance and reduced respiratory muscle strength, both the MVV and MSV are dramatically decreased [91,92] It has been shown that the relationship between MVV and MSV is the same in COPD patients as in normals [74], however, given their reduced ventilatory reserve, their resting VE may account for as much as 40% of the MVV [74,91,93] Also, the oxygen consumption of the respiratory muscles during quiet breathing has been shown to be about 15% of total body O, consumption [71,94] Thus, the COPD patient can only achieve moderate increases in ventilation during exercise [91,95,96] and these increases in VE are largely due to disproportionate increases in fb as compared to Vt [97,98,99] Patients with even moderate degrees of air-flow obstruction have been shown to have expiratory flows which lie on the MEFV curve at quiet breathing [91,97,100], thus the COPD patient is faced with limited strategies to further increase VE The patient may shorten inspiratory time (Ti), to allow greater time for expiration, however, this results in a decreased duty cycle (Ti/Ttot) and a large increase in mean inspiratory flow (Vt/Ti) requiring the inspiratory muscles to shorten much more rapidly [68,98] The second possible strategy is to increase endexpiratory lung volume [97], allowing higher maximum expiratory flows, a more moderate increase in Vt/Ti and preservation of Ti/Ttot [68,98] This compensation for the limitations of expiratory flow result in much higher inspiratory elastic work requirements as well as an increase in the flow-resistive components of inspiratory work due to the higher Vt/Ti for a given minute ventilation [100] This second strategy has generally been observed by investigators [89,98,100], but adoption of either the first or second strategy or a combination of both may depend upon the severity of the disease and to the extent of expiratory flow limitation As in normal subjects, some investigators have observed that COPD subjects can achieve expiratory flow rates during maximum exercise that exceed their resting MEFV curve [91,97], but the accuracy of these results have also been

questioned for the same reasons presented in the previous section concerning normal subjects [89]

4.3 Limitations of Exercise

Normally, the ventilatory system has not been regarded as a limiting factor of exercise in normal subjects; however, in patients with COPD, their severely limited vertilatory capacity will result in a reduced exercise capacity [89,91,93,100,101] Some evidence, though, has been proposed indicating that indeed the respiratory system may somewhat limit exercise in normal subjects at severe workloads. Artenal hypoxemia has been observed, during heavy exercise, in some highly trained athletes [102,103,104] These authors suggested that the mechanics of the respiratory system, muscle fatigue or the energetics of respiratory muscles may have set the VE at a level lower than could have been achieved, since ventilatory reserve still existed, at the cost of arterial hypoxemia [103] It has also been demonstrated that a mechanical constraint to increasing ventilation due to an expiratory flow limitation may exist for normal subjects undergoing heavy exercise since expiratory flows approach or even exceed the maximum expiratory flows as defined by the resting MEFV curve at the end of expiration [86] or achieve maximum values throughout most of expiration [105] In order to further increase their expiratory flows, and thus further increase ventilation, the athletes would be forced to breath at higher lung volumes. This hyperinflation would place the new shortened inspiratory muscles at a mechanical disadvantage, increasing the elastic work of breathing, decreasing mechanical efficiency and increasing the vulnerability of the inspiratory muscles The flow-volume loop limitation to ventilation may thus be to develop fatigue [106] valid for normal subjects as it has been shown that exercise ventilation does improve with inco, breathing, possibly through a reduction in the resistive work of breathing [107,108,109] or to an enlargement of the maximal flow-volume loop [110] Another alternative to increasing VE without changing resting lung volume would be to increase inspiratory flow rates, however, studies in normal subjects at high intensities of exercise show that both inspiratory flows and pleural pressures are already close to maximum achievable values [86,105]. Based upon this observation, it has been suggested that the force-velocity characteristics of the inspiratory muscles may therefore impose a constraint on further increases in inspiratory flows [111]

The above discussion concerning mechanical limitations to exercise also directly applies to patients with COPD, whereby levels of ventilation achieved during exercise often reach the subject's maximum breathing capacity [91,93] and further increases in expiratory flow is limited by maximal flow-volume loop constraints [89,91,93,100]. Exercise limitation in COPD subjects is also related to reduced respiratory muscle strength (RMS) [92]. Hyperinflation (associated with increases in FRC) is often marked in COPD subjects, resulting in shortening of the inspiratory muscles and diaphragmatic flattening, thereby resulting in decreases in these muscles' contractile forces [61,112] and ultimately decreasing the maximal inspiratory pressures for a given volume that they are capable of generating. Other factors such as prolonged inactivity, steroid administration and chronic hypoxemia also weaken RMS in COPD patients [113].

Respiratory muscle fatigue can be defined as "a condition in which there is a loss in the capacity for developing force and or velocity of a muscle in response to a load and which is reversible by rest" [114] Inspiratory muscle fatigue during heavy exercise has been shown in some patients, indicated by electromyographic spectral analysis [115,116,117] and transdiaphragmatic pressure measurements [118]. However, other investigators have failed to observe changes in the electromyographic frequency spectrum indicative of diaphragm fatigue in their exercising COPD patients despite very high pressure generations [119] They also failed to observe any changes in post exercise breathing patterns [120] or increases in the values of the pressure time index of the diaphragm (TTdi) which exceeded the known fatigue threshold [121]

The combination of hyperinflation and further dynamic increases in FRC which occur during exercise in COPD subjects [89,122] forces the inspiratory muscles to operate at a disadvantageous length on the length-tension curve where they are predisposed to developing fatigue [74,106,123,124], however, there is disagreement as to whether inspiratory muscle fatigue does limit exercise

Finally, dyspneahas been shown to significantly limit exercise in COPD patients and in normal subjects excrosing at extremely high levels of work.
CHAPTER 5 MECHANICS OF BREATHING AND RESPIRATORY MUSCLE COORDINATION

Mechanically, chest wall behaviour is a function of both the active contractile properties of the respiratory muscles and the passive mechanical properties of the chest wall's major components, that is the rib cage, diaphragm and abdomen. Coordination is used to describe the neuromechanical interaction among different respiratory muscle groups as well as between respiratory and non-respiratory muscles [125]

5.1 Historical Development

Early investigators considered the chest wall as a single compartment and studies focused on the static volume-pressure relationship of the respiratory system [126,127] It was assumed that mouth pressure (P_{etre}) represented the sum of the pressure severted by the recoil of the chest wall (P_{etre}) and the lung (P_{etr}) . The inability to directly measure these pressures, however, posed a major limitation, forcing early investigators to use known in-vitro pressure-volume relationships of the lung to determine P_{etre} :

$$P_{el,w} = P_{elrs} - P_{ell}$$
(1)

This limitation was overcome in 1949 with the introduction of the esophageal balloon technique for measuring transpulmonary pressures *in vivo* [128] Now the respiratory system could be partitioned into both its chest wall and lung components

Agostoni and Rahn [129] developed the method for measuring transdiaphragmaticpressure (Pdi), permitting a more complete analysis of the contributions of the thoracic, diaphragmatic and abdominal components to the mechanical behaviour of the respiratory system. Pdi was determined by subtracting the gastric pressure (Pga) (measured as an index of abdominal pressure) by the esophageal pressure (Pes) (measured as an index of intrathoracic or pleural pressure). They determined that Pdi could be generated not only by active diaphragmatic contraction but also through the passive stretching of the muscle. Measuring Pes and Pga also permitted the quantification of the pressures acting to displace both the abdomen (AB) and the rib cage (RC) [125].

A parallel action of the RC and diaphragm was assumed, leading investigators to assess increases in lung volume as the summation of RC expansion and diaphragmatic descent [130,131,132]

5.2 Analysis of Rib Cage and Abdominal Displacement

Konno and Mead in 1967 hypothesized that the chest wall enjoyed two degrees of freedom, thus changes in lung volume could be accommodated independently through RC or diaph/agm-AB displacement [133] Using linear differential transducers, the anteroposterior (A-P) displacements of the RC versus the AB were plotted under both active (isovolume manoeuvres) and passive conditions (giving the RC-AB relaxation line) at volumes ranging from residual volume (RV) to total lung capacity (TLC) [Figure 5.1] Isovolume manoeuvres are performed when the subject, with glottis closed, alternately contracts his abdominal muscles while simultaneously expanding his rib cage, thus displacing volume from the abdomen into the rib cage, and then relaxing his abdominal muscles and depressing his rib cage, thus displacing volume from the rib cage into the abdomen. The result is a series of plotted isovolume isopleths from RV to TLC representing the different configurations and maximum excursions (combinations of rib cage and abdominal wall) that the chest wall can have for any single lung volume

It was proposed that respiratory muscle activity could be inferred from plotting subsequent relative motion diagrams, whereby A-P motion of the RC is plotted versus A-P motion of the AB during breathing, in relation to the plotted relaxation line and isovolume isopleths. Any deviations from the relaxation line would require muscular activity Deviations to the left of the relaxation line inferred either inspiratory activity operating on the rib cage, expiratory abdominal activity or a combination of the two, whereas deviations to the right of the relaxation line inferred expiratory muscle activity operating on the rib cage, inspiratory activity operating on the abdomen or some combination of the two [133].

RC and AB displacement during quiet spontaneous breathing was found to lie very close to the respective relaxation line, however it was noted that slight deviations to the right of the relaxation line occurred during inspiration, indicating that some inspiratory activity was operating on the abdomen [133]. The relative volume changes of the RC and the AB during



Figure 5.1 Konno-Mead Diagram [134].

increasing ventilation stimulated through hyperventilation [133], re-breathing [135] and exercise [136] demonstrated substantial deviations from the RC-AB relaxation line. During exercise, a shift towards the left of the relaxation line, representing a decrease in AB volume subsequent to increased expiratory abdominal recruitment, was observed. This was especially marked at end-expiration. However, RC volume at end expiration was found to be unchanged from values recorded at quiet breathing [136].

Konno-Mead analysis, however, is limited as it does not measure diaphragmatic displacement (whose action results in combined RC and AB displacement) nor does it measure cranial displacement of the rib cage or spinal flexion/extension [133]

Incorporating the method of RC and AB volume partitioning [133] with established Pes and Pga relaxation measurements [129], the static volume-pressure characteristics of the rib cage and abdomen were evaluated [137]

Utilization of the Konno-Mead diagrams permits a non-invasive measurement of ventilation and the resulting volume measurements have been shown to be within 10% of those volumes measured with spirometry [125]

Magnetometers (for measuring RC and AB linear displacement) [138] and more recently, a respiratory inductive plethysmograph (for measuring the cross-sectional area of the RC and AB) [139] are commonly used tools to measure and characterize chest wall displacement. Utilizing respiratory inductive plethysmography permits changes in the transverse direction to be measured, something not permitted in early studies utilizing linear transducers

5.2.1 Respiratory Inductive Plethysmography (RIP)

Briefly, RIP (Respitrace, Ambulatory Monitoring, Inc., White Phans, NY.) consists of 2 coils of Teflon-insulated wire, sewn onto separate elastic bands which encircle the rib cage and the abdomen and connect to an oscillatory module. Therefore, changes in the cross-sectional areas of the RC and AB alter both the self-inductance of the coils and the frequency of their oscillators, which in terms is reflected as changes in tidal volume [140,141]

Several calibration methods of RIP have been described [139,142,

143,144] A commonly used and simplistic calibration method is the isovolume manoeuvre calibration method, based upon the work of Konno and Mead **[139]** The validity of measured volumes by RIP during moderate exercise has been shown to be within 20% of those measured with spirometry, however its accuracy in monitoring end-expiratory volume is accurate only if body position remains constant (which is often impossible during exercise testing) **[139]** Satisfactory results have been obtained using Respitrace in workloads up to 1800 kpm/min, though extraneous body movement and either RC or AB wall distortion associated with much higher levels of exercise can limit its accuracy for measuring volumes. Respitrace, however, does accurately monitor relative RC and AB motion during breathing and offers a simple, non-invasive method to qualitatively assess changes in chest wall displacements. Qualitatively, the inference of muscle recruitment and coordination is allowed during moderate to high levels of exercise.

5.3 Methods to Infer Respiratory Muscle Activity-The Analysis of Pressure Changes

Previously, only the measurement of global inspiratory or expiratory muscle activity could be quantified as the difference between the dynamic PpI developed during breathing and the static relaxation pressure measured at the same volume. In an attempt to quantify individual contributions of chest wall muscles to breathing, Goldman *et al* **[16]** investigated the relationship of Pes and Pga to RC and AB volumes. They hypothesized that when the subject is upright and RC and AB muscles are relaxed, the diaphragm is the only importantly active muscle. At rest, therefore, diaphragmatic contraction would result in an increase in Pga which would displace both the RC and the AB along their relaxation characteristics and the resultant fall in PpI would act to inflate the lungs. As an extension to this early study, the concept of inferring respiratory muscle action from volume-pressure partitioning was introduced **[145]**. It was suggested that any observed departures from the static relaxation line when volume was plotted against pressure was a combined result of increasing muscle activity (ie: intercostals/accessories) and the coordination of that activity with respiratory muscle activity in other parts of the chest wall. Plotting Rib cage volume (V_{RC}) versus PpI during breathing, observed deviations from the relaxation were said

to reflect the combined influence of the diaphragm, intercostals and accessories. Departures from the static relaxation line of V_{RC} to Pab observed during increasing ventilation was said to represent solely intercostal and accessory muscle activity

It was determined that diaphragmatic length is reflected by changes in abdominal volume [146] It was observed that during increasing levels of exercise, end-inspiratory diaphragmatic length did not change however at end-expiration, the diaphragm was lengthened (inferred by observed decreases in abdominal volume). This led to the proposal that active expiratory abdominal muscle contraction appears to have an important inspiratory role. This active expiratory abdominal activity results in a lengthening of the diaphragm to a more optimal length [7], subsequently decreasing the load placed upon the diaphragm by the subsequent inspiration [146]. In conclusion, it was believed that increasing activity of both RC and AB muscles observed during exercise functioned to optimize diaphragmatic function [145]. During exercise, the observation was made that during inspiration, no change in Pab sometimes occurred. It was thus concluded that the diaphragm was no longer functioning to displace the chest wall but rather the RC and AB muscles, working in series with the diaphragm, had taken over that role and the diaphragm's sole function was to inflate the lungs ($\Delta Pdi=-\Delta Ppl$) [145]

In contrast, Macklem *et al* proposed that the intercostal/accessory muscles and diaphragmact more in parallel, and that the diaphragm was capable of contracting quasiisometrically, acting as a fixator to prevent the transmission of PpI to the AB, hence not performing any external work [147]. Therefore, if the AB muscles are relaxed and Δ Pab =0, these investigators deduced that - Δ PpI was achieved by a combination of contracting the intercostal/accessory muscles and by quasi-isometrically contracting the diaphragm where in contrast to earlier studies where they concluded that the intercostal/accessory muscles were solely responsible for all work required to inflate the lung [147].

5.3.1 The Macklem Diagram

The Macklem diagram was thus developed as another attempt to infer inspiratory muscle activity [fig 5 2] The Macklem diagram plots Pel (or esophageal pressure (Pes)) measured via an esophageal balloon versus Pab (measured via a gastric balloon) and



Figure 5.2 Macklem diagram [147].

•

provides a means of inferring the pattern of recruitment of muscles without having to measure volume or chest wall displacement. During quiet breathing, similar results were found by previous investigators [16], whereby the diaphragm appeared to be the major muscle contracting and the PpI-Pab relationship lies along the relaxation line (line AB). Displacement off this relaxation line infers inspiratory muscle recruitment. When only the intercostal/accessory muscles are active, PpI decreases and Pab decreases by the same amount (PpI is transmitted through a passive diaphragm), resulting in rib cage expansion and a paradoxical inward movement of the relaxed abdominal wall (line AC). Intermediate patterns are said to represent the combined action of intercostals and the diaphragm (line AD) [125,147].

The idea of a quasi-isometric diaphragmatic contraction has now been rejected. It now appears that mechanically the diaphragm and rib-cage muscles are linked both in parallel and in series as they act together upon inspiration [148]. If the rib cage expands and the diaphragm is solely acting as a fixator, abdominal pressure should decrease since the RC would act to expand that part of the abdomen contained within the diaphragm, thus resulting in a decrease in Pab unless abdominal muscle tone increased or some diaphragmatic contraction occurred [149]

Although the quantification of the contribution to breathing of individual muscles can not be achieved through the use of Macklem diagrams (ie: the abdominal wall is never totally relaxed during exercise), the Macklem diagram does permit the qualitative inference of respiratory muscle recruitment. Pdi iso-lines, whereby Pdi = 0, can be construed and thus permitting the calculation of the total diaphragmatic contribution in terms of pleural and abdominal pressure changes from the x-y coordinates of the PpI-Pab plot [125]. Combining both the relative motion diagrams introduced by Konno and Mead and the Pes-Pga analysis achieved through the use of Macklem diagrams provides the investigator with a simplified method of qualifying respiratory muscle activity.

CHAPTER 6 RESPIRATORY PSYCHOPHYSICS

6.1 Historical Development

The exploration and elucidation of the subjective sensory process has largely been addressed through the science of psychophysics. Unlike objective sensory physiology, where one is able to measure both the strength of the stimulus and the amplitude of the response through either physical or chemical means, subjective sensory studies utilize urique subjective measurement systems [150]

In 1846, Weber reported that the "just noticeable difference" (JND) in intensity between two stimuli is a constant fraction of the intensity of the first stimulus (Weber's Law) [151] In 1859, G T Fechner defined psychophysics as an exact theory of the relation of body and mind [152] and based on Weber's findings, developed a law stating that sensation intensity grows as a function of the logarithm of the stimulus (Fechner's Law) [153]

S S Steven's modified these early assumptions of Fechner and reported that the psychological magnitude of a given sensation is a power function of the physical magnitude of the stimuli, (Steven's Power Law), [153],

	$\psi = \mathbf{k} \Phi^{\prime\prime}$	(1)
whereby	ψ = the psychological magnitude of the perceived	
	sensation	
	Φ = the physical magnitude of the stimulus	
	k = a constant	
	n = the associated exponent	

The value of n can be obtained by performing a logarithmic transform of the relationship, $\log \Psi = \log K + n \log \Phi$. Therefore, n is the slope of the best-fitting regression line between stimulus magnitude and stimulus intensity, and its value has been used to quantify subjects' responses or perceptual sensitivities to a variety of added stimuli Perceived magnitude may rise faster than stimulus magnitude, n>1, it may rise at the same rate, n=1, or it may rise slower, n<1 [154]. Most psychophysical relations may be described by Steven's Power Law, with exponents of n ranging from 0.3 to 3. The value of n has been utilized to describe the relationship between sensory magnitudes and to

evaluate differences in sensory perception between subjects and under different experimental conditions

A large range of subjective sensations, examined under many stimuli and by varying methodologies have been quantified through the use of Steven's Power Law [153,155,156] These studies have provided the basis for the direct scaling methods used in studies of respiratory sensations.

6.2 Scaling Methods

A number of potential scales exist, each demonstrating intrinsicly different mathematical and statistical properties [157] There are two main ways one can construct a sensation or response scale, indirectly and directly.

6.2.1 Indirect Scaling

The first approach is "indirect" scaling, whereby the subject perceives one stimulus from another This approach applies Weber's JND concept and its use has largely been associated with qualifying, rather than quantifying perceptions [156,158]

622 Direct Scaling

The concept of direct scaling in psychophysics has largely been contributed through the works of S S Stevens, based upon his Power Law relationship. In direct scaling tasks, observers make quantitative judgements of sensory magnitude [153,159,160]. There are two types of direct scaling, ratio scaling and category scaling.

6.2.2 1 Ratio Scaling

A ratio scale is one that possess a unique zero. Subjects are asked to rate their perceived sensation based on a ratio relationship. For example, the subject may use any number, including fractions, to assess sensation. The subject is to'd to focus on ratio relationships among stimulus intensities. If one stimulus appears five times as intense as another, this judgement could be represented by the numbers 1 and 5 or 5 and 25 respectively [159].

6.2.2.1.1 Magnitude Estimation

Magnitude estimation, developed by Stevens [161], is a commonly used ratio technique to assess perceived stimulus intensities. Subjects are asked to give a direct numerical estimation of their subjective sensation for each of several stimulus levels, either with reference to a given standard stimulus or with no previously given standard. The majority of magnitude estimation studies employ the latter, termed open magnitude scaling Studies have demonstrated the validity of Steven's power law for a number of perceived sensations scaled through magnitude estimation [153,156]

6 2.2.1.2 Magnitude Production

This technique involves the experimenter presenting numbers one at a time in a random order, and the subject is asked to adjust the stimulus to match the given numerical value [161]

6.2.2.1.3 Cross-modality Matching

Subjects are asked to manipulate the intensity of a stimulus in one modality, e.g. loudness, to match the perceived intensity of the magnitude of the stimulus of a different modality, e.g. light intensity [159]

One major drawback of ratio scaling is in its ability to quantitatively permit intersubject comparison and evaluation

6.2.2.2 Category Scaling

The subject is asked to rate their perceived sensation on a provided scale, which is limited by an interval range and type of numbers (ie-subject is not permitted to fractionize) A true zero does not exist. Though Stevens has cautioned on the use of such category scales to quantify perceived sensations [152], recent investigators have returned to modified category scales for investigating perceived sensations in the clinical setting as they permit inter-subject comparisons

6 2 2 3 Visual Analogue Scale (VAS)

The visual analogue scale (VAS) is one type of a ratio, interval scale which has been utilized in the quantification of a variety of sensations [162,163,164] The VAS consists of a line, usually 100 mm in length, which represents the full range of severity of the investigated sensation. The line may be oriented either horizontally [162], or vertically [164] In studies of dyspnea, the bottom of the scale is described by a phrase such as "no breathlessness" while the top of the scale is tagged by the phrase "greatest breathlessness" or "maximum imaginable breathlessness" [164,165]. Subjects are asked to mark on the scale their level of the sensation under investigation at given intervals of time and the distance (ie in millimeters) from the bottom of the scale to the marked level of sensation is quantified. Subjects freely set their own interval size, consequently, inter-individual comparisons are still problematic [166].

6.2.2.4 Category-Ratio Scaling-The BORG Scale

One major drawback with previously mentioned ratio-scaling methods is that they do not provide any direct "levels" for inter-individual comparisons. However, utilizing a ratio scale does permit mathematical calculations (ie using Steven's Power Law) and within subject comparisons of perceptual sensitivity [167]

Owing to his studies of perceived exertion, Borg's Rating of Perceived Exertion (RPE) scale was devised to create a linear relationship between heart rate and exercise intensity for work on a cycle ergometer [168,169] The scale consists of both numerical values and adjective-adverbial expressions, to allow easy interpretation of the meaning of a rating value. The RPE scale has been widely used in clinical studies to predict heart rate. However, each individual works within their own absolute range of perception, thus inter-individual studies of perceptual sensitivity are not possible.

Addressing the above issues concerning the limitations and dvantages of category v ratio scales. Borg has developed a simple category scale incorporating the positive attributes of a general-ratio scale, yet the simplicity of a category scale [170,171]. Numbers from 0 to 10 are anchored by verbal expressions, whose simplicity and clarity are easily understood by test subjects. These expressions are placed in the correct position.



- 10 MAXIMAL
- 9 VERY VERY SEVERE (almost maximal)
- 8
- 7 VERY SEVERE
- 6
- 5 SEVERE
- 4 SOMEWHAT SEVERE
- 3 MODERATE
- 2 SLIGHT
- 1 VERY SLIGHT
- 0.5 VERY VERY SLIGHT
 - 0 NOTHING AT ALL (just noticeable)

Figure 6 1 Modified Borg Scale

on a ratio scale, according to their qualitative meaning [fig 6.1] The resultant modified Borg Scale permits inter-subject comparisons yet has been proven to produce psychophysical functions approximating those obtained with magnitude estimation [156,167,171,172,173]

6 3 Respiratory Psychophysics

Progress in the study of respiratory sensations has developed slowly. Experiments relating to perceived respiratory sensation have met with methodological and conceptual difficulties, including controlling and identifying the appropriate stimulus parameters and selecting the appropriate scaling technique to quantify perceived sensations [174]

6.3.1 Threshold Load Detection in Normal Subjects

The first studies directly related to perceived respiratory sensation were not carried out until the early 1960's. These studies focused on "indirect" measurements of respiratory sensations, specifically the JND, or Difference Threshold concept, based upon Weber's Law, Δ S/S = k, which states that the change in stimulus intensity needed to produce a perceived JND (Δ S) is a constant fraction (k) of the background stimuli (S) [160]

Resistive loads to either inspiration, expiration or both, alter normal pressure-flow relationships and are utilized in studies of normal subjects to mimic the type of flow limitations experienced by patients with obstructive lung disease. Conversely, elastic loads, which alter normal pressure-volume relationships mimic the type of volume restrictions characterized by patients with restrictive lung diseases [159]

Early studies of respiratory perceptions focused on the ability of man to detect a JND when small resistive and elastic loads were added at quiet breathing. These investigators measured the added elastic, ΔE_{50} , or resistive load, ΔR_{50} , which elicited detection in 50% of the loading trials (threshold detection level) and confirmed that man is able to consciously detect the addition to breathing of both inspiratory elastic [175] and inspiratory and expiratory resistive loads [176]

Studies addressing resistive load threshold detection in sitting and supine positions in normals demonstrated that although the ΔR_{50} was increased in the supine position,

Weber's fraction, defined as the ratio of ΔR_{50} to the total initial background resistance, R_0 (eg. of the breathing circuit and the subject's pulmonary resistance) was found to remain a constant [177]. A study addressing the effects of timing, flow and lung volume on load threshold detection in normal subjects demonstrated that resistive threshold detection is subserved by the relationship between pressure and flow over the early part of inspiration and neither increasing lung volumes or increasing background loading caused a decrease in resistive load threshold detection [178]. Also, other investigators demonstrated that just detectable loads were detected very early in the inspiration, close to peak flow [179,180].

6.3.2 Threshold Load Detection in Patients with Asthma and COPD

Studies on resistive load detection have been carried out in both asthmatics and patients with COPD. It has been demonstrated that asthmatics demonstrate a higher than normal ΔR_{50} , yet comparable Weber fractions when compared to normal subjects. Studies in patients with COPD, however, have yielded conflicting results. Wiley *et al* [177] found that although the ΔR_{50} was higher in a patient with COPD, the Weber fraction was similar to normal subjects. However, Gottfried *et al* [181,182] found that in a larger group of COPD subjects, both the resistive load detection threshold and the Weber fraction were significantly higher when compared to normals

6.3.3 Mechanisms of Threshold Load Detection

Later studies have attempted to elucidate the mechanisms underlying resistive load detection. It has been demonstrated that threshold detection of added resistive loads was similar before and after vagal block, excluding the role of the vagus in resistive load detection [183]. The role of the airways in resistive threshold load detection was excluded through demonstrating normal resistive threshold detection when the loads were added via a tracheostomy [184,185], normal resistive and elastic load detections following anaesthesia of both the upper and lower airways [186,187], and through the reduced sensitivity for load detection in patients with airflow obstruction [177,181,182]. Resistive load threshold detection was also normal in patients with complete cervical cord

transection [185,188], with spinal anaesthesia of the thoracic cord (chest wall block) [189] and during partial curarization to weaken the respiratory muscles [190] During assisted mechanical ventilation, a deterioration in the ability to detect added resistive loads occurred [191] Based on the above, it was hypothesized that resistive load detection is in part due to afferent information generated by the inspiratory muscles and that load detection was mechanically related to the relationship between force and displacement and not merely a function of either force or displacement alone [175,176,192] and that active respiratory muscle contraction plays an essential role in the detection of added resistive loads [191,193] It has been suggested that patients suffering from COPD may have a defect in resistive-load detection (ie perception is blunted) [181,182]

6.3.4 Threshold Detection of Ventilation and Pressure

Further studies addressing ventilation and pressure threshold detections have demonstrated the ability of man to detect increases in ventilation [194], and to detect negative pressure changes at the mouth [195] Since passive ventilation and chest vibration significantly increased the pressure load detection thresholds, it was concluded that afferent information generated from actively contracting inspiratory muscles was a necessary component for pressure and ventilation threshold detection

6.3.5 Direct Scaling of Perceptions of Volume, Ventilation, Frequency and Pressure

The first attempts to apply direct scaling techniques to respiratory studies utilized open magnitude estimation and magnitude production to investigate the respiratory perceptions of ventilation, volume and pressure changes – It was established that all of these sensations, regardless of the scaling technique utilized, fit Stevens' psychophysical power law model [196]

Further studies utilizing the direct scaling method of magnitude production and volume matching confirmed the power law relationship or subject's ability to perceive tidal volume, inspiratory flow, minute ventilation and respiratory frequency, with perceptual sensitivities (n) >1 [197,198,199] When comparing individual subjects, a large range of exponents exist for each measured sensation [196,197,200] although the

sens fivity of an individual to respiratory variables tends to be similar for each variable (egsubjects with a high sensitivity to Vt also tend to have a high perceptual sensitivity for pressure or ventilation scaling) [197] One group of investigators determined that the exponent for minute ventilation was significantly higher than either tidal volume or frequency, leading them to conclude that the assessment of ventilation probably provides the most significant respiratory sensation [196] while other investigators found no statistical increase in the exponent for ventilation when compared to Vt or inspiratory flow [197]

6.3.6 The Perception of Achieved Force (Tension) and Pressure

Studies on the perception of force and pressure have utilized the direct scaling method of magnitude estimation, where subjects are asked to estimate the magnitude of their perceived sensation of load elicited by either the addition of a resistive or elastic load at the mouth. Loading inspiration consequently increases the respiratory effort required to maintain a given level of ventilation and causes pressures (measured as airway pressure at the mouth) generated by the inspiratory muscles to subsequently increase to match ventilatory demand, as described by the equation of motion,

 $Pmus = Vt X E_0 + V X R_0 [201]$ (2)

whereby:

Pmus = pressure developed by the inspiratory muscles

Vt = tidal volume

 $E_0 = elastance$

V = flow

 $R_0 = resistance$

It has been demonstrated that the perceived magnitude of added loads to breathing in normal subjects and patients with COPD grows as a power function of the added load [181,202] Within subjects, the rate at which the sensory magnitude increased was highly correlated during both elastic and resistive loading, but again, great inter-subject variability existed. This indicates that some individuals were more "sensitive" perceiving the added loads than others [202]

It was also demonstrated that fixing flow rates during resistive loading and volumes during elastic loading caused the perceived magnitude of the loads to increase. It was concluded that either the actual tidal volume and flow rates adopted or factors related to them, such as muscular effort, were important attributes of the sensation elicited by loading breathing [202]

Further investigation of load sensation in normals confirms that the perceived magnitudes of added resistive and elastic loads at rest and during increased ventilatory drive (by exercise, CO2-stimulated breathing and hypoxia) is a direct function of peak airway pressure during inspiration (an index of changes in respiratory muscle tension) generated to overcome the load and an indirect function of the actual added resistance or elastance or combination of both (at comparable inspiratory pressures) [203,204,205] Also, it has been observed that perceptual magnitude of added resistive and elastic loads increases as the inspiratory duration of pressure development (during loading and static inspiratory manoeuvres) increases [204,206]

6.3.7 Magnitude Estimation of Loads in COPD Patients

A study examining the perceptual sensitivity of a group of COPD patients found that the exponent of magnitude estimation to added resistive loads was significantly lower when compared to normals and that adjustment for background airway resistance did not improve the measured sensitivity [182] A group of asthmatic patients, however, demonstrated similar sensitivity when compared to normal subjects. Patients suffering from COPD have long term, intrinsic resistive loading of their respiratory systems, and consequently are already experiencing chronic background resistive loading. In contrast, asthmatic patients only experience periodic increases in background resistive loading during acute exacerbations of the disease Therefore, it was hypothesized that COPD patients have developed blunted perception to increasing resistive loads [182] and that the mechanism responsible for load detection can be altered by the loading history of the respiratory muscles [207] Moreover, COPD patients tend to be older and it has been demonstrated that age does play an important role in the perceived sensation of added loads Perceptual sensitivity of added elastic and resistive loads is less in older subjects, when compared to younger normals [208,209,210] In contrast, the perceived sensation of respiratory muscle force during static inspiratory pressure manoeuvres does not differ between ages [211]

The early study of the perceptual sensitivity of COPD patients to resistive loading [182] did not take into account age, however, later studies using COPD patients and agematched normals have shown conflicting results. One study showed that when COPD subjects are compared to age-matched normals, the perceived magnitude of added loads were similar [210], whereas another study demonstrated that even after correction for inspiratory durations, COPD subjects demonstrated lower sensitivity to added elastic and resistive loads when compared to age-matched normals [212]

6.3.8 Mechanisms of the Perceived Magnitude of Volume and Added Loads

It has been suggested that the sensation of volume is mediated by afferent information generated by receptors affected by active respiratory muscle contraction [197] Many factors influence the sensory magnitude of added loads to breathing, however, the pattern of breathing adopted in overcoming the load (ie: the combination of inspiratory flow and time with resistive loading, and the combination of tidal volume and time with elastic loading) [204,210] and the age of the subjects [208,209,210] appear to play roles. These authors concluded that the blunted perceptions to added loads observed in COPD patients was due to their adopted breathing patterns of low inspiratory flow rates and shorter inspiratory duration times, compounded by their increased age. Both of these mechanisms act to lower the perceived sensation of their disease-added intrinsic resistive load [210]. This conclusion is disputed by those investigators who demonstrated that neither age nor reduced inspiratory durations could explain the abnormalities in load sensations observed in COPD patients [212].

During a magnitude production study of respiratory force at varying lung volumes, it was determined that the sensation of respiratory muscle force is based primarily on signals related to tension developed by the actively contracting muscles [203]. Resistive and elastic loading studies in low cervical cord transections demonstrated that their sense of loading was impaired, supporting the hypothesis that the sense of force is in part due to afferent signals (tension) from rib cage muscle receptors during active breathing [213]. However, it is believed that under conditions of extreme inspiratory muscle faligue or weakness, whereby afferent information from the muscles is greatly impaired, subjects associate the perception of force with the sensation of innervation associated with the outgoing motor command to the muscles from the central nervous system [190,214]

6.3.9 Effects of Background Loading on Magnitude Estimation of Resistive Loads

It has been shown in normal subjects that the sensitivity of applied resistive loads increased when background resistive or elastic loads were applied, measured by cross modality matching and nagnitude estimation. However, adjustment of the magnitude of the loading stimulus, by subtracting either the actual added background resistance or the increase in pressure necessary to overcome the background load, resulted in similar responses to loading as with non-background loaded conditions [215]. Other investigators also observed that the perceptual performance of magnitude estimation of resistive loads was not altered by background loading in normal subjects [216]. It was demonstrated, however, that short term exposure to extreme background loads in normal subjects resulted in adaptation of the mechanisms subserving load perception. The subject's response (n) was not changed but the actual intensities of the added loads were perceived to be less than during non-background loaded conditions [207].

6.3 10 Perception of Force Versus Perception of Effort

There is debate as to whether the perceived magnitude of added loads is related to the sense of force/tension or to the sense of effort required to overcome the load

It has been shown that subjects are able to distinguish between the sense of force and sense of effort, indicating two distinct sensory mechanisms. A subject's sense of force/tension is their perceived magnitude of the actual amount of force/tension being developed by the muscles to overcome a given load. Their sense of effort, meanwhile, is their measurement of how much actual effort or work they are exerting to overcome that same load. It also appears that when subjects are asked to rate perceived magnitude of load, they are actually rating their perceived sense of force and not effort. This is supported by studies whereby the perceived magnitude of an added load or of a static pressure manoeuvre is not increased as lung volumes are increased [203,206]. At higher lung volumes, more motor output command to the muscles is necessary to achieve the required tension (since the muscles are shortened). In fact, in one of the above studies, the subjects stated that the effort required to produce the target pressures was markedly increased at increased lung volumes [206]. Also, subjects suffering from inspiratory muscle fatigue have demonstrated the ability to clearly distinguish and estimate the perceived magnitude of tension and effort during static maximum inspiratory pressure (MIP) manoeuvres. The perceived tension decreased with fatigue whereas effort increased in studies where the inspiratory muscles were weakened by fatigue [214]. Killian *et al* [217] demonstrated, utilizing open magnitude estimation and the Borg scale (category scale), that the sense of effort and sense of tension both increase with increasing inspiratory pressure generation, however only the sense of effort was increased with increasing lung volumes.

These authors also addressed the sensation of dyspnea or breathlessness. They found that breathlessness, like effort sensation, was increased with increasing pressure generation at higher lung volumes. This study was important in the development of the concept that dyspnea is related to the conscious sensation of the outgoing motor command to the inspiratory muscles and is mediated by the same mechanism which subserves effort sensation.

CHAPTER 7 DYSPNEA

Dyspnea is a medical term used to characterize the non-specific complaint of difficult or laboured breathing. It is often the primary symptom of patients suffering from respiratory disease. Although a symptom, dyspnea is also a subjective sensation, and many authors have used such terms as breathlessness and shortness of breath synonymously [159,218]. One of the major problems with addressing the issue of a specific definition for dyspnea is that the exact stimulus for this sensation is unknown [219]. Numerous authors have proposed definitions of dyspnea, but as of yet, no clear and unanimously accepted definition exists [Table 7.1].

Table 7.1 Definitions of Dyspnea Proposed by Various Authors

- 1 The consciousness of the necessity for increased respiratory effort [220,221]
- 2 A sensation of laboured or difficult breathing [222,223]
- 3 Pathological breathlessness [224]
- 4 Undue awareness of breathing or awareness of difficulty in breathing [218].
- 5 Conscious awareness of outgoing motor command to the inspiratory muscles [225].
- 6 Quantitative non-threshold sensation of the mctor effort required of the respiratory muscles [201]
- 7 Increased effort in act of breathing [226]
- 8 Inspiratory effort sensation [217]

7.1 Methods Used to Quantify Dyspnea

Clinical dyspnea ratings (ie the Baseline Dyspnea Index, the Transition Dyspnea Index [227], the Medical Research Council Scale [228]) are indices often used in the hospital setting to evaluate the functional impact of dyspnea on patients' activities of daily living. However, the contribution of both emotional and psychological factors play a major role in a patient's subjective response [229]. Psychophysical tests provide different results in dyspnea perception when compared to clinical dyspnea ratings in the same subject group [230]. Clinical dyspnea ratings are mainly used as a monitoring tool of the impact of chronic dyspnea on patient's daily life, whereas psychophysical testing addresses the perception of acute dyspnea, induced in the laboratory setting

In psychophysical testing, the intensity of dyspnea is directly related to measures of ventilation, respiratory muscle contractility and force output, and cardiovascular performance, therefore providing an objective measure of the contributing physiological mechanisms involved in the perception of dyspnea [230,231]

7.2 Psychophysical Testing and Dyspnea

Open magnitude scaling [165], the Borg scale [166,232] and the VAS [165,166,232] have all been used to quantify the perception of dyspnea during a variety of methods of ventilatory stimulation. Ventilatory stimulation has been achieved through hypercapnia [165], hypoxia [165], added resistive loads to breathing [233] and exercise [165,166,232,233]

7.2.1 Dyspnea During Hypercaphic and Hypoxic Ventilatory Stimulation

It has been hypothesized that afferent signals arising from both central and peripheral chemoreceptors in response to changes in the PO2, PCO2 or hydrogen ions in the blood may be responsible for the perception of breathlessness. The chemical stimulation of ventilation through hypercapnia or hypoxia has been shown to induce breathlessness [222,234,235]. However, it has been concluded that with hypercapnic stimulation, increased breathlessness cannot result from the straightforward perception of afferent information arising from arterial chemoreceptors. Given that a clear and reproducible relationship of dyspnea to increasing ventilation due to chemical stimuli has been demonstrated [236,237], it has been suggested that dyspnea perception is related to the effective ventilatory response [233,238].

The effects of hypoxia on breathlessness remains more controversial. Some investigators have observed that the reduction in breathlessness produced by preventing exercise desaturation in COPD patients. 1 was in proportion to the reduction in ventilation [239], 2 could not be explained solely by the accompanying decrease in ventilation [240]; and in contrast, 3 that an increase in breathlessness was observed in relation to ventilation when breathing supplemental oxygen [241].

7 2.2 The Role of Vagal Afferents in the Perception of Breathlessness

Pulmonary afferent neural information has been hypothesized to play a direct role in the mediation of dyspnea [222,242] This has been based on studies whereby vagal blockade has led to a reduction in shortness of breath in patients with pulmonary disease [243,244] However, recent studies in patients with heart-lung transplantation (pulmonary denervation) supports the hypothesis that vagal mechanisms contribute to dyspnea only indirectly, through the modulation of the pattern of ventilation, thus influencing the pattern of respiratory muscle activation [222,245]

7.2.3 Exercise Testing and Dyspnea

Exercise offers several advantages in the study of dyspnea when compared to other artificial means of stimulating ventilation (ie hypercaphia and hypoxia). It directly simulates demands placed upon the inspiratory muscles (increasing force and effort requirements) which an individual is faced with during daily physical activities [219,246]

7.2.3.1 Dyspnea as a Function of Workload

During progressive exercise testing, normal subjects exhibit an absence of breathlessness at very low levels of work (measured in kpm/min or watts). Dyspnea, however, increases in a curvilinear fashion as exercise progresses, demonstrating a rapid increase and reaching values between 7 and 10 as rated on the Borg scale near maximal exercise [233,247] The addition of resistive inspiratory loads in normal subjects increases the perceived level of dyspnea for a given level of work and with increasing loading, exercise capacity is diminished [233]

In a group of cardiorespiratory patients during a progressive exercise test, breathlessness has also been shown to systematically increase with increasing levels of work, with dyspnea ratings on the Borg scale of severe (5) to maximal (10) at maximum exercise. However, a wide inter-subject variation was shown to exist in the rating of dyspnea for any given level of work. Some patients presented with a similar response to exercise as normals, with no dyspnea perceived at rest or for low levels of work. However others presented with dyspnea at rest and more rapid increases in perceived dyspnea with

increasing work It was observed that a wide variability still existed when dyspnea was plotted as a function of workload, normalized for the predicted maximum [248]

7.2.3.2 Dyspnea as a Function of Ventilation

Numerous authors have demonstrated that the intensity of dyspinea progressively increases with increasing levels of minute ventilation (VE) during progressive exercise, in both normal and COPD subjects [165,240,249,250,251,252,253,254] COPD subjects, however, experience higher degrees of dyspinea for a given level of VE when compared to normal subjects

The rationale for using VE is that it represents the total output from the respiratory system and correlates significantly with ratings of breathlessness during exercise [219] Using both the VAS and Borg scale, the intensity of breathlessness at comparable levels of ventilation has been shown to be quite reproducible within a subject upon repeat testing [165,232]. However, inter-subject variation exists in the perceived dyspinea for any given level of absolute ventilation, both in normal [222,245] and COPD subjects [248]. It is believed that it is not simply the level of ventilation, though, which determines the intensity of the dyspinea perceived, but rather how much of the total ventilatory capacity is being utilized. This idea has led investigators to normalize VE by the maximum voluntary ventilation (MVV). Investigators have utilized both the true measured MVV, calculated through pulmonary function testing [255], or a predicted value for the subject's MVV, based upon the equation.

$MVV = FEV_1 \times 35 \quad [245, 248, 256, 257] \quad (1)$

A recent study, however, demonstrated quite a large variability in the MVV/FEV₁ ratio between subjects (both normals and patients) and demonstrated that

$MVV = FEV_1 \times 42$ [258] (2)

represents a more average value for predicting the MVV across a wide range of subjects. There have been conflicting results when dyspnea has been expressed as a function of VE/MVV. Some authors demonstrated that no difference existed between normal and COPD subjects' perceived difficulty of breathing when VE was expressed as a fraction of the MVV [203]. Others have shown that in both normal subjects, heart lung recipients [245], and in a group of cardiorespiratory patients [248], a wide inter-subject variation existed in the dyspnea response to VE/MVV

Others have expanded further to suggest that breathlessness is not simply the sensing of the actual ventilation achieved but other mechanisms such as fatigue may play a role in the perception of dyspnea [259]

7 2.3 3 The Effects of External Loads on Dyspnea Perception

External resistive loads have been added to inspiration in normal subjects during progressive exercise to simulate airway obstruction. Attempts to determine the effects of inspiratory loading on breathlessness have met with conflicting results. Some investigators have shown that the addition of an inspiratory load during a progressive exercise test significantly increased breathlessness for a given ventilatory level [233,251,260]. It was also observed that the threshold of dyspnea perception is lowered and the slope of the response of dyspnea to VF is increased with added inspiratory loads. Other investigators, however, observed no significant increase in breathlessness with the addition of low level inspiratory loads during a constant workload exercise test [261]. The differences probably lie in methodology. A progressive exercise test causes much higher achieved ventilation when compared to steady state exercise, and this added ventilatory stimulus may be necessary to observe the increases in dyspnea associated with the addition of inspiratory loads.

It has been hypothesized that subjects' rated levels of breathlessness is also based upon their past experiences of their relationship between breathlessness and ventilation (eg. loading breathing and then removing the load) [260]

7.2.34 Dyspnea as a Function of Pressure Development

Mouth occlusion pressure (P₁₀₀) has been used as a measure of respiratory efferent activity during ventilation. In order to correct for differences in maximum respiratory muscle strength and contractility, mouth pressure is expressed as a percentage of the maximum generated inspiratory airway pressure at FRC, with the airway occluded (MIP) [222]. Again, results have been conflicting. One study demonstrated that during progressive treadmill exercise testing, both with and without inspiratory resistive loading, the relationship between the intensity of breathlessness and normalized occlusion pressure is the same in COPD patients as in normal subjects [222]. This led to the conclusion that at a given level of respiratory motor activity, the intensity of breathlessness does not differ from normals to patients with COPD. However, a more recent study demonstrated that although a direct linear relationship existed between dyspinea and $P_{0.1}$ /MIP in all subjects, a wide variation in the responses (slopes of the relationships) existed [245].

Studies have also demonstrated a significant correlation between peak inspiratory mouth pressure (Pm) and perceived breathlessness, both in a group of normal subjects [217] and asthmatic patients [219]

A close relationship between dyspnea and inspiratory effort sensation (IES) is assumed [262], therefore IES and dyspnea are often used synonymously by investigators studying the mechanisms of dyspnea

Esophageal balloons can be utilized to measure esophageal pressure (Pes), which approximates pleural pressure [263] Studies in normal subjects have demonstrated a direct linear relationship between pleural pressure swings (Δ Pes), Δ Pes normalized as a percentage of the maximum inspiratory pressure (Δ Pes/MIP) [264], and mean Pes (Pes) [265] to the subjects' IES – A large variability in subjects' sense of effort is observed when IES is related to the pleural pressures generated. This variability was minimized when Pes was expressed as a percentage of the MIP [233]

A similar linear relationship of Pes/MIP and Pm to breathlessness, as rated on the **Borg scale**, has been shown in both cardiorespiratory and asthmatic subjects [248,219]

The MIP has generally been utilized as an index of respiratory muscle strength (RMS). In normal subjects, a large variation in MIP has been observed, 60 cmH20 to 300 cmH20. COPD patients also exhibit a wide variability in RMS (i.e. a mean value of 66 cmH20 ±20 cmH20 in a group of cardiovascular and respiratory diseased patients) [201]

Variations in RMS have been demonstrated to result in an equally large variability in IES during resistive loading in both normal and COPD subjects. The sense of effort for a given pressure generated was greater for the subjects with weaker inspiratory muscles [201,246,262] The majority of all studies have utilized MIP as the normalizing value for pleural pressure swings, however, a recent study addressed the fact that the capacity of the inspiratory muscles to generate pressures decreases with both increasing lung volumes (used as an indication of changes in respiratory muscle length) and increasing inspiratory flow rates (used as an indication of changes in the respiratory muscle velocity of contraction) [247,266,267,268,269]

This concept has led to the quantification of the capacity of the inspiratory muscle to generate pressure (Pcap) during exercise. It was found that Pcap is preserved between 30 and 55% TLC, but declines by 17% for each 1% increase in end inspiratory lung volume above 55% TLC. In addition to this volume effect, a 5% reduction in the muscles' generating capacity was observed for each litre per second increase in flow at any given lung volume. Therefore, normalization of pleural pressures by MIP will underestimate the percentage of capacity being utilized by the respiratory muscles during exercise. The following equation quantifies the relationship,

 $P_{cap}(\% \text{ max}) = (MIP - 1 7(\% TLC)) - 50 (I/sec) [247]$ (3)

These investigators found that when breathing was loaded during exercise, the sense of respiratory effort was closely correlated to the pressures generated by the respiratory muscles expressed as a percentage Pcap [247,270]

72.3.5 Dyspnea as a Function of other Physiological Parameters

A study in normal subjects undergoing exercise testing has demonstrated a significant and independent relationship of breathlessness (ψ) to peak Pes (indication of muscle tension), inspiratory flow (\dot{V} i) (an indication of the velocity of inspiratory muscle shortening), the frequency of breathing (fb) and the duty cycle (Ti/Ttot), giving the predictive equation

$$\psi = 0.11(\text{Pes}) + 0.61(\text{VI}) + 1.99(\text{Ti/Ttot}) + 0.04(\text{fb}) - 2.60$$
 [233] (4)

Similar studies in COPD patients during exercise have also demonstrated a multivariable relationship to breathlessness. In a group of cardiorespiratory patients, the following predictive relationship for breathlessness was found: ψ = 3.0(Pp1/MIP) + 1 2(Vi) + 4 5(VVC) + 0 13(fb) + 5 6(Ti/Ttot) - 6 2 (4)

Similarly, in a study of asthmatic patients, VI, VT/FVC, fb and Pm each individually demonstrated linear relationships to subjects' levels of breathlessness as rated on the Borg scale and all variables together accounted for 63% of the variance in the rating of dyspnea [219].

7.3 The Relationship of Dyspnea to Muscle Fatigue

The role of inspiratory muscle fatigue in the development of dyspnea remains an unresolved issue. Studies have largely focused on IES during induced fatigue in normal subjects. Diaphragmatic fatigue, induced through inspiratory resistive loading, has been shown to be associated with increases in the sense of respiratory effort. However, both the intensity of the diaphragmatic contraction and changes in diaphragmatization levels were unrelated to the magnitude of the sensory perception [271]. A further study whereby diaphragmatic fatigue was induced through fatiguing contractions of the diaphragmatic fatigue was induced through fatiguing contractions of the diaphragmatic fatigue was induced through fatiguing contractions of the diaphragmatic between the levels of activation of both the stemomastoid and rib cage muscles and the sense of respiratory effort, leading to the conclusion that the increasing effort sensation during a diaphragmatic fatigue run was due to either the loading of the rib cage muscles or to the central perception of an overall increase in the central motor output directed towards these "accessory" muscles [271]

Other studies have induced generalized inspiratory muscle fatigue (IMF), both through continuous maximal inspiratory contraction [273] and by inspiratory threshold loading [274]. It has been demonstrated that a faster rate of increasing IES is observed with a high pressure (Pm/Pmax)-short duty cycle (Ti/Ttot) pattern of loaded breathing when compared to a low pressure-long duty cycle pattern, despite the maintenarice of a constant pressure time index of 0.24. This led to the hypothesis that IES is related to but not a direct function of the onset and severity of IMF. IES is therefore dependent on the intensity rather than the duration of muscle contraction and may possibly occur as a result of the central perception of the subsequent increasing motor command to the fatiguing muscles.

[273,274] Varying the breathing patterns (ie predominantly rib cage, diaphragm or a combination of both) during inspiratory resistive loading in normal subjects resulted in an observed direct linear and unique relationship between Δ Pes and IES, regardless of the thoracoabdominal breathing pattern adopted [265] Although imposing fatiguing diaphragmatic patterns of breathing did demonstrate a strong correlation between IES and Pes/Pesmax, IES was independent of the development of diaphragmatic fatigue [264].

It is important to note that a study addressing the perception of dyspnea associated with changing levels of ventilation demonstrated that dyspnea intensifies when the level or pattern of breathing is voluntarily changed from the spontaneously adopted level [275]. Therefore, this may compound the dyspnea which subjects perceive during studies where their breathing patterns are set or constrained.

Studies of diaphragmatic fatigue and IES in COPD subjects are limited, nevertheless, it has been demonstrated that, as in normal subjects, diaphragmatic fatigue appears independent of IES [276] Also controversy exists as to whether COPD patients develop respiratory muscle fatigue during progressive exercise. It has been observed that COPD subjects often develop a fatiguing pattern of inspiratory muscle contraction and/or diaphragmatic contraction during exercise [115,277,278, 279,280]. In contrast, non-fatiguing patterns of both inspiratory muscles (p.eural pressure-time indices (PTI) of 0.10) and the diaphragm (PTI of 0.12) have also been measured upon the cessation of exhaustive exercise [281].

Utilizing fatiguing patterns of PTI to draw conclusions regarding the relationship of IES to fatigue must be interpreted cautiously. What is usually being indicated is incipient fatigue, whereby the muscles would fatigue should that pattern of contraction continue for a given length of time. For example, a diaphragmatic PTI of 0.15 could be sustained for approximately 90 minutes, whereas an inspiratory muscle PTI of 0.25 could be sustained for approximately 15 minutes before signs of overt fatigue would develop [272]. Also, subjects may cyclicly alternate between diaphragmatic and inspiratory muscle contraction in an attempt to limit incipient fatigue [106]. It appears that normal subjects adopt a pattern of breathing which serves to optimize the sense of effort by minimizing the tension generated in the respiratory muscles [270].

Other observed signs of incipient fatigue (during observed fatigue (p. PTI) include a fall in the high to low ratio or centroid frequency of the EMG power spectrum, the slowing of observed inspiratory muscle relaxation, rapid shallow breathing and chest-abdomen asynchrony [282]

COPD patients often demonstrate respiratory muscle weakness and subsequent decreases in their inspiratory muscle force response after exercise (ie decreasing sternomastoid force [279]), however, one cannot assume that overt muscle fatigue has occurred. They too will attempt to adopt a pattern of breathing to minimize the risk of IMF.

7.4 Mechanisms of Augmentation and Relief of Dyspnea In COPD Subjects

It has been shown that in patients experiencing acute asthma attacks, dyspnea is well correlated with both hyperinflation and recruitment of the sternomastoid muscles, and that dyspnea subsides when the heightened sternomastoid activity ceases [283] Patients suffering from respiratory failure experience a loss of dyspnea when mechanically ventilated, a loss which correlates with a cessation of diaphragmatic electrical activity [284] During abdominal strapping, COPD subjects experience increased levels of dyspnea during exercise, hypothesised to be caused by a shortening of the intercostal and neck muscles and the subsequent increase in the required motor command [285] COPD patients often experience a relief in dyspnea by assuming the leaning position (supporting their upper extremities) Postural relief of dyspnea is likely due to the fact that leaning forward both improves the length-tension state and increases the efficiency of the diaphragm as well as the unloading of the rib cage accessory muscles [286]. It has been observed that in a patient with bilateral diaphragm paralysis, acute dyspnea was experienced when the subject was immersed in water to the level of the abdomen. Since the water now fully supported the abdominal contents, immersion prevented the passive inflationary effects of abdominal muscle contraction and eliminated inspiratory abdominal muscle relaxation This resulted in a loading of the rib cage muscles [287] All of these studies serve to emphasize the apparent significance of respiratory muscle function to the perception of dyspnea

CHAPTER 8 METHODOLOGY

It is hypothesized that high levels of ventilation should be reached at lower perceived levels of dyspinea if respiratory muscle coordination increases and the use of the diaphragm-accessory muscles is optimized. Successful athletes ann/or COPD patients should therefore display high levels of ventilation at lower dyspinea. Hence we tested muscle coordination and its relationship to dyspinea in groups of athletes, normal sedentary subjects and COPD patients during high levels of exercise. Dyspinea was measured using the Borg scale. Coordination was inferred via esophageal and gastric pressure swings and chest well-abdominal displacement during a progressive exercise test on a cycle ergometer.

The rationale underlying this hypothesis is fourfold. First, dyspnea is a quantifiable sensation which is known to be proportional to the degree of inspiratory muscle activity. Second, COPD patients do have an increased load to breathing at rest as well as at lower ventilatory levels due to the nature of their disease. Similarly, athletes also are exposed to higher respiratory loads on a regular basis when performing training which requires high levels of ventilation. Finally, sedentary normal subjects are not exposed to either of these types of loading on a regular basis.

In order to investigate this hypothesis we tested fifteen male subjects. All subjects gave informed consent prior to their participation. Four of the subjects were normal, sedentary, non-smokers and were familiar with respiratory studies. Four were defined as elite athletes, having competed in their sport internationally and who underwent daily endurance training (3 rowers, 1 cyclist). The remaining seven subjects were classified as moderath to severe COPD patients.

Prior to the study, all subjects performed standardized pulmonary function tests, using a compact Vitalograph spirometer, to determine their forced vital capacity (FVC) and forced expired volume in one second (FEV_1). The relationship of FEV_1/FVC was thus calculated. All studied patients were currently being followed at Notre Dame Hospital, Montreal, Quebec and had undergone extensive testing in the pulmonary function laboratory prior to their participation in this study. Values of Total Lung Capacity (TLC), Functional Residual Capacity (FRC) and Residual Volume (RV) were obtained from the patients'

charts. Predicted spirometric values were obtained from standardized tables, based on the subject's sex, age and height [288,289]

Immediately prior to exercise testing, patients were sent for an electrocardiogram, whose results were verified by a physician to ensure their suitability for exercise testing

Subjects were seated on a cycle ergometer (N V. Gorlart, Holland #116473) where a metal bar with chest pads was adjusted and placed in front of the subject, enabling them to lean against it for improved stabilization and minimization of body movement and postural changes. The handlebars of the ergometer were also stabilized to decrease extraneous body movement [fig 8.1].

Subjects were fitted with a headpiece to which a large two way valve was attached (Hans Rudolph Inc. Kansas City, No. 2700) A pneumotachograph (Fleisch no. 3) was connected to the inspiratory port. A rubbel mouthpiece was connected to the valve and inserted into the subject's mouth. The pressure gradient within the pneumotachograph was measured via a Validyne pressure transducer (+2 cmH20) For monitoring purposes, flow was electronically integrated to volume and displayed on paper utilizing a Hewlett-Packard respiratory integrator (8815A) Signal calibration was performed to minimize drift of the flow signal and the inspiratory flow was set to automatically reset to 0 at the end of each inspiration Esophageal pressure (Pes) (measured as an estimation of pleural pressure) and gastric pressure (Pga) were measured [129,263,290] using commercially available latex rubber balloons, 5 cm long, 0 012-0 015 cm in thickness, 3 50 cm in circumference and 1.11 cm in diameter (Medical Corporation, New Jersey), attached to 100 cm Intramedic PE 200 non-radiopaque, non-toxic polyethylene tubing with a luminal diameter of 1.40 mm, and outer diameter o. 1.90 mm (Clay Adams, Division of Becton Dickinson and Company, New Jersey) The catheters were connected to separate Validyne pressure transducers (±200 cmH20 for measuring esophageal pressure changes and ±250 cmH20 for measuring gastric pressure changes) All pressure signals were demodulated and amplified by Hewlett-Packard carrier amplifiers (8805B) Both balloons were inserted via the nose into the stomach Placement of the esophageal balloon was performed by slowly retracting the balloon as the subject sniffed. When the pressure changes became negative in direction as opposed to positive and a baseline resting negative pressure was observed [291],



Figure 8.1 Experimental setup showing a subject seated and secured to the cycle ergometer, with mouthpiece, pneumotach, Pes and Pga balloons and Respitrace bands in place



the tubing was securely fastened at the nose via tape and a nose clip. Care was also taken to place the esophageal balloon where the cardiac artifact was minimized. This procedure was performed by a physician at Notre Dame Hospital who had significant experience in accurately positioning the esophageal and gastric balloons.

The esophageal balloon was filled with 2 ml of air, of which 1.5 ml was then removed, leaving 0.5 ml remaining. The gastric balloon was filled with 1 ml of air. The compliancy of the esophageal latex balloon at a volume of 0.5 ml as an acceptable minimum unstretched volume was validated by the pulmonary function laboratory at Notre Dame Hospital. Normally, the gastric balloon is thicker than the esophageal balloon, however, since the gastric balloon was not left in the stomach for a prolonged period of time, there was adequate resilience to gastric secretions and thus, the utilized balloon thickness was acceptable. The polyethylene catheters were found to demonstrate a flat frequency response up to 10 Hz. Frequency spectral analysis of both Pes and Pga swings demonstrated that even during maximum exercise, no frequencies of any power produced exceeded 5.5 Hz, thus validating the usage of these catheters

Rib cage (RC) and abdominal (AB) excursions were measured via respiratory inductive plethysmography (Respitrace, Ambulatory Monitoring, Inc., White Plains, N.Y.). The upper edge of the rib cage band was placed at the mid-axillary level, about the second to fourth intercostal area, while the upper edge of the abdominal band was placed at the level of the umbilicus above the iliac crest [292]. Both bands were taped and secured in place. Respitrace signals were amplified using two Hewlett-Packard medium gain amplifiers (8802 A) for RC and AB signals respectively.

All signals were displayed on a Hewlett-Packard 8 channel paper recorder (7758A) and recorded on analog magnetic tape (Hewlett-Packard instrumentation recorder 3968A) for future playback and computer analysis

A four lead electrocardiogram was also utilized, connected to an oscilloscope for the monitoring of cardiac electrical activity throughout the exercise test for safety reasons and an attending physician was present throughout the study

All channels were calibrated prior to beginning the exercise test and calibrated again at the end of the session, to minimize error and assure accuracy. Once the setup was completed, subjects were asked to perform a static maximum inspiratory pressure manoeuvre (MIP) at functional residual capacity (FRC) with airways closed, while in the selection on the ergometer. This was repeated until consistent results were obtained

Subjects were also asked to perform a static maximum transdiaphragmatic pressure manoeuvre (Pdi_{max}), also at functional residual capacity with airways closed while seated The Pdi_{max} manoeuvre was performed as follows. Subjects were first asked to maximally activate their diaphragm-abdomen by asking them to "bear down as for defecation". While maintaining this maximal expulsive effort, subjects were asked to perform a maximal inspiratory (Mueller) manoeuvre where they vigorously attempted to suck in air against a closed glottis [293]. This manoeuvre was repeated until consistent results were obtained

The Respitrace was calibrated by having the subjects perform isovolume manoeuvres [133]

Subjects were informed that during the exercise test, they would be asked periodically to rate "how short of breath" they were on a modified Borg Scale [171], which was clearly displayed in front of them. They were informed that they could choose any number, or fraction thereof and to exclude any other sensations (i.e. limb pain) they might be experiencing when they made their rating.

A period of quiet breathing was monitored until all signals had reached a steady state. Subjects then began a progressive exercise test, with 20 watt increments in workload introduced every minute. Subjects were requested to cycle at a constant rate of 40 RPM. At the end of each minute, prior to the next increase in workload, heart rate was recorded and subjects were asked to rate their level of breathlessness.

Subjects were encouraged to continue cycling until exhaustion. After cessation of the exercise test, subjects were asked to continue to cycle at their own pace, and at the end of each minute, their level of breathlessness and pulse rate were again monitored. The test ended when both of these values reached their pre-exercise resting levels. Although the data recorded post-exercise was not included in the data analysis, the subjects performed this "cool down" segment for safety reasons.
8.1 Computer Analysis

The signals of the five recorded channels (Flow, Pes, Pga, RC, AB) were digitally sampled to a personal computer (IBM 386) using a software, acquisition package (Labdat Version 2.0), through an A\D Board (National Instruments No MC-MIO-16) and stored for future off-line analysis. The patients' data was sampled at 100 Hz, whereas the normals' and athletes' data were sampled at 200 Hz, given the extremely rapid breathing responses observed during maximum exercise. Inspiratory flow (V) was electrically integrated to volume using a software respiratory analysis program (Resp. written by Doug Wight, Notre Dame Hospital, 1991) and the transdiaphragmatic pressure (Pdi) signal was calculated by the following equation

$$Pdi = Pga - Pes [129]$$
 (1)

A breath by breath analysis was performed on five breaths of quiet breathing prior to exercise, and on the last five breaths of each consecutive minute of exercise at a given workload for each subject

The following ventilation parameters were calculated for each breath, utilizing 0 flow as the beginning of Ti and the end of inspiratory flow as the end of Ti [Figure 8 2a] inspiratory tidal volume (Vt), inspiratory time (Ti), expiratory time (Te), total time of the breath (T_{TOT}), the ventilatory duty cycle (Ti/ T_{TOT}), frequency of breathing in one minute (fb) by the following equation.

$$fb = 60 / T_{101}$$
 (2)

Also minute ventilation (VE) whereby

$$VE = fb X Vt$$
(3)

and minute ventilation normalized for lung size and disease obstruction (VE/FVC) were calculated. Each subject's individual maximum voluntary ventilation (MVV) was predicted utilizing the equation

$$MVV = FEV_1 X 42 [258]$$
 (4)

permitting the estimation of the percentage of the subject's maximum ventilatory capacity that was being utilized during exercise (VE/MVV)

Since the decline of Pes often precedes flow (pressure events can occur prior to the observation of a mechanical event), the following pressure parameters were separately



Figure 8.2 a Measured Ti and Te utilizing the onset of inspiratory flow as the beginning of Ti. This method was utilized to calculate the Ti/Ttot (duty cycle) of ventilation.



Figure 8.2.b Measured Ti and Te, utilizing the onset of the negative pleurol pressure swing as the beginning of Ti. This method was utilized to calculate the pressure Ti/Ttot (duty cycle).



Figure 8.3 The pre-inspiratory flow phase, defined as the time from the onset of the negative pleural pressure swing to the onset of inspiratory flow, and the initial phase of pressure generation, defined as the time from the onset of inspiratory flow to the peak flow acceleration, are illustrated

measured, utilizing the onset of the negative Pes swing (when simultaneous changes in Pga were observed) as the onset of the Ti related to pressure and the end of Ti as the point where inspiratory flow ends [Figure 8.2b] peak to peak pleural pressure swing (Δ Pes), mean Δ Pes (Δ Pes), peak to peak gastric pressure swing (Δ Pga), peak to peak transdiaphragmatic pressure (Δ Pdi), mean Δ Pdi (Δ PdT), Δ Pes normalized by maximum inspiratory pressure generated at FRC (Δ Pes/MIP), Δ Pes normalized by the maximum capacity pressure the inspiratory muscles are capable of producing corrected for volume and flow (Δ Pes/Pcap) which assumed a progressive 10% reduction in end expiratory lung volume throughout the exercise run in both normal and athletic subjects [247], mean Δ Pes/Pcap (Δ Pes/Pcap), the pressure duty cycle (Ti/Ttot) and the pressure-time index of the inspiratory muscles (TT_{IM}) where

$$TT_{IM} = \Delta \overline{Pes} / Pcap X TI / T_{TOT}$$
(5)

Also measured using this timing was $\Delta \overline{PdT}/PdI_{max}$, and the pressure-time index of the diaphragm (TT_{dl}) [272] where

$$TT_{di} = \Delta P d I / P dI_{max} X T I / T_{IOI}$$
(6)

Two phases of the onset of each inspiratory effort were also identified. The first was termed the "Pre-Inspiratory Flow" phase of inspiration and was defined as the segment of time between the onset of the negative Pes swing and the onset of inspiratory flow. The second was termed the "Initial Phase", and was defined as the time between the onset of inspiratory flow and the peak of the acceleration of the inspiratory flow [fig. 8.3]

Further parameters were thus calculated. These were the rate of change of Pes with respect to time for the pre-inspiratory flow phase of inspiration ($\Delta Pes/\Delta t_{PRE}$) and the initial phase of inspiration ($\Delta Pes/\Delta t$). Similarly, the rate of change of Pga with respect to time for the pre-inspiratory flow phase of inspiration ($\Delta Pga/\Delta t_{PRE}$) and the initial phase of inspiration ($\Delta Pga/\Delta t$) were calculated.

Each of the above ventilation and pressure values which were calculated for each of five breaths at the end of each minute of exercise were then averaged utilizing Sigmaplot (Jandel Scientific, Version 4.1), giving the mean \pm 1 standard deviation (sd) of each measured parameter at a given workload. An analysis of variance between the three groups was performed with a post Scheffe test. Firatios <0.05 were considered statistically

significant For graphics purposes, signals were converted from Resp files to ASCII files, analyzed in 123 (Lotus Development Corporation, Release 2 01) and subsequently plotted utilizing Sigmaplot.

CHAPTER 9 RESULTS

The results will be divided into three main sections 1 Anthropometrics, 2 Dyspnea as a function of work, ventilation and pressure generation during progressive exercise and 3 Mechanics of breathing and coordination during progressive exercise.

91 Anthropometrics

Individual subject data is provided in Table 9.1 and Table 9.2 respectively. The mean age (+1 standard deviation) of the patients was 62 ± 10 years, the mean age of the normals was 37 ± 5 years, while the mean age of the athletes was 22 ± 1 years

9.1.1 Pulmonary Function

All pulmonary function tests (PFT) of the normals and athletes were within a norm. I range when compared to predicted values. The only exception was N1 who presented with an FEV₁/FVC of 65%. When considering the normal predicted values for a male, aged 37 and 175 cm in height [289], it is found that N1's abnormally low FEV₁/FVC ratio was directly due to his extremely high measured FVC (6.15 litres versus 5.00 litres predicted). His FEV₁ was comparable to that predicted (4.0 litres)

The COPD patients presented with significant limitations in all aspects of pulmonary function. Five patients were clinically and functionally diagnosed as severely obstructed and two as moderately obstructed based upon their individual FEV_1/FVC results. The following values are an average of all seven COPD patients (±1 standard deviation). FEV_1 (% predicted) 33.8 ±15.8%, FEV_1/FVC 40.97 ±10.93%, and RV/TLC (% predicted) 184.16 ± 35.94%

9 1.2 Respiratory Muscle Strength

The mean group Maximal Inspiratory Pressure (MIP), generated against a closed glottis at FRC of both normals and athletes were not significantly different, -130 88 \pm 8 29 cmH20 and -143 13 \pm 19 08 cmH20, p>0 05. The group mean MIP value of patients was significantly reduced (-63 94 \pm 18 81 cmH20, p<0 05) when compared to both the athletic

SUBJECT	AGE	нт	WT	FEV ₁	FEV,	FVC	FEV,/FVC	RV/TLC	MIP	Pdi _{max}
	(yrs)	(cm)	(kg)	(I)	(%Pred)	(1)	(%)	(%Pred)	(cmH20)	(cmH20)
P1	49	175	91	2 30	55 40	4 27	54 00	144 17	-50 70	86 53
P2	49	167	54	0 65	17 30	2 40	27 00	252 31	-40 00	62 46
P3	73	16 <i>i</i>	58	1 07	42 30	2 51	42 60	172 19	-50 00	60 00
P4	67	177	84	1 02	26 60	2 60	39 40	167 45	-65 00	80 19
P5	70	160	71	0 72	24 90	1 83	39 50	190 19	-83 16	99 41
Pô	69	178	81	1 58	51 90	2 89	55 00	158 69	-92 30	121 8
P7	61	165	77	0 52	18 50	1 80	28 89	204 13	-66 43	68 00
Mean ±sd	62 ±10	170 ±7	73 ±13	1 12 ±0 63	33 80 ±15 80	2 61 ±0 83	40 97 ±10 93	184 16 ±35 94	-63 94 ±18 81	82 62 ±22 25

Table 9.1 Patient Data

SUBJECT	AGE	Нт	WT	FEV,	FVC	FEV./FVC	MIP	Pdimax
	(yrs)	(cm)	(kg)	(1)	(i)	(%)	(cmH20)	(cmH20)
N1	37	175	72	- 100	6 15	65 00	-138.00	194 00
N2	38	181	76	4 19	5 27	79 OO	-138 00	204 00
N3	43	176	72	4 38	5 34	82 00	-125 13	156 32
N4	30	170	63	3 97	4 57	86 80	-122 40	157 85
Mean ±sd	37 ±5	176 ±5	71 ±6	4 14 ±0 19	5 J)±0 65	78 30 ±9 40	-130 88 ±8 29	180 54 ±21 97

A1	21	188	86	5 95	7 05	84 40	-120 00	210 00
A2	21	190	85	5 46	6 70	81 50	-135 00	169 64
A3	23	176	75	4 90	5 70	86 00	-160 00	178 60
A4	23	173	70	4 76	5 71	83 40	-157 50	195 40
Mean ₊sd	22 ±1	182 ±9	79 ±8	5 27 ±0 55	6 29 ±0 69	83 80 ±1 90	-143 13 ±19 08	188 41 ±17 92

Table 9.2 Norma' Sedentary Subjects (N) and Athletes (A)

64

and normal groups. Mean group Maximal Transdiaphragmatic Pressures (Pdi_{max}) of both normals and athletes also were not significantly different, 188–41 ± 17–92 cmH20 and 180–54 ±21.97 cmH20, whereas the mean Pdi_{max} value of the patient group was significantly reduced, 82–62 ± 22–24 cmH20, p<0.05. The range of Pdi_{max} values within the patient group was 60–00 to 121.8 cmH20

9.2 The Relationship of Dyspnea to Work

9.2.1 Dyspnea as a Function of Workload (Watts)

The athletic group reached a statistically higher mean level of absolute work (measured in watts) at maximum exercise, range 320-400 watts, than both the normal group, range 200-320 watts and patient group, range 40-120 watts, p<0.05 Since the workload was increased at one minute intervals in all subjects, athletes also demonstrated much longer endurance time, range 17 to 21 minutes, when compared to the normal subjects, range 11 to 17 minutes, and the patients, 3 to 7 minutes

The mean maximum perceived dyspnea at maximum exercise as rated on the Borg scale was not different amongst the athletic group (7.50 ± 1.80), normal group (7.50 ± 1.73) or patient group (6.00 ± 1.73), p>0.05

Mean group dyspnea was examined as a function of the workload measured in watts (W) [fig 9 1.a] All groups demonstrated a curvilinear relationship of dyspnea to work There was no difference between mean group dyspnea values at quiet breathing of the athletic, normal and patient groups, p>0.05. It should be noted, though, that four of the seven patients did present with slight dyspnea at rest (2 with a dyspnea rating of 1, 2 with a dyspnea rating of 2). During the exercise test patients perceived significantly higher dyspnea than either the normals or athletes from the point when absolute workload was increased to 20 watts to the point when exercise was ceased, p<0.05. The athletes and normal subjects presented with similar dyspnea ratings from 0.W to 120.W, however, from 140.W to maximum exercise, normal subjects perceived higher levels of dyspnea than the athletic group for a given level of work, p<0.05.



Figure 9.1 a Dyspnea plotted as a function of work, measured in watts Points represent group mean values ± 1 sd. QB represents quiet breathing.



Figure 9.1 b Dyspnea plotted as a function of workload, normalized as % maximum Points represent group mean values ± 1 sd QB represents quiet breathing



Figure 9 1.c Common log of Dyspnea plotted as a function of workload, normalized as % maximum. Lines represent the linear regressions for individual subjects.

9.2.2 Dyspnea as a Function of Workload (% Maximum)

Individual subject's dyspinea was examined as a function of workload, normalized as a percentage of the maximum workload achieved by that subject [fig 9.1 b] Workload was normalized in order to examine the dyspinea response observed relative to what percentage of the individual's maximum workload capacity he was exercising at

All subjects demonstrated a curvilinear relationship of dyspnea to normalized workload. No difference existed between any groups from 0% to 100% maximum workload, p>0.05

The response rate of dyspnea to workload is defined as the rate of change of dyspnea per a given change in work. The response rate was assessed by plotting the dyspnea scores on a common logarithmic (log) scale versus the % of achieved maximum workload. A "1" was added to all dyspnea scores for all subjects to permit a logarithmic analysis. This mathematical addition would not change the slope of the dyspnea/% maximum workload relationship but permitted the logarithmic analysis of 0 dyspnea values.

The relationship of the common log of dyspnea to workload (% maximum) was linear in all subjects (means \pm 1 sd of the regression coefficients were 0.95 \pm 0.05, 0.97 \pm 0.01, 0.96 \pm 0.04 for the athletes, normals and patients respectively) [fig. 9.1.c] The slope of each individual regression line represents the response rate of dyspnea to work (% maximum). All athletes presented with similar response rates (0.83 \pm 0.09). Three of the four normals presented with very similar response rates (1.00 \pm 0.04), however one normal subject (N3) presented with a very low response rate (0.72) which was similar to the athletes. The patient group presented with a significantly lower response rate (0.57 \pm 0.18) than either normals or athletes (p<0.05).

9.3 The Relationship of Minute Ventilation and Work

9.3.1 Minute Ventilation as a Function of Work

At rest, minute ventilation (VE) was examined and no difference was found between the athletic group, 10.81 ± 3.33 l/min, the normal group, 12.89 ± 3.89 l/min, or patient group 12.70 ± 4.19 l/min, p>0.05 Initially, VE was plotted as a function of workload measured in absolute Watts [fig 9 2.a]. No difference in mean VE existed between the normals and athletes at any levels of absolute workload up to 240 watts. At extremely high levels of work, greater than 240 watts, normals demonstrated significantly higher mean ventilation levels, p<0.05. Patients presented with similar mean VE for the first five levels of measured work (0 to 80 W), however at >80 W, patients presented with significantly higher mean VE than either normals or athletes, p<0.05.

When assessing the maximum value of VE achieved at maximum exercise, athletes and normals reached similar values of VE, 138 09 \pm 29 49 l/min and 102 56 \pm 33 51 l/min, whereas patients achieved lower values of VE, 35 26 \pm 16 20 l/min, p<0.05

9.3.2 Normalized Minute Ventilation as a Function of Work

 \dot{V} E was normalized for lung volume by dividing VE by the individuals' Forced Vital Capacities (FVC). At quiet breathing, no difference existed between athletes and normals mean \dot{V} E/FVC, 1 73 ±0 59 and 2 41 ±0 60, however, patients presented with a significantly higher mean VE/FVC at rest, 5 22 ±2 34, p<0.05. All groups, however, reached similar maximum VE/FVC, 22 04 ±4 90 sec , 19 13 ±5 51 sec , 13 61 ±4 94 sec. for the athletic, normal and patient group respectively, p>0.05.

Mean VE/FVC versus absolute workload measured in watts (W) was plotted for each group [fig 9.2.b] All groups demonstrated a curvilinear relationship of VE/FVC to work Patients presented with significantly higher VE/FVC than athletes and normals at all levels of absolute workload. Athletes and normals presented with similar VE/FVC from 0 to 140 watts. However at >160 watts, nearing the maximum workload, achieved by normal subjects, the VE/FVC of the normal group became significantly higher than the athletes, p<0.05

9.3.3 Normalized Minute Ventilation as a Function of Workload (% Maximum)

The mean VE/FVC of individual subjects was expressed as a function of the absolute workload in watts, normalized as a percentage of the maximum workload in watts achieved by each subject [fig 9 2 c] All subjects demonstrated a curvilinear relationship of VE/FVC



Figure 9.2 a Minute ventilation (VE) plotted as a function of work, measured in watts. Points represent group mean values \pm 1 sd QB represents quiet breathing



Figure 9.2 b VE/FVC plotted as a function of work, measured in watts Points represent group mean values ± 1 sd. QB represents quiet breathing



Figure 9.2 c VE/FVC plotted as a function of work, normalized as % maximum Points represent group mean values + 1 sd QB represents quiet breathing



Figure 9.2 d Dyspnea plotted as a function of VE/EVG Linear regression lines are presented for each individual subject

to work (% maximum) No difference existed between the mean VE/FVC of the athletic or normal groups at any level of normalized workload. Comparing the athletic and normal groups to the patients, the patient group demonstrated higher VE/FVC from 0 to 20%, p<0.05, but thereafter, similar VE/FVC up to 100% of their maximum workloads, p>0.05

The rate of increase of VE/FVC with respect to normalized workload was examined by plotting VE/FVC on a common logarithmic scale versus % maximum workload. All groups demonstrated a positive, linear relationship, mean correlation coefficients of 0.99 $\pm 0.00, 0.97 \pm 0.02, 0.97 \pm 0.03$ for the athletic, normal and patient groups respectively. The rate of increase of VE/FVC with increasing workload expressed as a % of maximum was similar between the normal and athletic groups, 0.84 ± 0.16 and 0.97 ± 0.11 , however was significantly lower in the patient group, 0.36 ± 0.13 , p<0.05.

9.4 Dyspnea as a Function of Minute Ventilation

Dyspnea was plotted as a function of VE normalized by FVC for each individual subject **[fig 9.2 d]** All subjects demonstrated a positive, linear relationship between dyspnea and VE/FVC, mean correlation coefficients of 0.97 \pm 0.03, 0.97 \pm 0.02 and 0.91 \pm 0.09 for the athletic, normal and patients groups respectively. The mean response rates of dyspnea to changes in VE/FVC (slopes of the regression lines) were also similar; 0.34 \pm 0.08, 0.45 \pm 0.07, 0.87 \pm 0.52 for the athletic, normal and patient groups respectively, p>0.05. However, as observed, a large variation in response rates existed within the patient group. The range of response rates within the patient group was 0.41 to 1.80. This demonstrates that although some patients did present with similar response rates as the athletes and normals, other patients demonstrated much higher dyspnea response rates to increasing ventilation normalized for lung size **[fig 9.2 d]**

It is intriguing to see that as a group, patients, normals and athletes can reach very different levels of ventilation [fig 9.2.a], however, when that ventilation ($\dot{V}E$) is normalized for lung size (VE/EVC), significant overlap exists within individual subjects' dyspnea responses independent if the subject is an elite athlete, normal or COPD patient.

9.5 Minute Ventilation Normalized as a Percentage of Reserve Capacity

9.5.1 VE/MVV as a Function of Work

In order to determine what percentage of their maximum dynamic ventilatory capacity was being utilized throughout the exercise test, subject's VE was normalized as a percentage of their predicted maximum voluntary ventilation (MVV). At quiet breathing, the athletic and normal groups presented with similar mean VE/MVV, $4.94 \pm 1.74\%$ and $7.47 \pm 2.39\%$ respectively, while the patient group presented with a significantly higher mean VE/MVV at rest, $32.18 \pm 18.60\%$, p<0.05. At maximum exercise, all groups demonstrated similar mean maximum VE/MVV, $58.15 \pm 14.67\%$, $59.26 \pm 20.00\%$ and $8.1.24 \pm 32.72\%$, for the athletic, normal and patient groups respectively. Once again, a large variance existed within the patient group, range 45.38% to 135.75%. This indicated that while some patients were utilizing similar percentages of their maximum ventilatory capacity as the athletes and normals during maximum exercise, other patients minute ventilatory levels were far in excess of the defined maximum sustainable ventilatory level of 60% MVV

Group mean VE/MVV values were plotted versus absolute workload measured in watts (W) [fig 9 3 a]. All groups demonstrated a curvilinear relationship of VE/MVV to absolute work. Patients presented with significantly higher VE/MVV at all levels of absolute work, p<0.05. Athletes and normals presented with similar VE/MVV from 0 W to 120 W. However, at > 120 W, normals'mean VE/MVV was significantly higher at each level of work up to maximum workloads, p<0.05.

9.5.2 VE/MVV as a Function of Workload (% maximum)

Individual subject's mean VE/MVV were expressed as a function of workload normalized as a percentage of the maximum workload achieved by each subject [fig 9 3 b] Please refer to section 9 2.2 for the purpose for normalizing workload in this manner

All subjects demonstrated a curvilinear relationship of VE/MVV to normalized work (% maximum) No difference existed between the mean VE/MVV of the athletic or normal groups at any level. The patient group demonstrated significantly higher mean VE/MVV only up to 40% of maximum work

The rate of increase of VE/MVV with respect to normalized workload was examined



Figure 9.3.a VE/MVV (%) plotted as a function of work, measured in watts. Points represent mean values ± 1 sd. QB represents quiet breathing.



Figure 9.3 b VE/MVV (%) plotted as a function of workload, normalized as a % maximum. Points represent group mean values ± 1 sd QB represents quiet breathing.



Figure 9.3.c Common log of VF/MVV plotted as a function of workload, normalized as % maximum Lines represent the linear regressions for individual subjects.





by plotting the common logarithmic relationship of $\dot{V}E/MVV$ versus normalized workload (% maximum) for each individual subject [fig 9.3.c] All groups demonstrated positive linear relationships, mean correlation coefficients of 0.99 ± 0.00, 0.97 ± 0.02 and 0.96 ± 0.03 for the athletes, normals and patients respectively. The rate of increase of VE/MVV was similar between the normal and athletic groups, 0.01 ± 0.00 and 0.01 ± 0.00, however patients presented with significantly lower mean rates of increase of $\dot{V}E/MVV$ with increasing levels of normalized workload, 0.004 ± 0.00, p<0.05

9.6 Dyspnea as a Function of VE/MVV

Dyspnea was plotted as a function of VE/MVV for individual subjects [fig 9.3.d] All subjects demonstrated a positive linear relationship between dyspnea and VE/MVV; mean group correlation coefficients of 0.97 \pm 0.03, 0.97 \pm 0.02 and 0.92 \pm 0.08 for the athletes, normals and patients respectively. Response rates of dyspnea to increases in VE/MVV (slopes of the regression lines) were also similar between athletic, normal and patient groups, 0.12 \pm 0.03 0.15 \pm 0.02 and 0.13 \pm 0.09, p>0.05. Similar to the observed dyspnea responses to VE/FVC of subjects, overlap also existed between individual subject's response rates when dyspnea was related to ventilation, normalized by their ventilatory reserve capacity.

97 The Relationship of Dyspnea to Pleural Pressure Changes

The analysis of the relationship of dyspnea to pleural pressure changes consists of regression analysis of the relationships between changes in individual subject's dyspnea levels to changes in inspiratory pleural pressure swings (ΔPes). Values of ΔPes are also compared, both at quiet breathing and at maximum exercise for each of the three subject groups

971 APes Normalized by Maximal Inspiratory Pressure (MIP)

Previous investigators have normalized pleural pressure changes by expressing ΔPes as a percentage of the Maximal Inspiratory Pressure (MIP). At quiet breathing,

athletes and normal subjects presented with similar $\Delta \text{Pes/MIP}$, 0.04 ±0.01 and 0.04 ±0.01, p<0.05. Patients were generating significantly higher $\Delta \text{Pes/MIP}$ at quiet breathing, 0.18 ±0.08. At maximum exercise, the athletic and normal groups presented with similar mean $\Delta \text{Pes/MIP}$, 0.28 ±0.09 and 0.25 ±0.04. The patient group presented with a significantly higher mean $\Delta \text{Pes/MIP}$, 0.54 ±0.21, p<0.05. As noted, a large variance existed within the patient group. Patient 6 reached a maximum $\Delta \text{Pes/MIP}$ value of 0.27, similar to both athletes and normals. Whereas, at the other extreme, P2 reached a maximum $\Delta \text{Pes/MIP}$ of 0.92.

Dyspnea was plotted as a function of $\Delta \text{Pes/MIP}$ for each individual. Individual regression lines for this relationship are presented [iig 94 a]. All subjects demonstrated a positive, linear relationship between Dyspnea and $\Delta \text{Pes/MIP}$, mean group regression coefficients of 0.97 ±0.01, 0.97±0.02, 0.87±0.11 for the athletic, normal and patient groups respectively. This relationship represents the subject's dyspnea response to pleural pressure changes.

Much variation existed, within and between groups when dyspinea responses to pleural pressure changes were compared. Given this variation, the data was analyzed according to the following criteria. Within each group, a cut off response rate of 40 was set. Therefore any subjects with dyspinea responses above 40 were classified as high dyspinea responders, whereas any subjects within that group with dyspinea responses less than 40 were deemed low dyspinea responders.

9.7.1.1 Dyspnea Response to $\Delta Pes/MIP$ of Athletes

All four athletes presented with dyspnea responses which classified them as low dyspnea responders, mean group dyspnea response of 28.4 + 6.4

9.7.1.2 Dyspnea Response to $\Delta Pes/MIP$ of Normals

Two of the four normals (N1, N2) presented with high dyspnea responses, significantly greater than the athletes, mean response rate of 45-14+1-46, p<0.05. The two

remaining normals (N3,N4) presented with low response rates of 23.44 and 29.06, similar to that of the athletic group

9.7 1.3 Dyspnea Response to $\Delta Pes/MIP$ of Patients

All patients presented with dyspinea responses which were classified as low responses, group mean of 17.03 \pm 9.60, similar to the athletic group, normal subject low dyspinea responders, but significantly lower than the two normal subjects who were classified as high dyspinea responders, p<0.05

What was interesting was that within the patient group, a wide range of responses did exist. In fact, five of the seven patients (P1, P2, P4, P5, P7) presented with very low response rates, mean of 11.69 \pm 3.33 while the remaining two patients (P3,P6) presented with much higher response rates, mean 30.37 \pm 3.15. So, in terms of the definition of high and low response, as a group, the patients were defined as low responders, however within the patient group, it appears that two distinct response patterns also existed.

9.7.2 ΔPes Normalized by the Maximum Pressure Generating Capacity of the Inspiratory Muscles during Exercise

ΔPes was expressed as a function of Pcap – At quiet breathing, the athletic and normal groups presented with similar mean ΔPes/Pcap of 0.05 ±0.02 and 0.04 ±0.01 respectively, p>0.05 – The patient group demonstrated significantly higher ΔPes/Pcap at quiet breathing, 0.21 ±0.09, p<0.05 – At maximum exercise, the athletic and normal groups demonstrated similar mean ΔPes/Pcap, 0.45 ±0.17 and 0.34 ±0.03, p<0.05 – Patients presented with significantly higher mean ΔPes/Pcap when compared to the normal group, p<0.05, but similar mean ΔPes/Pcap when compared to the athletic group, 0.63 ±0.24, p>0.05 – However, a large variance existed within the patient group – P6 reached a maximum ΔPes/Pcap value of 0.34, similar to both athletes and normals. Whereas, at the other extreme, P2 reached a maximum ΔPes/Pcap of 1.01 far in excess of that achieved by any normal or athlete

The relationship between dyspinea and $\Delta Pes/P_{cap}$ was assessed individual regression lines of this relationship for each individual subject are plotted [fig 94b]. A

positive, linear relationship existed between dyspnea and $\Delta \text{Pes/Pcap}$ for all subject groups, mean correlation coefficients of 0.95 ±0.01, 0.95±0.02, 0.87±0.11 for the athletes, normals and patients respectively.

The dyspnea response rates of both athletes and normals were statistically lower when using Pcap normalization for Δ Pes versus MIP normalization, p<0.05. Although the dyspnea response rates were lower for both low and high responding patients with Pcap normalization, the differences in response rates when compared to MIP normalization were not significant, p>0.05.

Similar to observations made with MIP normalization, a lot of variation existed, within and between groups when dyspnea responses to pleural pressure changes, normalized by Pcap, were compared

In order compare varying dyspnea response rates, not only between groups but within groups as well, a new lower cut off response rate of 25 was set, to compensate for the statistically lower response rates observed in athletes and normals with Pcap normalization. Therefore any subjects with dyspnea responses above 25 were classified as high dyspnea responders, whereas any subjects with dyspnea responses less than 25 were deemed low dyspnea responders.

9.7 2 1 Dyspnea Response to Δ Pes/Pcap of Athletes

All athletes (A1, A2, A3, A4) presented with similar dyspnea responses below 25, mean 18 30 +3 50, and therefore were classified as "low dyspnea responders".

9.7 2.2 Dyspnea Response to Δ Pes/Pcap of Normals

Two of the four normals (N1, N2) presented with dyspnea responses above 25, mean value of 27.45 \pm 1.48, which is significantly higher than the mean response of the athletic group, p<0.05. Therefore, N1 and N2 will be referred to as "high dyspnea responders" in all subsequent analysis. Two of the four normals (N3, N4) presented with similar dyspnea responses below 25 as the athletes, mean of 18.70 \pm 0.26, p>0.05 and therefore are referred to as "low dyspnea responders" in all subsequent analysis.



Figure 9.4.a Dyspnea is plotted as a function of ΔPes/MIP Lines plotted represent the linear regression lines of each individual subject for this relation ship





9723 Dyspnea Response to ∆Pes/Pcap of Patients

As a group, the COPD patients all presented with dyspnea response rates below 25 It is observed, though, that much variation in dyspnea response existed within this patient group. In fact, five of the seven patients (P1,P2,P4,P5,P7) presented with extremely low dyspnca responses, statistically lower than the athletic group or low responding normals, mean 8.85 + 1.51, p<0.05. It was therefore proposed that although all patients presented with a "low dyspnea response" given the above mentioned criteria, that patients as a distinct group may display both "low" and "high" dyspnea responses. Indeed, the remaining two patients (P3,P6) presented with statistically higher dyspnea response rates, mean 22.53 ± 0.15 , when compared to the lower responding patients

Future results focus upon the respiratory inuscle coordination and whether an individual subject's adopted breathing pattern appears to have any relationship to his observed dyspnea response rate to changes in pleural pressure normalized by Pcap. In order to facilitate this analysis, and based upon the above results the following definition will be used throughout the remainder of this thesis. Within the athletic and normal groups, all athletes are termed low dyspnea responders (LDR), and N3 and N4 are also referred to as low dyspnea responders (LDR). N1 and N2 will be referred to as high dyspnea responders. The patient group will be treated as distinct, therefore within that group P1,P2,P4,P5,P7 will be referred to as low dyspnea responders whereas P3,P6 will be referred to as high dyspnea responders.

9.7 3 Validity of Regression Analysis in Predicted Maximum Achieved △Pes/Pcap Values

The actual measured maximum $\Delta \text{Pes/Pcap}$ achieved by individual subjects was compared to that predicted by linear regression analysis. All subjects' actual maximum $\Delta \text{Pes/Pcap}$ values were within 0.10 of that predicted by linear regression (the dyspnea response) with the following exceptions [fig 9.5]

- 1 N4 The regression line overestimates his maximum achieved $\Delta Pes/Pcap$ by 0.17
- 2 P1 The regression line overestimates his maximum achieved ΔPes/Pcap by 0.26
 Both of these subjects demonstrated a change in their response patterns midway



Figure 9.5 Individual data points and linear regression lines are presented for normal subject 4 (N4) and patient 1 (P1) See text for description.

through the exercise test. At approximately 75% of his maximum exercise level achieved, N4 no longer achieved significant increases in $\Delta \text{Pes/Pcap}$, despite further increasing dyspnea. P1 demonstrated a similar pattern, whereby at 60% of his maximum exercise level achieved, dyspnea increased at an accelerated rate despite smaller changes in $\Delta \text{Pes/Pcap}$.

9.8 Analysis of "Pre-Inspiratory Flow" Pressure Generation

A combined four parameter graph is presented in **Figure 9.6** For each of the four relationships presented, the regression lines for each individual subject are presented. In the upper right quadrant, the relationship of Dyspnea to end expiratory pleural pressure (EEP) measured in cmH20 is plotted. The lower right quadrant plots individuals' relationships between EEP and the rate of change of pleural pressure (prior to the onset of inspiratory flow) with respect to time $(\Delta Pes/\Delta t)_{PRt}$, measured in cmH20/second. The upper left quadrant plots Dyspnea as a function of the rate of change of gastric pressure (prior to the onset of the onset of inspiratory flow) with respect to time $(\Delta Pga/\Delta t)_{PRt}$, measured in cmH20/second. The upper left quadrant plots the relationships between $(\Delta Pga/\Delta t)_{PRt}$, measured in cmH20/second. The lower left quadrant plots the relationships between $(\Delta Pes/\Delta t)_{PRt}$, measured in cmH20/second. The lower left quadrant plots the relationships between $(\Delta Pes/\Delta t)_{PRt}$, measured in cmH20/second. The lower left quadrant plots the relationships between $(\Delta Pes/\Delta t)_{PRt}$ versus $(\Delta Pga/\Delta t)_{PRE}$. Numerical values (mean \pm sd) for each subject for each of the four variables at quiet breathing and at maximum exercise are presented in Table 9.3

9.8.1 Dyspnea versus EEP [Figure 9.6, upper right]

Changes in EEP can result from changes in either end expiratory lung volume (EELV), increased expiratory muscle recruitment or a combination of both. Observed changes in EEP occur during exercise, both in normal, athletic and COPD subjects. These changes in EEP will directly effect the magnitude of the load placed upon the inspiratory muscles for the next inspiratory effort. The purpose of the following analysis was not to imply a causal effect between EEP and dyspnea, but rather as a method to infer how much load was placed upon the inspiratory muscles and whether changes in this loading pattern affected dyspnea.

At quiet breathing, no difference existed between the measured EEP of athletes and normals, -2 80 +2 37 cmH20 and -5 82 ±3 98 cmH20. Patients as a group presented with



Figure 9.6 Lines plotted represent the linear regression for each individual subject for each of the four presented relationships Solid lines (-----) represent athletes, dashed lines (-----) represent normal subjects and dotted lines (-----) represent COPD patients See text for a detailed description of each plot 84

SUBJECT	QB Dyspnea	<mark>м</mark> ах Dysphea	QB EEP (cmH20) mean ±sd	MAX EEP (cmH2D) mean ±sd	QB APes At _{ens} (cm-420 sec) mean tsu	MAX APes Albus omm20 sec) Mean tisd	QB A ^p ga Ate _{ne} Jom420 sect mean tso	MAX APya At _{ona} om420 sect mean 183
A1	0	95	-0.41 ±0.42	22 84 ±4 24	00 ±0 00 C	-228 89 ± 27 59	0.00 ± 0.00	-188 74 -42 54
A2	J	40	-1 39 ±0 60	6 31 ±3 29	0.00 ±0.00	-94 25 + 39 15	0.00 ± 0.00	-94 50 ±30 70
A3	0	60	-5 68 ±0 44	9.01 ±3.47	00 0± 00	-127 90 ± 75 13	0.00 ± 0.00	-96 74 ±34 42
A4	0	70	-3 72 ±0 76	12 94 ±4 01	06 S± 00 C	-354 57 ±130 21	0.0 ± 0.00	-45 84 ± 7 35
N1	0	90	-3 42 ±1 28	5 91 ±2 52	000±000	-91 96 ± 43 87	000±000	-64 53 ±57 77
N2	0	80	-1 99 ±0 50	10 82 ±1 22	000 ±0 00	-140 02 ± 48 87	000±000	-68 51 ±24 74
N3	0	50	-10 90 ±0 48	6 03 ±4 48	0 00 ±0 00	-101 26 ± 45 26	0 00 ± 0 00	-104 26 ±46 98
N4	0	80	-6 95 ±0 26	3 14 ±4 50	000 ± 000	-37 01 ± 25 89	000 ± 000	27 98 ±29 50
P1	0	70	-0 74 ±0 29	11 45 ±3 25	0 00 ± 0 00	-268 30 ± 39 41	0 00 ± 0 00	-53 68 ±43 98
P2	0	70	-1 16 ±0 72	14 96 ±4 36	0 00 ±0 00	-192 24 ± 27 52	0 00 ± 0 00	-130 98 ±28 18
P3	0	50	-3 18 ±1 41	-0 20 ±2 66	-93 46 ±12 43	-90 04 ± 2 21	-12 22 ±39 57	-4 85 ±44 34
P4	10	50	-1 09 ±0 96	16 79 ±1 00	-20 12 ±11 63	-149 79 ± 23 25	-1 13 ±11 25	-46 72 ± 7 56
P5	10	50	1 62 ±1 13	5 54 ±0 86	-30 12 ±6 49	-77 60 ± 8 20	-25 27 ± 6 51	-51 24 ± 6 19
P6	20	90	-0 95 ±1 02	6 92 ±2 92	0 00 ±0 00	-58 93 ± 22 64	000 ± 000	-32 06 ±19 78
P7	20	40	-0 20 ±0 20	9 02 ±1 12	000 ±0 00	-102 90 ± 12 84	0 00 ± 0 00	-36 44 ± 4 71

Table 9.3

similar results to the athletes, -0.81 \pm 1.42 cmH20, but higher EEP than the normals, p<0.05

At maximum exercise, the athletes, normals and patients presented with a wide range of EEP (9.02 cmH20 to 22.84 cmH' Ω , 3.14 cmH20 to 10.82 cmH20, -0.20 cmH20 to 16.79 cmH20). Although group means were not statistically significant, it was observed that within the patient group, 4 of those patients previously defined as low responders (P1,P2,P4.P7) presented with higher EEP (mean EEP of 13.05 ± 3.49cmH20) when compared to the two patients termed high responders (P3,P6)

9.8.1.1 Dyspnea Response to EEP of Athletes

The athletic group presented with a strong linear relationship between dyspinea and EEP, mean group correlation coefficient \pm 1 sd of 0.93 \pm 0.02, and a mean dyspinea response to EEP of 0.48 \pm 0.03

9.8.1.2 Dyspnea Response to EEP in Normals

All normal subjects presented with a positive relationship between dyspnea and EEP, however, the mean group correlation coefficient of the group was significantly lower than the athletic group, 0.78 \pm 0.06, p<0.05

Three normal subjects, N1,N2,N4, presented with higher dyspnearesponses to EEP when compared to the athletes, range of 0.63 to 0.91 while N3 presented with a low dyspnea response to EEP, similar to the athletes, 0.29

9.8.1.3 Dyspnea Response to EEP in Patients

As a group, a large variation in dyspnearesponse to EEP existed. Five of the seven patients (P1,P2,P4,P6,P7) demonstrated a strong linear relationship between dyspnea and EEP, mean correlation coefficient of 0.88 ±0.11. P3, a high responding patient, demonstrated a weak relationship between dyspnea and EEP, correlation coefficient of 0.58. P5 presented with no significant correlation. Observing responses for those patients who did present with a strong relationship, those patients termed low responders (P1,P2,P4,P7) had a similar low dyspnea response as observed in the athletes, mean

response of 0.36 ± 0.17 . The remaining patient, P6, presented with a much higher mean response rate of 0.84 ± 0.25 when compared to the athletes, N3 and P1,P2,P4,P7. This value was similar to the dyspnea response measured in N1,N2 and N4.

982 EEP versus $(\Delta \text{Pes}/\Delta t)_{\text{pri}}$ [figure 9.6, lower right]

At quiet breathing all athletes and normals presented with 0 $\Delta \text{Pes}/\Delta t_{\text{PRE}}$ (ie. no negative pleural pressure swing preceded the onset of inspiratory flow). Four of the seven patients (P1,P2,P6,P7) also presented with 0 $\Delta \text{Pes}/\Delta t_{\text{PRE}}$ at quiet breathing. P3 and P5 presented with negative mean $\Delta \text{Pes}/\Delta t_{\text{PRE}}$ (-93 46 ± 12 43 cmH20/sec and -30 12 ± 6 49 cmH20/sec), whereas P4 presented with very small and inconsistent negative pleural pressure swings prior to the onset of inspiratory flow.

At maximum exercise, two of the four athletes (A1 & A4) presented with very rapid mean $\Delta Pes/\Delta t_{PRE}$ (-281.73 ± 74.73 cmH20/sec) A3 presented with a lower mean $\Delta Pes/\Delta t_{PRE}$ of -127.90 cmH20/sec however he ceased exercise at 80% of his maximum workload

In the normal subjects, three of the four (N1,N2,N3) presented with similar, slower $\Delta \text{Pes}/\Delta t_{\text{PRE}}$, (-111.08 + 25.49 cmH20/sec). N4 presented with a very slow $\Delta \text{Pes}/\Delta t_{\text{PRE}}$ of - 25.89 cmH20/sec, with a very large variance about the mean.

In the patients, a wide range of $\Delta \text{Pes}/\Delta t_{\text{PRE}}$ were observed at maximum exercise. Three of the patients, all low responders, (P1, P2, P4) presented with very rapid $\Delta \text{Pes}/\Delta t_{\text{PRE}}$ similar to the athletes (mean rate of -203.44 ±60.04 cmH20/sec) = P5 presented with a slower $\Delta \text{Pes}/\Delta t_{\text{PRE}}$ of -77.63 ±8.20 = P3 and P7 presented with similar $\Delta \text{Pes}/\Delta t_{\text{PRE}}$ of -90.04 ±-2.21 and -102.9 ±12.84. Finally P6 presented with the lowest $\Delta \text{Pes}/\Delta t_{\text{PRE}}$, -58.93 ±22.64.

9.8.2.1 The Relationship of EEP to $\Delta Pes/\Delta t_{PRF}$

All athletes and normals demonstrated a strong linear relationship between EEP and $\Delta \text{Pes}/\Delta t_{\text{PRE}}$, (mean correlation coefficients of 0.92 ±0.05 and 0.87 ±0.08 respectively). Four patients (P1,P2,P4,P7), all defined as low responding patients, also demonstrated a very strong correlation between EEP and $\Delta \text{Pes}/\Delta t_{\text{PRE}}$, mean correlation coefficient of 0.98 ±0.01. P5 and P6 demonstrated a weaker, yet still positive relationship between EEP and

 $\Delta Pes/\Delta t_{PRE}$, mean correlation coefficient of 0.67 ±0.07, while P3 (high responder) presented with a poorer correlation coefficient of 0.35

9.8.3 Dyspnea versus $\Delta Pga/\Delta t_{PRE}$ [fig 9.6, upper left]

At quiet breathing, neither athletes or normals demonstrated negative qastric pressure swing prior to the onset of inspiratory flow. Similarly, four patients (P1,P2,P6,P7) also demonstrated no negative gastric pressure swing preceding flow. The remaining patients (P3,P4,P5) presented with inconsistent patterns of negative gastric pressures swings prior to the onset of flow with values of $\Delta Pga/\Delta t_{PRE}$ of -12.22 +39.57 cmH20/sec, -1.16 ±11.25 cmH20/sec, -25.27 ±6.51 cmH20/sec, respectively

At maximum exercise, all subjects except for one normal (N4), presented with negative gastric pressure swings prior to the onset of inspiratory flow. The athletes presented with a high variation in $\Delta Pga/\Delta t_{PRE+}$ range -45.84 +7.35 (A4) to -188.74 +42.54 (A1). The normals also presented with a large range of values, +27.98 +29.50 (N4) to -104.26 ±46.98 (N3). Patients also presented with large ranges of, and sometimes inconsistent patterns reflected by a large standard deviation, of $\Delta Pga/\Delta t_{PRE}$ (-4.85 ±44.34 (P3) to -130.98 ±28.18 (P2)).

9.8.3.1 Dyspnea Response to $\Delta Pga/\Delta t_{PRF}$ of Athletes

A direct linear relationship existed between dyspnea and $\Delta Pga/\Delta t_{PRF}$ for all athletes, mean group correlation coefficient of 0.90 +0.04. All athletes also presented with similar low dyspnea responses to $\Delta Pga/\Delta t_{PRF}$, -0.06 ±0.02.

9.8.3 2 Dyspnea Response to $\Delta Pga/\Delta t_{pre}$ of Normals

Three normal subjects (N1,N2,N3) presented with strong positive relationships between dyspnea and $\Delta Pga/\Delta t_{PRE}$ mean correlation coefficient of 0.80 ± 0.03. N4 presented with an insignificant correlation between dyspnea and $\Delta Pga/\Delta t_{PRE}$. Looking within the normal group, the two normal subjects defined previously as high responders (N1,N2) presented with higher dyspnea responses to $\Delta Pga/\Delta t_{pre}$ than the athletic group, -0.14 and -

0.10, p<0.05 N3 presented with a low dyspnea response similar to the athletes, -0.05

9.8.3.3 Dyspnea Response to $\Delta Pga/\Delta t_{pRI}$ of Patients

All patients demonstrated a positive linear relationship between dyspnea and $\Delta Pga/\Delta t_{PRL}$ mean group correlation coefficient of 0.84 ±0.11. Three patients, P2,P4,P7, (all low responders) presented with similar low mean dyspnea response to $\Delta Pga/\Delta t_{PRE}$ when compared to the athletic group, -0.06 ± 0.02. Three patients (P1,P5,P6) presented with similar dyspnea responses when compared to high responding normal subjects (N1,N2), -0.14 ±0.02. Although P3 demonstrated a positive dyspnea response to $\Delta Pga/\Delta t_{PRE}$ of 0.33, the correlation coefficient of the relationship was only 0.69.

9.8.4 The Relationship of $\Delta Pes/\Delta t_{PRI}$ to $\Delta Pga/\Delta t_{PRI}$ [fig 9.6 lower left]

The relationship of $\Delta \text{Pes}/\Delta t_{\text{PRI}}$ to $\Delta \text{Pga}/\Delta t_{\text{PRE}}$ indicates the contribution of the rate of the gastric pressure swing to the rate of the pleural pressure swing and is defined as the Pdi ratio. A ratio of 1 indicates that for every change in the rate of the gastric pressure swing, an equal and similar directional change in the rate of the pleural pressure swing occurred (Pdi = 0), whereas a ratio of -1 indicates that a large Pdi is being generated

9.8.4.1 $\Delta Pes/\Delta t_{PRI}$ to $\Delta Pga/\Delta t_{PRI}$ of Athletes

All athletes demonstrated a positive linear relationship of $\Delta Pes/\Delta t_{PRE}$ to $\Delta Pga/\Delta t_{PRE}$, mean group correlation coefficient of 0.90 ± 0.08. Three athletes (A1,A2,A3) presented with very low Pdi ratios, mean 1.03 ± 0.25 whereas A4 presented with a higher Pdi ratio of 4.15.

9842 $\Delta Pes/\Delta t_{PRF}$ to $\Delta Pga/\Delta t_{PRF}$ of Normals

Three normals (N1,N2,N3) demonstrated a positive linear relationship of $\Delta \text{Pes}/\Delta t_{\text{PRE}}$ to $\Delta \text{Pga}/\Delta t_{\text{PRI}}$, mean correlation coefficient of 0.94 ±0.05. No correlation existed between $\Delta \text{Pes}/\Delta t_{\text{PRI}}$ and $\Delta \text{Pga}/\Delta t_{\text{PRI}}$ for N4. N1 and N2 presented with a mean Pdi ratio of 1.67 ± 0.43. N3 presented with a low Pdi ratio, similar to the athletes (A1,A2,A3) of 1.03.

9.8.4.3 $\Delta Pes/\Delta t_{PRE}$ to $\Delta Pga/\Delta t_{PRE}$ of Patients

All patients demonstrated a positive linear relationship between $\Delta \text{Pes}/\Delta t_{\text{PRE}}$ and $\Delta \text{Pga}/\Delta t_{\text{PRE}}$ mean correlation coefficient of 0.89 ± 0.10. Patients presented with varying Pdi ratios, range 0.97 (P6) to 6.84 (P3)

9.9 Analysis of the Initial Phase of Pressure Generation (from 0 to Peak Inspiratory Flow Acceleration)

A combined four parameter graph is presented in **Figure 9.7** For clach of the four relationships presented, the regression lines for the relationships of each individual subject are displayed. In the upperright quadrant, Dyspnea is plotted versus $\Delta Pes/P_{cap}$. The lower right quadrant plots $\Delta Pes/P_{cap}$ versus rate of change of pleural pressure, measured from the onset of inspiratory flow to the peak of the inspiratory flow acceleration ($\Delta Pes/\Delta t$), measured in cmH20/second. The upper left quadrant plots Dyspnea versus the rate of change of gastric pressure with respect to time, again from the onset of inspiratory flow acceleration ($\Delta Pes/\Delta t$), measured in spiratory flow acceleration ($\Delta Pga/\Delta t$), measured in cmH20/second. The tower left quadrant plots $\Delta Pes/\Delta t$ versus $\Delta Pga/\Delta t$. For each individual subject, numerical values for each of the four variables at quiet breathing and at maximum exercise are presented in **Table 9.4**.

9.9.1 Dyspnea versus $\Delta Pes/P_{cap}$ [fig 97, upper right]

This relationship has been previously discussed (section 9.7.2). It is re-plotted in order to compare its results to other investigated parameters.

9.9.2 \Delta Pes/\Delta t versus \Delta Pes/Pcap [fig 97, lower right]

The relationship of $\Delta Pes/\Delta t$ to $\Delta Pes/Pcap$ expressed in percentage indicates, for a given 1% increase in the amount of pleural pressure being generated, expressed in relationship to the maximum pressure generating capacity available, what is the change in the velocity of the pleural pressure swing from 0 inspiratory flow to peak inspiratory flow acceleration.

All subjects demonstrated strong relationships between $\Delta Pes/\Delta t$ and $\Delta Pes/Peap.$



Lines plotted represent the linear regression for each individual subject for each of the four presented relationships Solid lines (----) represent athletes, dashed lines (----) represent normal subjects and dotted lines () represent COPD patients. See text for a detailed description of each plot. 91

SUBJECT	QB Dyspnea	MAX Dyspnea	QB ∆Pes/Pcap mean ±sd	MAX ∆Pes/Pcap mean ±sd	QB ∆Pes/∆t (cmH20/sec) mean ±sd	MAX ∆Pes/∆t (cmH20/sec) mean ±sd	QB ∆Pga/∆t (cmH20/sec) mean ₊sd	MAX ∆Pga/∆t (cmH20/sec) mean ±sd
A1	0	95	0 03 ±0 01	0 64 ±0 05	-9 86 ± 3 95	-423 95 ±66 40	-3 03 ± 0 79	-212 00 ±41 34
A2	0	40	0 07 ±0 02	0 27 ±0 02	-3 67 ± 1 16	-123 38 ±14 33	-3 36 ± 0 90	-119 19 ±19 69
A3	0	60	0 03 ±0 00	0 33 ±0 04	-5 58 ± 1 86	-213 20 ±40 31	-2 76 ± 0 96	-23 44 ±57 16
A4	0	70	0 05 ±0 01	0 39 ±0 07	-8 98 ± 3 90	-293 85 ±54 46	6 51 ± 5 39	-151 44 ±31 34
N1	0	90	0 04 ±0 01	0 32 ±0 06	-11 28 ±10 14	-71 86 ±23 75	-2 41 ±16 42	-56 15 ±19 35
N2	0	80	0 04 ±0 01	0 32 ±0 32	-6 93 ± 2 92	-166 02 ±37 73	0 32 ± 2 30	-29 70 ±24 13
N3	0	50	0 03 ±0 00	0 34 ±0 03	-6 18 ± 1 13	-83 84 ±45 36	4 03 ± 2 14	-69 11 ±38 17
N4	0	80	0 05 ±0 00	0 51 ±0 09	-4 70 ± 0 43	-86 75 ±27 42	590±060	56 23 ±39 36
P1	0	- 0	0 21 ±0 04	0 74 ±0 08	-18 54 ± 5 53	-150 95 ±25 97	6 55 ± 2 09	-95 04 ±30 30
P2	0	70	0 20 ±0 02	1 01 ±0 11	-23 92 ± 3 15	-167 21 ±30 23	-5 30 ± 2 50	-109 45 ±11 58
P3	0	50	0 38 ±0 05	0 59 ±0 12	-109 62 ±30 59	-165 70 ±29 57	-122 ±14 17	-48 62 ±24 87
₽4	10	50	0 20 ±0 02	0 80 ±0 06	-28 78 ±12 71	-200 00 ±43 82	-9 20 ± 6 09	-59 73 ±36 75
₽5	10	50	0 22 ±0 04	0 59 ±0 05	-34 62 ± 6 09	-81 64 ±11 39	-20 59 ± 4 48	-59 28 ± 3 69
P6	20	90	0 09 ±0 02	0 34 ±0 05	-12 64 ± 5 89	-72 87 ±17 55	0 02 ± 2 01	-39 91 ±14 60
P*	20	40	0 17 ±0 02	0 3 ⁻ ±0 32	-30 23 ± 5 72	-92 12 ±20 93	-1 92 ± 2 51	-50 19 ± 8 76

Table 9.4
mean group correlation coefficients of 0.97 \pm 0.03, 0.91 \pm 0.03, 0.96 \pm 0.01 for the athletic, normal and patient groups respectively

9921 The $\triangle Pes/\Delta t$ to $\triangle Pes/Pcap$ Relationship in Athletes

All athletes presented with similar, high ($\Delta Pes/\Delta t$) / ($\Delta Pes/Pcap X 100$) relationships with a mean value of -6 51 +1 86 cmH20/sec/1%

9.9.2.2 The $\Delta Pes/\Delta t$ to $\Delta Pes/Pcap$ Relationship in Normals

The normal group presented with a mean relationship that was significantly lower than the athletic group, mean of -2.98 \pm 0.76 cmH20/sec/1%, p<0.05

9.9 2 3 The $\triangle Pes/\Delta t$ to $\triangle Pes/P_{cap}$ Relationship in Patients

The patient group presented with similar mean values as the normal group; however significantly lower than the athletic group -2 28 ± 0 60 cmH20/sec/1%, p<0 05

9.9.3 Dyspnea versus $\Delta Pga/\Delta t$ [fig 9 7, upper left]

Three of the four athletes (A1,A2,A3) presented with a mean Δ Pga/ Δ t of -3.05 ±0 30 cmH20/sec at quiet breathing. The remaining athlete (A4) presented with a positive mean Δ Pga/ Δ t at rest of 6 51 cmH20/sec

One normal subject (N1) demonstrated a negative mean $\Delta Pga/\Delta t$ of -2.41 cmH20/sec at quiet breathing. The remaining three normals (N2,N3,N4) demonstrated positive mean $\Delta Pga/\Delta t$ at rest, ranging from 0.32 cmH20/sec to 5.90 cmH20/sec

Five of the seven patients (P2,P3,P4,P5,P7) demonstrated negative mean Δ Pga/ Δ t at quiet breathing, ranging from -1.92 cmH20/sec (P7) to -20.59 cmH20/sec (P5). The remaining two patients (P1,P6) presented with positive mean Δ Pga/ Δ t at rest of 6.55 cmH20/sec and 0.02 cmH20/sec respectively. Addressing the variability of patient mean values, P3, P4 and P6 each presented with large variances about their individual means **[Table 9.4]**

At maximum exercise, two of the four athletes (A1,A4) presented with very rapid negative $\Delta Pga/\Delta t$ of -212 0 cmH20/sec (A1) & -151.44 cmH20/sec (A4) A2, who only

reached 80% of his maximum exercise also presented with a fast negative $\Delta Pga/\Delta t$ of -119 19 cmH20/sec. Even though A3 demonstrated a small negative mean value of -23 44, the standard deviation was ±57 16 cmH20/sec. This allows the inference that for some breaths he demonstrated negative gastric swings, sometimes there was no change in gastric pressure and sometimes a slightly positive gastric swing existed. Three of the four normals (N1,N2,N3) demonstrated a negative mean $\Delta Pga/\Delta t$ at maximum exercise of -51 65 ±20 08 cmH20/sec, significantly slower than athletes A1,A2,A3. The remaining normal (N4) presented with a positive Pga/ Δt at maximum exercise of +56 23 cmH20/sec All normal subjects presented with large variances about their individual mean values [Table 9.4]

All patients demonstrated similar negative mean $\Delta Pga/\Delta t$ at maximum exercise, -66 03 ±25 98 cmH20/sec At the individual patient level, four of the patients (P1,P3,P4,P6) demonstrated large variances about their individual mean $\Delta Pga/\Delta t$ [Table 9 4]

9.9.3.1 Dyspnea Response to $\Delta Pga/\Delta t$ of Athletes

Three of the four athletes (A1,A2,A4) demonstrated a strong correlation between dyspnea and $\Delta Pga/\Delta t$, mean correlation coefficient of 0.93 +0.06. A3 presented with a poor correlation coefficient of 0.40. As a group the athletes presented with a mean dyspnea response to $\Delta Pga/\Delta t$ of -0.04 +0.01.

9.9.3.2 Dyspnea Response to $\Delta Pga/\Delta t$ of Normals

Three of the four normals, N1,N3,N4 demonstrated a strong correlation between dyspnea and Δ Pga/ Δ t, mean correlation coefficient of 0.87 ±0.06. The correlation for N2 was insignificant. Within the normal group, each individual subject presented with a very different dyspnea response to Δ Pga/ Δ t. N1 presented with a high dyspnea response, when compared to the athletes, of ±0.14. N3 presented with a low dyspnea response, similar to the athletes, ±0.06, and N4 presented with a positive dyspnea response of 0.14.

9.9.3.3 Dyspnea Response to $\Delta Pga/\Delta t$ of Patients

All patients demonstrated a strong correlation between dyspnea and $\Delta Pga/\Delta t$, mean

correlation coefficient of 0.89 ± 0.09 Low responding patients (P1,P2,P4,P5,P7) demonstrated a similar mean dyspnea response to Δ Pga/ Δ t when compared to the athletes, -0.07 \pm 0.02, p>0.05 High dyspnea responding patients (P3,P6) demonstrated significantly higher mean dyspnea responses when compared to both the athletic group and the low responding patient group, -0.13 \pm 0.02, p<0.05

994 The Relationship of $\Delta Pes/\Delta t$ to $\Delta Pga/\Delta t$ [fig 9.7, lower left]

The relationship of $\Delta \text{Pes}/\Delta t$ to $\Delta \text{Pga}/\Delta t$ indicates the contribution of the rate of the gastric pressure swing to the rate of the pleural pressure swing (Pdi ratio). A ratio of 1 indicates that for every change in the rate of the gastric pressure swing, an equal and similar directional change in the rate of the pleural pressure swing occurred (Pdi = 0), whereas a ratio of -1 indicates that a large Pdi is being generated

9.9.4 1 $\Delta Pes/\Delta t$ to $\Delta Pga/\Delta t$ Relationship of Athletes

Three athletes (A1,A2,A4) demonstrated a positive relationship between $\Delta \text{Pes}/\Delta t$ and $\Delta \text{Pga}/\Delta t$, mean correlation coefficient of 0.97 ±0.02. A weaker relationship was demonstrated by A3, correlation coefficient of 0.64. As a group, the athletes presented with a mean value of $\Delta \text{Pes}/\Delta t / \Delta \text{Pga}/\Delta t$ of 1.81 ± 0.74.

9.9.4.2 $\Delta Pes/\Delta t$ to $\Delta Pga/\Delta t$ Relationship of Normals

Two normals (N1,N3) demonstrated a much strong relationship between $\Delta Pes/\Delta t$ and $\Delta Pga/\Delta t$ than either N2 or N4 – As a group, the normal subjects had a mean correlation coefficient of 0.73 ±0.22 – N1, N2 and N3 demonstrated Pdi ratio values of 1.22, 2.30 and 0.99, while N4 presented with a negative ratio of -1.14

9943 $\Delta Pes/\Delta t$ to $\Delta Pga/\Delta t$ Relationship of Patients

All patients demonstrated a strong linear relationship between $\Delta \text{Pes}/\Delta t$ and $\Delta \text{Pga}/\Delta t$, mean correlation coefficient of 0.94 ±0.08. The patient presenting the weakest correlation was P3, correlation coefficient of 0.77. All patients demonstrated similar ratios when compared to the both the athletic group and the three normals presenting with positive Pdi ratios (N1,N2,N3), mean patient Pdi ratio of 1.61 ±0.65, p>0.05

9.10 The Pressure-Time Index of the Inspiratory Muscles

The relationship between the duty cycle measured from pressure (Ti/Ttot) [fig 8 2b] and $\Delta Pes/P_{cap}$ is plotted (fig 9 8 a). Symbol and line definitions are presented in Table 9.5. Mean values at quiet breathing and at maximum exercise are plotted for each individual subject. Since the product of $\Delta Pes/P_{cap}$ and pressure Ti/Ttot represents the pressure-time index of the inspiratory muscles (TTim), iso-TTim lines were calculated and plotted, TTims of 0 10, 0 15, 0 20, 0 25.

9.10.1 Pressure Duty Cycle (Ti/Ttot)

The athletic, normal and patient groups presented with similar mean pressure derived Ti/Ttot values at quiet breathing, 0.38 ± 0.06 , 0.42 ± 0.07 and 0.38 ± 0.07 , p>0.05

At maximum exercise, athletes and normals presented with similar mean Ti/Ttot, 0.55 ± 0.05 and 0.57 ± 0.04 . The patient group had significantly lower mean Ti/Ttot when compared to both the athletic and normal group, 0.40 ± 0.07 , p<0.05

9.10.2 ΔPes/Pcap

Athletes and normals presented with similar mean $\Delta Pes/Pcap$ at quiet breathing, 0.03 ±0.01 and 0.02 ±0.00; however patients' mean $\Delta Pes/Pcap$ at quiet breathing was significantly higher, 0.14 ±0.06, p<0.05

At maximum exercise, the mean $\Delta Pes/P_{cap}$ was not statistically different between the athletic, normal and patient groups, 0.32 ±0.12, 0.26 ±0.05, 0.45 ±0.18. Looking at individual subjects within the groups, A1 presented with a very high $\Delta Pes/P_{cap}$ of 0.46, resulting in the large standard deviation observed about the mean of the athletic group. Individual patients also presented with widely varying values of $\Delta Pes/P_{cap}$ at maximum exercise. P1,P2,P4 (all low responding patients) presented with higher $\Delta Pes/P_{cap}$ of 0.62 when compared to the remaining patients, athletes and normals, mean $\Delta Pes/P_{cap}$ of 0.62 ±0.09. P3,P5,P6,P7 presented with similar mean $\Delta Pes/P_{cap}$ when compared to the



PATIENTS		
•	\bullet	Ρ1
▼	▼	Ρ2
	G	Ρ3
		Ρ4
•	•	Ρ5
0	0	Ρ6
∇	∇	Ρ7

Table 9.5



Figure 9.8.a Ti/Ttot is plotted versus APes/Pcap Points represent values at quiet breathing and maximum exercise for each subject See text for details



for each subject. See text for details

9 10 3 TTim

At quiet breathing, the athletic and normal groups presented with similar mean TTim, 0.01 \pm 0.00 and 0.01 \pm 0.00. The patient group presented with significantly higher mean TTim at quiet breathing compared to both athletes and normals, 0.05 \pm 0.02, p<0.05. At maximum exercise, no significant difference existed between the mean TTim values achieved by the athletic group, 0.16 \pm 0.08, normal group, 0.15 \pm 0.02, or patient group, 0.18 \pm 0.07, p>0.05

Within groups, A1 achieved a much higher mean TTim of 0.27 when compared to the remaining athletes. This resulted in the observed large standard deviation about the athletic group mean. A3 and A4 achieved similar maximum TTim when compared to the group of normal subjects, 0.13 and 0.15. Patients again demonstrated a large variance in maximum achieved TTim. P1,P2,P4 (all low responders) achieved higher TTims of 0.26, 0.23, 0.27, when compared to the remaining patients. Their TTim resembled that of A1 P3,P5,P6,P7 reached similar levels of TTim to the remaining athletes A2,A3,A4 and all normal subjects, mean TTim of 0.12 \pm 0.01

9.11 The Pressure-Time Index of the Diaphragm

The relationship between the pressure duty cycle (Ti/Ttot) and $\Delta PdT/Pdi_{max}$ is plotted (fig 9 8 b) Please see Table 9 5 for definitions of symbols and lines. Mean values at quiet breathing and at maximum exercise are plotted for each individual subject. Given that the pressure-time index of the diaphragm (TTdi) is the product of Ti/Ttot and $\Delta PdT/Pdi_{max}$, iso-TTdi lines were calculated and also plotted (continuous curving lines of TTdi = 0.05, 0.10, 0.15, 0.20). Each iso-TTdi line represents all combinations of $\Delta PdT/Pdi_{max} \times Ti/Ttot$ that produce that given TTdi

9.11 1 Pressure Duty Cycle (Ti/Ttot)

The same TI/Ttot used in the above TTim analysis were also utilized for the plotting

of the TTdi relationship

9.11.2 APdi/Pdimax

At quiet breathing, the athletic and normal groups demonstrated similar mean $\Delta PdT/Pdi_{max}$, 0.04 ±0.02 and 0.03 ±0.01 while the patient group presented with significantly higher mean $\Delta PdT/Pdi_{max}$ at quiet breathing, 0.12 ±0.06. P3 presented with an extremely high $\Delta PdT/Pdi_{max}$ at quiet breathing, 0.24. At maximum exercise, no significant difference existed between the mean group values of the athletes, normals and patients. Within specific groups, there was considerable variability in the $\Delta PdT/Pdi_{max}$ achieved at maximum exercise. A3,A4,N1,N2 presented with similar mean $\Delta PdT/Pdi_{max}$.0.11±0.02. A1 and N3 presented with similar lower $\Delta PdT/Pdi_{max}$ of 0.06 and 0.07 at maximum exercise. N4 presented with a high $\Delta PdT/Pdi_{max}$ of 0.19. Within the patient group a large variation existed between individual subjects. P2 and P6 (both low responders) presented with similar mean $\Delta PdT/Pdi_{max}$ as A3,A4,N1,N2, of 0.11 (P2) and 0.12 (P6). At the opposite extreme, P4 demonstrated a mean $\Delta PdT/Pdi_{max}$ of 0.43 at maximum exercise.

9.11.3 TTdi

At quiet breathing, the athletic and normal groups demonstrated similar mean TTdi's of 0.01 \pm 0.01 and 0.01 \pm 0.00. The patient group demonstrated significantly higher mean TTdi at quiet breathing, 0.04 \pm 0.02, p<0.05. At maximum exercise, the athletic, normal and patient groups demonstrated similar mean TTdi's, 0.05 \pm 0.01, 0.06 \pm 0.03 and 0.08 \pm 0.05 respectively.

Individual subjects within each group demonstrated varying mean TTdi's at maximum exercise. A1 and N3 both presented with the lowest TTdi's at maximum exercise, measured at 0.04. Individual patients also demonstrated a wide range of TTdi values. For example, P2 achieved only a maximum TTdi of 0.04 whereas P4 achieved a maximum TTdi of 0.20, in excess of the fatigue threshold of 0.15.

9 12 The Relationship of Dyspnea to TTim and TTdi

The relationship of dyspnea to TTim and TTdi is plotted in a combined graph in **fig 9.9.** Please see **Table 9 5** for definitions of symbols and lines. On the left graph, Dyspnea is plotted versus TTdi. The right side of the graph represents the relationship between Dyspnea and TTim. Mean values are plotted for individual subjects at quiet breathing and at maximum exercise. The Δ dyspnea (the change between quiet breathing and maximum exercise) response to Δ changes from quiet breathing to maximum of TTim and TTdi are examined

9 12.1 Dyspnea versus TTim [fig 9 9 right]

9.12.1.1 ΔDyspnea Response to ΔTTim of Athletes

Athletes presented with similar Δ dyspnearesponses to Δ TTim, group mean of 46.56 ± 7.81, with a range of individual subject values between 35.74 (A1) to 53.92 (A3).

9.12 1.2 Dyspnea Response to ATTIM of Normals

The mean Δ dyspnea response to TTim of the normal group was not statistically different from the athletes, 54 67 ±16.17, p>0.05 Within the normal group, a large variability existed with a range of mean Δ dyspnea responses between 36 88 (N3) to 75 59 (N1)

9.12.1.3 ADyspnea Response to ATTIM of Patients

As a group, patients presented with a similar Δ dyspnea responses to Δ TTim when compared to the athletes and normals, 50.37 +32 40, p>0.05. When the responses are grouped into the previously defined "low responding" patients (P1,P2,P4,P5,P7) v "high responding" patients (P3,P6) a differences did exist. Low responding patients demonstrated much lower Δ dyspnea responses to TTim, 33 19 ±11 34, when compared to high responding patients 93 30 ±24 99, p<0.05. The Δ dyspnea responses of the low responding patients were similar to that of A1 and N3 whereas the high responding patients demonstrated similar responses to N1 and N2



Figure 9.9 Dyspnea is plotted versus TTdi and TTim Points represent values at quiet breathing and maximum exercise for each individual subject. See text for description.

9.12.2 Dyspnea versus TTdi [fig 9.9 left]

9.12.2.1 ADyspnea Response to ATTdi of Athletes

The athletic subjects demonstrated a wide variation in Δ dyspnea responses to Δ TTdi. Two athletes (A1,A2) presented with very high Δ dyspnea responses (324.61 and 8620.00). This was due to the fact that TTdi had increased very little at maximum exercise from that measured at quiet breathing. The remaining two athletes (A3,A4) presented with Δ dyspnea responses to TTdi of 116.29 and 155.3

9.12.2.2 ADyspnea Response to ATTdi of Normals

The mean Δ dyspnea response to Δ TTdi of normals was 164.84 ±53 98, similar to A3,A4 but much lower than that of A1 and A2

9.12.2.3 ADyspnea Response to ATTdi of Patients

A large variance in Δ dyspnea responses to Δ TTdi also existed within the patient group. In general, P2 and P3 exhibited very small increases in TTdi when maximum exercise was compared to quiet breathing, thus their Δ dyspnea responses were very high (4046 24 and 761 14), similar to A1 and A2. The remaining patients presented with maximum dyspnea responses ranging from 28.68 (P4) to 281 69 (P7).

9.13 Comparison of the Δ Dyspnea Response to Δ TTim versus Δ TTdi

Comparing individual subject's Δ dyspnea response to Δ TTim versus Δ TTdi, some general observations can be made. The two athletes with the lowest Δ dyspnea response to Δ TTim (A1,A2) presented with the largest Δ dyspnea response to Δ TTdi. N3 who presented with a low Δ dyspnea response to Δ TTim also presented with a reasonably high Δ dyspnea response to Δ TTdi. In the patient group, P1,P2,P5 and P7 (all defined low dyspnea response to Δ TTdi. In the patient group, P1,P2,P5 and P7 (all defined low dyspnea response to Δ TTdi. In contrast, both patients termed high responders (P3,P6) demonstrated a high Δ dyspnea response to both Δ TTdi and Δ TTim. Conversely, P4, a low responder, exhibited a low Δ dyspnea response to both Δ TTim and Δ TTim.

9.14 The Mechanics of Breathing' A Qualitative Analysis

Muscular coordination and breathing mechanics are qualitatively evaluated through the use of both Macklem diagrams, plotting Pes versus Pga, and Konno Mead diagrams, plotting rib cage versus abdominal walf movement (RC versus AB) Representative examples of the observed breathing patterns during exercise at an iso-dyspinea level of 5 will be presented for the athletes, normals and patients. Iso dyspinea is defined as the same numerical value of dyspinea, thus an iso-dyspinea level of 5 was chosen to permit a standardized comparison between subjects. All subjects experienced this level of dyspinea at some point in time during their progressive exercise test.

The following analysis expands upon the concept of "low responding" and "high responding" subjects. This is based upon the already observed differences in the response of dyspnea to changes in pleural pressure ($\Delta Pes/Pcap$) of individual subjects (see 9.7.2)

This analysis addresses the question of whether breathing coordination and mechanics are explanatory mechanisms for observed variations in the dyspinea response of low responders versus high responders

In order to qualitatively assess and compare mechanics and coordination between subjects, standardized timing is presented on all graphics. The definitions of all timing symbols and lines used on all of the following graphics is detailed in Table 9.6. This method of analysis will be used throughout the remainder of the results section unless otherwise specified. It should also be noted that the units of measurement for both RC and AB displacement are referred to as "Arbitrary Units" on all graphics. This is not meant to indicate that changes in RC and AB excursions presented are not quantitatively equivalent. Delta changes in amplitude are equal in unit measurement. The units of measurement, however at individual data points cannot be compared. For example, a 2 on the RC axis does not necessarily correspond to a 2 on the AB axis. A negative baseline value does not necessarily indicate a change in compartmental EEV compared to rest. Rather if a change downward is seen in the baseline value from one breath to another, this would represent a decrease in compartmental EEV.



Definitions of Symbols Used in the Qualitative Analysis

	The onset of the negative inspiratory pressure swing (0 sec)
٠	5 % of the Ttot of that individual breath
0	10% of the Ttot of that individual breath
Δ	25% of the Ttot of that individual breath
	The end of inspiratory flow (ie. the end of inspiration)

Definitions of Lines Used in the Qualitative Analysis

Macklem Diagrams

- Solid Line () represents a Pes of 0 cmH20
- Dashed Line (-----) represents an iso Pdi =0 line

Konno-Mead Diagrams

Dashed Line (-----) represents an isovolume line

Table 9.6

9.14.1 Mechanics of Breathing of Athletes versus Normal Subjects

All athletes have previously been defined as low dyspnea responders to $\Delta \text{Pes/Pcap}$ Meanwhile, in the normal group, 2 subjects (N3,N4) presented with a low dyspnea response to $\Delta \text{Pes/Pcap}$ and two normals (N1,N2) presented with a high dyspnea response to increases in $\Delta \text{Pes/Pcap}$

9.14.1.1 Low Responding Athletes versus High Responding Normals

Macklem diagrams (Pes v Pga) are presented for a representative breath of a low responding athlete (A1) [fig 9 10 2] and high responding normal (N1) [fig 9 11 a]. The corresponding inspiratory flows, Pes and Pga swings are also individually plotted as a function of time for the same breach for both A1 [fig 9 10 b] and N1 [fig 9 11 b]. The Konno-Mead diagrams (RC v AB) for the same breach of A1 and N1 are presented in fig 9.10 c and fig 9.11 c respectively. The corresponding RC and AB displacements as a function of time are individually plotted for A1 [fig 9.10 d] and N1 [fig 9.11 d] respectively.

Comparing the Macklem diagrams of A1 [fig 9 10.a] and N1 [fig 9 11 a], it is observed that the athletic subject begins breathing from a much higher, positive end expiratory pleural pressure (EEP) when compared to the normal subject. However, within 5% of the total breath time, the athlete has achieved a large negative Pes swing, coinciding with a large negative Pga swing Pes has reached 0 cmH20, whereby inspiratory flow begins In contrast, the high responding normal has only achieved a very small negative pleural and gastric pressure swing in the first 5% of Ttot. Within 10% of the Ttot, the athlete has achieved peak negative Pes, peak negative Pga and reached peak inspiratory flow [fig 9.10.b] In contrast, the normal subject has only achieved a small portion of his maximum negative Pes swing, relatively no change in Pga has occurred and inspiratory flow is still accelerating [fig 9 11 b] The athlete, at 25% of Ttot, demonstrates both Pes and Pga which are relatively unchanged from their peak negative values and the maintenance of a constant peak inspiratory flow. In contrast, the normal, is still generating a negative Pes. swing, Δ Pga is relatively unchanged and inspiratory flow has just now reached its maximum value From 25% Ttot to the end of inspiration, the athlete demonstrates both Pes and Pga swings in the positive direction, whereas the normal subject demonstrates a large positive





Δ Pga and a much smaller positive Δ Pes

Analysis of the corresponding Konno-Mead diagram [fig 9.10.c] and separate RC and AB displacements v time plot [fig 9 10 d] for A1, demonstrates a predominantly AB pattern of breathing In contrast, N1, [fig 9.11 c], [fig 9 11.d] presents with a predominantly RC pattern of breathing Comparing the timing of RC and AB displacement, within the first 5% and 10% of Ttot, A1 demonstrates a large outward abdominal displacement coinciding with very little outward RC movement, whereas N1 presents with both relatively small outward RC and AB movement. From 10% to 25% Ttot, A1 demonstrates similar RC and AB expansion, whereas N1 demonstrates large RC displacement with very little AB expansion. From 25% Ttot to the end of inspiration, RC and AB displacement is still increasing for A1 and maximum displacement is achieved at end inspiration. In contrast, the normal subject is still exhibiting a predominant pattern of RC expansion and the AB actually begins to move inwards prior to the end of inspiration.

9 14.1 2 Low Responding Athlete versus Low Responding Normal

Macklem diagrams (Pes v Pga) are presented for a representative breath of a low responding athlete (A4) [fig 9 12 a] and a low responding normal (N4) [fig 9.13 a] The corresponding inspiratory flows, Pes and Pga swings are individually presented as a function of time for A4 [fig 9 12 b] and N4 [fig 9.13 b] The Konno-Mead diagrams (RC v AB) for the same breaths of A4 and N4 are presented in fig 9 12 c and fig 9.13.c respectively In addition, individual RC and AB displacements versus time for A4 [fig 9 13 d] are presented

Comparing the Macklem diagrams of A4 [fig 9 12.a] and N4 [9 13 a], it is observed that both subjects begin breathing from similar positive end pleural pressures (EEP). Within 5% of Ttot, A4 has achieved a large negative pleural pressure swing, corresponding to a negative gastric pressure swing and inspiratory flow has almost reached its peak value. N4 presents with a very rapid negative pleural pressure swing, reaching 0 cmH20, however, no change in Pga is observed (Δ Pga =0). Inspiratory flow has just begun. Within 10% of the Ttot, A4 has almost reached peak negative Pes, peak negative Pga and peak inspiratory flow [fig 9 12.b]. Comparing N4 [fig 9 13 b], despite peak inspiratory flow being maximum,





only a small portion of the maximal Pes swing has been achieved and the Pga swing is in the positive direction. The athlete, from 10% to 25% of Ttot, demonstrates a relatively unchanged negative Pes swing, whereby Pga is moving in the positive direction. Inspiratory flow is maintained at a constant peak value. In contrast, the normal subject is still experiencing a negative Pes swing, and Δ Pga and peak inspiratory flow are relatively stable. From 25% Ttot to the end of inspiration, the athlete demonstrates both positive Δ Pes and Δ Pga. In the normal subject, Pes reaches its maximum negative value at the end of inspiration, and the Δ Pga has reversed to a negative direction.

Analysis of the corresponding Konno-Mead diagram [fig 9 12 c] and separate RC and AB displacements v time [fig 9 12 d] for A4, demonstrates large AB displacement outwards from 0 to 10% of Ttot, with concurrent RC paradoxical movement inwards. N4 demonstrates a similar AB breathing pattern, also with RC paradoxical movement inwards [fig 9 13.c] and [fig 9.13 d] From 10% Ttot to the end of inspiration, both subjects demonstrate continuing AB displacement outwards now accompanied with similar outward RC movement. At end inspiration, both RC and AB are at maximal outward displacement

9.14.2 Mechanics of Breathing in COPD Patients

Within the patient group studied, 5 patients (P1,P2,P4,P5,P7) were defined as low dyspnea responders, based upon their dyspnea response to Δ Pes/Pcap, whereas the remaining two patients (P3,P6) were defined as high dyspnea responders (see 9.7.2)

9.14.2.1 Low Responding Patients versus High Responding Patients

Macklem diagrams (Pes v Pga) are presented for a representative breath of a low responding patient (P2) [fig 9 14 a] and a high responding patient (P3) [fig 9 15 a] The corresponding inspiratory flows, Pes and Pga swings are also individually plotted as a function of time for P2 [fig 9 14 b] and P3 [fig 9 15 b] The Konno-Mead diagrams (RC v AB) for the same breath of P2 and P3 are presented in fig 9 14 c and fig 9 15 c respectively. The corresponding RC and AB displacements as a function of time are individually plotted for P2 [fig 9 14 d] and P3 [fig 9.15 d] respectively.

Comparing the Macklem diagrams of the low responding patient (P2) [fig 9 14.a]





and high responding patient (P3) **[fig 9 15.a]**, it is observed that P2 begins inspiration from a very large positive end expiratory pleural pressure (EEP) whereas P3 begins inspiration from 0 cmH20 Pes. Within 5% of the Ttot, P2 has achieved a very large negative Pes swing, with a parallel large negative Pga swing, with the corresponding Pes v Pga plot lying very close to an iso Pdi=0 line. Pes has reached 0 cmH20, whereby inspiratory flow begins **[fig 9 14 b]** In contrast, P3 experiences a large negative Δ Pes, with relatively no change in Δ Pga with the first 5% of Ttot **[fig 9 15 b]**. Within 10% of Ttot, P2 has achieved peak negative Pes, peak negative Pga and reached the peak of flow acceleration. Similarly, P3 has achieved peak negative Pes and peak inspiratory flow, however, Δ Pga is still relatively unchanged (Δ Pga-0). From 10% to 25% of the Ttot, P2 demonstrates a positive Δ Pes with a very slight positive Δ Pga. P3 also demonstrates a positive Δ Pes, however a very small negative Δ Pga. Inspiratory flow is decreasing for both subjects. From 25% Ttot to end inspiration, P2 demonstrates a relatively unchanging Δ Pga and slightly positive Δ Pes, whereby P3 demonstrates parallel unchanging Δ Pes and Δ Pga

Analysis of the corresponding Konno-Mead diagram [fig 9 14 c] and separate RC and AB displacements v time plot [fig 9 14 d] for P2, demonstrates a predominantly AB pattern of breathing In contrast, P3, [fig 9.15 c], [fig 9 15.d], presents with a predominantly RC pattern of breathing Comparing the timing of RC and AB displacements, within the first 5% of Ttot, P2 demonstrates a large outward AB movement with simultaneous inward RC paradoxical movement. P3 demonstrates very little RC or AB displacement. From 5% to 25% of Ttot, continuing large AB outward movement is observed for P2 with very little concurrent RC displacement. P3 presents with exactly the opposite pattern RC displacement is largely outward whereas AB is moving inwards in a paradoxical fashion. From 25% Ttot to end inspiration, P2 demonstrates outward RC displacement, reaching peak maximal outward displacement at end inspiration. In contrast, only from 25% Ttot to end inspirate outward AB movement coinciding with the continuing RC expansion.

CHAPTER 10 DISCUSSION

10.1 Limitations of Study Design

A major limitation of this study's design is the limited number of subjects being evaluated. While a total of fifteen subjects is appropriate in overall analysis, the division of this sample population into three distinct smaller groups poses a problem in making any statistical conclusions concerning the population general

The purpose of this study, however, was not to take the underlying statistical results and make specific inferences and statistical conclusions about the general population. What this study did attempt to do was to address differences both between and within the athletic, normal and patient groups studied and to identify possible parameters (both quantitative and qualitative) which could explain observed differences. This study also dealt with the qualitative analysis of respiratory muscle coordination and attempted to add a qualitative interpretation to the results observed for each subject studied.

A large variability in measured values is to be expected when one is performing psychophysical scaling tests. Although all subjects' data was included in group mean values, if was often observed that one subject (for example N3 of the normal group), consistently presented data which was different from the remaining subjects within his group. Although no statistical significance can be ascertained given the small sample sizes, these differences were pointed out in the respective results sections (i.e. section 9.2.2) to demonstrate that, in the case of N3, he appeared to present data which was more similar to the athletes than to the normal subjects. This point is important given the purpose of this study.

10.2 The Perception of Dyspnea during Exercise

an any study requiring a subject to perceive and scale their levels of breathlessness during exercise, there is always the possibility that leg fatigue, pain or other motivational factors may influence the subject's estimation of the investigated sensation. Indeed, motivation and psychological factors have been demonstrated to affect an individual's perception to various stimuli [229]. These factors are very difficult to control for, therefore It is plausible that they may have played a role in individuals' dyspnea responses during exercise. In an attempt to minimize these factors, subjects were given very specific instructions to exclude any other sensations which might arise during the exercise test and to specifically attempt to scale "how short of breath" they were. They were also encouraged throughout the exercise test.

At maximum exercise, each groups' mean dyspnea ratings were similar to those previously reported [233,247], however, no subject reached a rated dyspnea level of 10, maximal on the Borg scale. This fact is not surprising in the normal group, given that leg fatigue would most likely impose a significant limiting factor to the achievement of high levels of exercise on the cycle ergometer. Hence, the achievement of high dyspnea levels could have been significantly limited by the normal subjects' inabilities to reach high levels of work. Athletes did demonstrate significantly longer endurance times when compared to the normal subjects and achieved significantly higher levels of workloard (in watts) [section 9.2.1], yet they reached similar maximum mean dyspnea values. COPD patients would be expected to also achieve high levels of dyspnea perception despite lower levels of work and VE given the increases in work of breathing due to their diseased state. Interestingly though, all groups presented with similar mean dyspnea levels at maximum exercise and maximum dyspnea scores of 10 were not observed.

There are several explanations why maximum dyspnea scores were not achieved. A rating of 10 (maximal dyspnea) was described to the subjects as representing the worst possible shortness of breath they have ever experienced, therefore it is possible that this explanation in itself biased subjects not to perceive a score of 10. Also, it has been demonstrated that in general, subjects tend to utilize the Borg scale over a narrower range when compared with open scaling methods, possibly due to the effect of the verbal descriptors in the Borg scale imposing a certain threshold of sensation intensity at each level which must be exceeded before proceeding to the next digit. Furthermore, the Borg scale demonstrates an inherent ratio bias which tends to restrict scores to the lower half of the scale [232]. A 5 on the Borg scale is verbally linked as "severe"

Subjects were asked to rate their dyspnealevels each minute, which may have been too long an interval. Each subject's maximum dyspnea score was recorded for only the final

*complete*minute of exercise The final partial minute of exercise might have produced rapid increases in dyspnea perception, but the recording interval would have been insensitive to this.

Previous investigators have proposed that COPD patients demonstrate blunted sensitivity when estimating added loads to breathing [182] while others contend that normal perceptual mechanisms exist [210] If COPD patients do in fact experience blunted sensitivity to added loads, an analogy could be drawn to their ability to perceive dyspinea. This blunted sensitivity most probably would be a direct effect of the loading history of the respiratory system of individual subjects.

In the present study, three distinct subject groups were studied, all presenting with three very distinct respiratory loading histories. Elite athletes undergo daily endurance training to their maximum. Therefore they experience a type of "chronic" loading of their respiratory system, yet present with normal pulmonary and respiratory muscle function to overcome these loads imposed by physical activity. Normal sedentary subjects do not experience daily imposed loading of their inspiratory muscles when compared to athletes, yet they too present with normal lung and respiratory muscle function. In contrast, COPD patients experience chronic loading of their respiratory system, dependent on the degree of obstruction, and demonstrate limitations in both pulmonary and respiratory muscle function.

10.2.1 Dyspnea versus Work

Based upon the results of this study, patients did perceive significantly higher dyspnea levels than either the normals or athletes for a given level of absolute work measured in watts. However, when individual subjects' achieved workloads were normalized as a percentage of the maximum workload that subject attained, no difference existed between any group in dyspnea levels measured [see section 9.2.2]. Differences, though, did exist in dyspnea responses to increasing normalized work. COPD patients demonstrated lower, or brunted perceptual sensitivity to dyspnea with increasing normalized work. Inferring from this observation, though, that COPD patients have blunted perceptual sensitivity is premature.

No evidence by previous investigators has demonstrated that dyspnea is a direct consequence of workload. Rather, studies have focused on the dyspnea relationship to the ventilatory levels achieved during an exercise test.

10.2.2 Dyspnea versus Ventilation

As previously demonstrated by other investigators [165,240,249,250,251, 252,253,254] these results also display a strong correlation between dyspnea perception and ventilation, and confirm that for a given level of VE, COPD patients experience higher intensities of breathlessness. However, a large inter-subject variation with considerable overlap between individuals of each of the three groups existed when the dyspnea responses to ventilation (VE/FVC and VE/MVV) were examined. Patients presented with significantly higher mean VE/FVC and VE/MVV at all levels of absolute work, yet when workload was normalized as a percentage of maximum, no differences existed between observed VE/EVC or VE/MVV for a given level of normalized work. Where differences were observed was in the rate of change of the ventilatory response of the groups. Patients presented with significantly lower mean rates of increase in both VE/FVC and VE/MVV to increasing workload, normalized as a percentage of maximum. This observation is not surprising given that COPD severely limits ventilatory increases during exercise

What was surprising was that when dyspnea was examined, no differences in the dyspnea response to changes in either VE/FVC or VE/MVV were observed between the athletic, normal or patient groups, appearing to contradict the proposals that COPD patients demonstrate blunted perceptual sensitivity. Similar to previously reported studies, though, a lot of within group variability did exist in observed dyspnea responses. Indeed, some individual patients actually presented with higher dyspnea responses to changes in VE/FVC than either athletes or normals.

Although COPD patients demonstrated similar dyspnea responses to ventilation, the loads placed upon the inspiratory muscles to achieve a given level of ventilation are significantly increased. The mechanical work of breathing is increased whereby larger pleural pressure changes are necessary to overcome the increases in both the frictional and elastic work of breathing.

10.2.3 Dyspnea versus Pleural Pressure

To further investigate the idea of perceptual blunting, the relationship of dyspinea to changes in $\Delta \text{Pes/Pcep}$ was addressed if dyspinea is directly related to the outgoing motor command to the inspiratory muscles as previously hypothesized, then the concept of perceptual blunting should not be applied to studies evaluating the relationship of dyspinea versus ventilation but rather, the changes in dyspinea which occur with the increasing pressure changes required to bring about that level of ventilation.

As a group, the patients did not demonstrate differences in dyspnea sensitivity to Δ Pes/P_{cap} when compared to the athletic group but they did demonstrate lower sensitivity when compared to the two normal subjects defined as high dyspnea responders. As discussed in the results section 9 7.2 3, though, a lot of variation in dyspnea response did exist within the patient group. This variation is not surprising, given the expected variability recognized when evaluating any perceptual sensitivity. What was interesting was that within the patient group, some patients presented with extremely low dyspnea responses to changes in pressure (P1,P2,P4,P5,P7) and did appear to demonstrate a blunted dyspnea response to increases in Δ Pes/P_{cap} when compared to all other subjects. Two patients, though, did not demonstrate this apparent perceptual blunting, but instead demonstrated similar dyspnea responses as the athletes and low dyspnea responding normals in conclusion, while it appears that some COPD patients do appear to demonstrate perceptual blunting of dyspnea, others respond as normal subjects. Therefore, this study does not confirm that COPD patients as a group demonstrate blunted perceptual sensitivities to dyspnea.

One limiting factor of this study lies in the age disparity between each group. A decline in the perceptual sensitivity of added elastic and resistive loads has been demonstrated with increasing age [208,209,210], however, it is unknown if a similar age-related blunted sensitivity exists for dyspnea perception. All of the COPD patients studied were signit in the older than both the athletes and normals. Despite this, though, some patients demonstrated very similar dyspnea perceptions when compared to some normal and athletic subjects. Others, though, did appear to demonstrate perceptual blunting (when dyspnea was related to $\Delta Pes/P_{cap}$ as discussed above). Although age disparity could be

a possible explanation for the lower dyspnea responses observed in the apparent perceptually blunted patients, it fails to explain the normal or even enhanced dyspnea responses observed in other COPD patients

The normal subjects were older than the athletes, yet two normals were defined as high responding subjects. Similarly, the high dyspnearesponding patients were significantly older than the athletes and normals, yet displayed dyspnea responses similar to both the athletes and low responding normal subjects.

In conclusion, some COPD patients' sensitivities to increasing dyspnea appears blunted when dyspnea is examined as a function of pressure changes; however, the observed differences in response when all subjects (athletes, normals and patients) are considered cannot be explained solely by age disparities

10 3 Factors Limiting Exercise

10.3.1 Ventilation

As previously discussed in the literature review, the ventilatory system is not usually regarded as a limiting factor of exercise in normal subjects, however in patients with COPD, their severely limited ventilatory capacity will result in a reduced exercise capacity [89,91,93,100,294] Patients demonstrated similar mean VE at quiet breathing when compared to athletes and normals, similar to results of previous investigators [90] Throughout the progression of the exercise test, however, further increases in VE were significantly limited

Normalizing VE as a percentage of the FVC takes into account the differences in both lung size and ventilatory limitations imposed on the patients by their disease. Patients' VE/FVC at rest was significantly higher than both athletes and normal. However, at maximum exercise VE/FVC was similar to the normals yet lower than that of the athletes. Examining VE/FVC as a function of the % of maximum achieved work, it was demonstrated that although presenting with similar VE/FVC for a given % work throughout most of the exercise test, the rate of increase of VE/FVC was still significantly reduced.

Using the estimated maximum voluntary ventilation (MVV) to indicate the maximum

ventilatory capacity of the respiratory system, patients presented with significantly lower MVV when compared to both athletes and normals. One limitation of this estimated MVV was that it was derived mathematically by utilizing the equation, FEV₁*42. The FEV₁ is not limited by reduced respiratory muscle strength whereas a true measured MVV may be. Therefore, the lower MVV's observed in the patients may have been a direct consequence of solely their obstruction and not necessarily due to reduced respiratory muscle strength. The measured VE of patients at rest accounted for a mean value of 32.2% of their ventilatory capacity, increasing to a high of 81 ±33% at maximum exercise. Thus, patients were utilizing a much higher % of their ventilatory capacity and often exceeded their maximum sustainable ventilation. In contrast, both athletes and normal subjects, were working at, but not exceeding their maximal sustainable ventilation of approximately 60% of their predicted MVV at maximum exercise.

10.3.2 The Inspiratory Muscles

The role of the inspiratory muscles as potential factor limiting factors of exercise endurance must also be considered. The studied patients presented with moderate to severe airway obstruction, based upon the observed reductions in their measured FEV₁/FVC, and were extremely hyperinflated at rest, indicated by the disproportionate increase in the measured RV/TLC. As previously discussed, hyperinflation compromises the ability of both the inspiratory intercostals and the diaphragm to generate the necessary force for tension development.

Utilizing the measured maximum inspiratory pressure (MIP) at FRC as an index, the patients demonstrated significant reductions in inspiratory muscle strength compared to both athletes and normals. Also, the patients' abilities to generate maximum transdiaphragmatic pressures were significantly limited, indicative of their hyperinflated state and subsequent diaphragmatic inefficiencies.

In conclusion, these results indicate that although differences in lung volumes (re patients demonstrate reduced VC due to large increases in RV as well as variations due to body size) and the airway obstruction characteristic in COPD do account for some of the observed limitations to increasing VE in the COPD patients, other contributing factors must be considered. Ventilatory limitations, significant reductions in inspiratory muscle strength, reduced ventilatory reserve coupled with dyspnea probably act together to play a major role in the observed exercise limitations of COPD patients.

When considering exercise limitation of normal and athletic subjects, on average they had just achieved their maximum sustainable ventilation at the same time as maximum exercise was reached, thus it appears that ventilation alone was not a limiting factor for these subjects - Indeed, peripheral limb fatigue and dyspnea may have played significant roles in the limitation of exercise

One major limitation to the above interpretation of possible limiting factors of exercise lies in the methodology of this study. Cardiovascular data did not play a role in the analysis given that the main issue being addressed was the role of respiratory muscle coordination to dyspnea. Therefore parameters such as VO2, VCO2, SaO2, the anaerobic threshold and exercise induced hypoxemia were not measured and inferences concerning their role in limiting exercise cannot be commented upon.

10.4 Mechanisms of Dyspnea

10 4.1 The Relationship of Dyspnea to Pressure Changes

When evaluating possible mechanisms which underlie dyspnea perception, it can be noted that a relationship between dyspnea and ventilation does exist for individual subjects. It is concluded, though, that dyspnea is not a direct function of the actual ventilation or percentage of capacity being utilized per se. Since ventilation is the final output of the respiratory system, it is reasonable to believe that dyspnea perception will be more closely linked to some mechanism involved in achieving that given ventilation. Therefore, the relationship of dyspnea to changes in both pleural and gastric pressures is examined in more detail.

Previous investigators [219,233,248,264], have demonstrated a positive linear relationship between dyspnea and pleural pressure (Pes), normalized by the MIP in both normal and COPD subjects However, utilizing MIP as the normalization factor presents

serious limitations. MIP is a measure of the static, maximum pressure the inspiratory muscles are capable of generating, usually measured at FRC

Pleural pressures measured during this study and those previously mentioned, however, were under the dynamic conditions of exercise. It has long been recognized that the ability to generate pleural pressure decreases both with increasing inspiratory flow rates [267,269] and increasing lung volumes [266]. Thus, normalizing Δ Pes measured during an exercise test by MIP would theoretically underestimate the proportion of maximum pleural pressure being expended to achieve a given ventilation.

Normalization of Δ Pes by the maximum capacity the inspiratory muscles are capable of generating at a given inspiratory flow and volume (Pcap) is an attempt to correct for this Our results demonstrate a strong, linear correlation between dyspnea and limitauon $\Delta Pes/MIP$ and between dyspnea and $\Delta Pes/P_{cap}$. Both methods of normalization result in similar observed differences in the dyspnea responses both within groups and between groups Athletes presented as low dyspnea responders, two normals presented as low and two as high dyspnea responders, while within the patients group, five patients presented with a low and two with a high dyspnea response. Comparison of the two methods of normalization, however, demonstrated that MIP normalization did underestimate individual subjects' dyspnea responses to ΔPes . This underestimation was significant in the athletic and normal groups but not in the patient group. This is not surprising given the fact that athletes and normals are able to achieve large increases in both their inspiratory flow rates and volumes to meet the demands of exercise, whereas the patients are severely limited in their ventilatory response. It should be pointed out, though, that the full negative effects of increases in inspiratory volumes which occurs during exercise in both normal and athletic subjects will be somewhat offset by their ability to reduce EELV below FRC. This would somewhat decrease the volume effects upon Pcap, but only up to a volume of approximately 10% TLC

In conclusion, Pcapis a more accurate normalization tool for both normal and athletic subjects when assessing pleural pressure changes during progressive exercise testing

10.4.2 Dyspnea and the Mechanics of Breathing

The simultaneous measurements of esophageal pressure (as an estimate of pleural pressure) and gastric pressure (as an estimate of abdominal pressure) coupled with rib cage and abdominal wall displacements provides a qualitative means of inferring both respiratory muscle mechanics and coordination. Limitations of this method of analysis lies in its inability to identify the quantitative contribution of the diaphragm and the "intrinsic" muscle groups when antagonistic muscle activity is present such as seen during exercise [98,147]

In the results section, representative breaths are presented for numerous subjects at iso-dyspnea levels of 5. Comparable results are obtained at other points during the exercise for each subject, however, for simplicity, only single breath representations are presented. The relative workload at which a subject experiences a dyspnea level of 5 is not relevant to the following discussion concerning mechanics of breathing.

All subjects demonstrated some degree of a negative gastric pressure swings upon early inspiration, except for A3 and N4 who presented with a pattern of predominantly Δ Pga = 0 during early inspiration, as exercise intensity increased. Patterns of negative gastric pressure generations have been reported in both normal subjects during exercise, during hyperventilation induced through re-breathing [145,295,296], and in asthmatic [57] and COPD patients both during exercise and at quiet breathing [98,297]. There are several possible breathing strategies which would account for negative Δ Pga swings during early inspiration [fig 10.1 a, fig 10.1 b, fib 10.1 c].

The first would be an inspiratory effort initiated by active intercostal/accessory muscle contraction, where the negative pressure generated in the thorax is transmitted to the abdominal space through a passive diaphragm [fig 10.1 a] Examining the relative motion of both the RC and AB, RC expansion would be observed with a coinciding paradoxical inward movement of the AB, assuming the abdominal muscles are relaxed.

The second breathing strategy could be the passive relaxation of the triangularis sterni muscle upon early inspiration, again resulting in negative Δ Ppl being transmitted through a passive diaphragm, resulting in a negative Δ Pga swing [fig 10 1 b] A similar RC out, paradoxical AB inward motion would be observed, again assuming abdominal muscles



Figure 101 c

INSPIRATORY MECHANICS USING INTERCOSTAL & OP AUCE SCOPE SCHENE MENTERS

IN PRATERS MECHANICS U. N. ENSINE RELAXED IN OF THINK END. STERNE are relaxed To the best of my knowledge this hypothetical breathing pattern has never been demonstrated in normal subjects

The third strategy involves the employment of the abdominal muscles in an inspiratory "accessory" role. Following abdominal muscle contraction during expiration, their subsequent relaxation upon early inspiration would result in a negative Δ Pga swing [fig 10 1 c]. Thus, during expiration, abdominal muscle contraction acts to store both elastic energy and gravitational energy in the diaphragm/abdominal wall and through the subsequent relaxation upon inspiration, the gravitational potential energy of the abdominal contents being pulled caudally is sufficient to accelerate diaphragmatic descent and to decrease intrathoracic pressure. If the rib cage muscles are relaxed, an inward paradoxical movement would be observed, whereas if the rib cage volume increases or remains relatively constant, this would indicate the switching of intercostal activity from expiratory to inspiratory [98]

It has been proposed that a 0 cmH20 change in Δ Pga during early inspiration is a result of the coordination of release of abdominal muscle tone coupled with simultaneous diaphragmatic contraction against minimal abdominal impedance [298] If this were the case, outward AB movement would be observed upon early inspiration [fig 10.1.d], and consequent RC movement would be dependent on the level of activity of the inspiratory intercostal/accessory muscles

The importance of the abdominal muscles as "accessory" muscles to inspiration is now recognized. Increased expiratory abdominal muscle recruitment during exercise has been well documented in both normal and COPD subjects. Increased abdominal muscle recruitment causes a decrease in end-expiratory lung volume below the normal FRC in normal subjects by approximately 10% of TLC both during exercise to maximum and during inspiratory. loading [39,79,299,300,301,302]. In contrast, it has been demonstrated that expiratory muscle recruitment during exercise in COPD patients is largely ineffective in reducing EELV, due to expiratory flow limitations imposed by the obstruction [89,97,98,122], and dynamic hyperinflation results in increases in end expiratory pressure

It is argued, therefore, that in order for the abdominal muscles to adopt an "inspiratory" role, increasing positive end expiratory pressures in normal subjects is an

unavoidable occurrence. In normal subjects, the reduction of resting lung volume below FRC permits the elastic outward chest wall recoil to aid the proceeding inspiration. In contrast, though, the acute hyperinflation which occurs during exercise in the majority of COPD patients acts to place the inspiratory muscles at an even more shortened, disadvantageous initial length position for the proceeding inspiration

Most investigators have stressed the role of expiratory abdominal muscle recruitment on optimizing resting diaphragmatic length and improving its inspiratory efficiency to expand the rib cage through an increase in the zone of apposition [7,32,57,145,295,296,303]. However, few investigators have emphasized the role of inspiratory passive abdominal muscle relaxation in actually contributing to the inspiratory pressure generation [98]. Such a function would prove to be of great significance, especially in COPD patients, given that reduced inspiratory muscle strength coupled with the mechanical inefficiency of requiring larger pleural pressure swings to achieve a given ventilation, act to place their inspiratory muscles at risk to develop fatigue.

The significant role of respiratory muscle function in the perception of dyspnea has been proposed, with previous investigators hypothesizing that dyspnea is the perception of the outgoing motor command to the inspiratory intercostals/accessory muscles [222,245,259,271,284,285] Any mechanical readjustment of the breathing pattern which acts to "load" or increase the central motor output towards both the inspiratory intercostals/accessory muscles has been shown to result in higher levels of dyspnea [271,273,274]

Theoretically, therefore, any readjustment of respiratory muscle activity which would serve to decrease the load placed upon the inspiratory intercostal/accessory muscles would therefore reduce the amount of central motor output directed towards these muscles and would theoretically result in a lower dyspnea perception. Passive abdominal relaxation upon early inspiration would serve such a purpose

Indeed from our results, it appears that during progressive exercise, the utilization of abdominal muscle relaxation during early inspiration does play a significant role in reducing the dyspnea response of subjects to changes in pleural pressure. This conclusion is based upon numerous qualitative and quantitative observations made throughout the
subjects' individual progressive exercise tests

It appears that the athletes, low dyspnea responding normal subjects and low dyspnea responding patients all utilized some degree of passive abdominal muscle relaxation during early inspiration, whereas high responding normals and high responding patients did not employ this method to make a significant contribution to the necessary pleural pressure generation

All subjects studied presented with positive increases in both end expiratory pleural pressures (EEP) and gastric pressures, combined with inward RC and AB displacement, indicative of a combination of increased expiratory intercostal muscle recruitment and abdominal muscle recruitment and in the case of the COPD patients, dynamic hyperinflation

All athletes, low dyspnea responding (LDR) normal subjects and LDR patients demonstrated AB displacement outward upon early inspiration as observed in the presented Konno-Mead diagrams In addition, inward paradoxical or virtually no RC movement was observed in these abdominal breathers during the early phase of inspiration until inspiratory flow and volumes were increased. This time readjustment of the abdomen leading the rib cage and with the rib cage paradoxically moving inward upon early inspiration has been observed in normal [302] and COPD patients [98,297,304,305] during progressive exercise and loading experiments Examining the pressure generation profile, presented in Macklem diagrams, a large negative Δ Pga swing and concurrent negative Δ Pes swing is demonstrated. The observed combination of RC and AB displacement, coupled with the observed pattern of pressure generation implies that passive abdominal muscle relaxation is occurring, providing the mechanism for pressure changes. The similarity in both pressure generation profiles and RC and AB motion when athletes are compared with LDR COPD patients is striking. In the athletes and LDR patients, for example, these negative pressure swings occurred within the first 10% of the time of the breath, and upon examining the Macklem diagram, the pressure generated during this early phase lies v y close to an iso Pdi = 0 line It is therefore possible to infer that very little change in transdiaphragmatic pressure is occurring in this early phase of inspiration and that minimal diaphragmatic activation is required. In contrast, the high dyspnea responding (HDR) normals and HDR

COPD patients, despite achieving negative pleural pressure swings, demonstrated minimal gastric pressure swings during early inspiration. Examining the relative motion of the RC and AB, outward RC movement is observed upon early inspiration with no eutward movement or inward paradoxical movement of the AB. Therefore, it can be inferred that in these high dyspnea responding subjects, the RC muscles are actively contracting to generate the necessary change in pleural pressure.

Following these generalized observations, it is possible to divide early inspiratory pressure generation into two phases. The first is the "pre-inspiratory flow" phase of pressure generation and indicates the mechanisms utilized to return the respiratory system to a level whereby inspiratory flow can be initiated. Resistive and elastic loading studies have demonstrated that the perception of just detectable loads occurs very early in the inspiration, close to peak flow [179,180]. Applying this concept to the present study, the phase of pressure generation from 0 to peak flow acceleration is also explored.

All subjects presented with some degree of increases in pleural end expiratory pressures (EEP) as exercise progressed. These observed increases were due to either increased expiratory muscle recruitment and a reduction in end expiratory lung volumes below FRC, as occurred in the normal and athletic subjects, or due to the combination of increased expiratory muscle activity coupled with dynamic hyperinflation, previously observed to occur in COPD patients during exercise.

Examining the pre-inspiratory flow phase of pressure generation, it was observed that all athletes and N3 (LDR) presented with lower dyspnea responses to the positive end expiratory pleural pressures (EEP) when compared to high responding normals N1, N2 and the LDR subject N4 — Similarly, the LDR patients (P1,P2,P4,P5,P7) presented with low dyspnea responses to EEP when compared to the HDR patients (P3,P6) — Of the HDR patients, P3 actually presented with an inconsistent relationship of dyspnea to EEP while P6 presented with a response similar to N1,N2 and N4

Evaluating actual EEP measured, some athletes presented with much higher EEP when compared to the normal subjects, four of the seven patients who presented with the highest EEP were also LDR. Mechanically, increasing EEP causes work to be performed on the lung which would produce no ventilation and the subsequent high levels of

intrathoracic pressure may impose a limitation to cardiovascular function [91] It is therefore surprising that in general, the low dyspnea responders were the subjects who were faced with the largest mechanical impedance to overcome in order for inspiratory flow to begin. As previously mentioned, both athletes and normal subjects have the benefit of increased outward chest wall recoil to aid in overcoming the EEP, but the COPD patients cannot rely on such a mechanism. Passive abdominal relaxation in COPD patients may thus constitute an optimal response of the respiratory system aimed at achieving the maximum ventilation possible in the face of a severe mechanical impediment [98]. Conversely, in the athletes, abdominal muscle relaxation would be additive to the outward chest wall recoil which occurs when lung volumes are below FRC. The two coupled together would result in inspiratory mechanics which are very energy efficient.

Three of the four athletes (A1,A2,A3) and one low responding normal (N3) presented with much faster negative rates of change in gastric pressure ($\Delta Pga/\Delta t_{par}$) when compared to the remaining subjects. The one remaining athlete (A4) and low responding normal (N4) both presented with lower rates of gastric pressure change and in the case of the normal, Δ Pga was in the positive direction. However, the role of abdominal muscle relaxation, in these two subjects was still significant (as seen in the respective Konno-Mead diagrams) Abdominal muscle recruitment during expiration would not only place the diaphragm at a more lengthened position but the diaphragm would be primed for subsequent inspiratory contraction Indeed, it has been shown that the force developed by a muscle when it shortons after being stretched is greater than that developed at the same speed and length when starting from a state of isometric contraction [306] Thus as the abdominals relaxed the diaphragm was faced with optimal conditions in which to contract downwards A wide variability existed within the patient group when the $\Delta Pga/\Delta t_{PRE}$ was examined at maximum exercise As seen on Table 9.3, P2 (LDR) presented with a similar rapid rate of change in gastric pressure as seen in the athletic subjects. The remaining subjects presented with varying rates of change, however, it should be noted that the two lowest observed rates of change were by the two HDR patients (P3,P6) These two subjects also demonstrated large variations in their breath to breath measured $\Delta Pga/\Delta t_{PRF}$, whereas P4,P4,P7 (all LDR) presented with relatively consistent $\Delta Pga/\Delta t_{PRE}$ as inferred from the small standard deviations about the means P1 (LDR) demonstrated some very rapid changes in gastric pressures while in other breaths, he demonstrated relatively no change

The rate of change of negative pleural pressure ($\Delta Pes/\Delta t_{ext}$) in this pre-inspiratory flow phase also appeared to be related to the ability to utilize abdominal muscle release. In general, the athletes all demonstrated very rapid $\Delta Pes/\Delta t_{PRt}$, as did four of the five low responding patients (P1,P2,P4,P7). It is also seen that those subjects who experienced the larger development of EEP during exercise, also demonstrated faster $\Delta Pes/\Delta t_{PRt}$. It can be speculated that abdominal muscle relaxation provided an accelerative mechanism, permitting a more rapid rate of change of pleural pressure to overcome the EEP and to initiate the next inspiration.

Similar to this concept, Dodd *et al* **[98]** hypothesized that at the onset of inspiratory flow at the mouth, the diaphragm/abdomen have already acquired a descending velocity, which must be additive to that produced throughout inspiration by the release of gravitational energy. They also proposed that the elastic energy derived from thoracic gas decompression, added to the "sling shot" effect of abdominal muscle relaxation would result in a more rapid development of negative pleural pressure on inspiration than would otherwise be possible. These results appear to corroborate this hypothesis.

When comparing the maximum rate of change of pleural pressure achieved from the onset of inspiratory flow to peak flow acceleration ($\Delta Pes/\Delta t$) at maximum exercise of each subject, it is observed that three of the four athletes (A1,A3,A4) achieved very high $\Delta Pes/\Delta t$. The remaining athlete while achieving a lower $\Delta Pes/\Delta t$ than the others, was the athlete who only reached 80% of his maximum workload. When considering the ratio of $\Delta Pes/\Delta t$ to $\Delta Pes/Pcap$, the athletes were able to achieve a much faster rate of change of pleural pressure for a given % change in $\Delta Pes/Pcap$. This supports the idea that abdominal muscle relaxation may in effect "accelerate" the negative change in pleural pressure required for ventilation. In contrast, the normal subjects all presented with much lower $\Delta Pes/\Delta t$, regardless of whether they were low or high responders. A possible explanation as to why the two normals deemed "low responders" do not exhibit the same accelerative forces working on ΔPes as do the athletes lies in their level of workload achieved. Leg fatigue, motivation or other factors previously discussed may have played a limiting role in both of

these subjects and therefore they demonstrated smaller positive increases in EEP. Therefore, they would have had lower levels of abdominal muscle recruitment, coupled with their lower levels of achieved work, thus the possible accelerative force which abdominal muscle relaxation provides would be limited. The patients demonstrated similar ratios of $\Delta Pes/Pcap$ to $\Delta Pes/\Delta t$ when compared to normals and indeed, often achieved higher $\Delta Pes/\Delta t$ at maximum exercise.

It is important, however, to consider the rates of change of gastric pressure $\Delta Pga/\Delta t$ which accompany and perhaps contribute to the rate of change of pleural pressure. If abdominal muscle relaxation does play a role in reducing dyspnea perception, it would be expected that for a given increase in dyspnea, a concurrent rapid $\Delta Pga/\Delta t$ would occur. This relationship is expressed by the ratio of $\Delta dyspnea$ to $\Delta Pga/\Delta t$

Three athletes (A1,A2,A4) and one low responding normal (N3) both presented with strong relationship correlations between dyspnea and $\Delta Pga/\Delta t$ throughout the exercise test Also they demonstrated a much faster rate of change in gastric pressure for a given change in dyspnea when compared to both high responding normal subjects (N1,N2). In contrast, again A3 and N4, both low responders presented either a poor correlation or positive (ie $\Delta Pga/\Delta t > 0$) ratio. As discussed previously, these two subjects demonstrated patterns of either relatively no change in gastric pressure or positive ΔPga swings. Looking at the patients, all low responders also demonstrated similar fast rates of change in gastric pressure for a given change in dyspnea, similar to the athletes. The two patients termed high responders exhibited much smaller changes in the rate of gastric pressure change for a given change in dyspnea.

Qualitatively, looking at subjects who presented with early inspiratory passive abdominal relaxation, it was consistently observed that upon higher levels of work, the generation of peak negative pleural pressure, coupled with the rapid negative gastric pressure swing resulted in the complete acceleration of inspiratory flow to peak values in both athletes and LDR patients. If an analogy can be made to the previously mentioned loading studies **[179,180]**, the perception of dyspnea may occur very early in inspiration if this is true, early inspiratory abdominal muscle relaxation is well suited as a "dyspnea limiting" mechanism.

Researchers have demonstrated that dyspnea is not directly related to the level of diaphragmatic fatigue [264,271,272] but rather to the "loading" effects that induced diaphragmatic fatigue would impose on the inspiratory intercostal/accessory muscles. In the subjects studied, it does not appear as if any subjects achieved above threshold pressure-time indices of the diaphragm, excluding the presence of diaphragmatic fatigue. The exception to this is i?4, who exceeded the threshold level of 0.15 TTdi. This patient, however, demonstrated a strong abdominal relaxation breathing pattern. This mechanism would theoretically permit the diaphragm to generate tension with less amount of work being performed, therefore, permitting a TTdi >0.15 to be achieved without impending fatigue

When measuring the pressure-time index of the inspiratory muscles (11im), several factors must be considered. The TTim is based upon the assumption that all of the pleural pressure changes occurring are due solely to active inspiratory muscle contraction. Increased expiratory muscle recruitment during exercise would cause an over exaggeration of the measured pleural pressure swings, therefore, the effects of abdominal muscle relaxation on changes in pleural pressure must be considered when evaluating the TTim of subjects during exercise. Indeed, from the results, A1,P1,P2 and P4 all present with TTim exceeding the designated inspiratory muscle relaxation, resulting in a portion of the pleural pressure change to be generated passively. However, given the design of this study, it is impossible to discern quantitatively what the contribution to pleural pressure changes is being accomplished through abdominal relaxation and by inspiratory muscle activity.

Therefore, it appears that passive abdominal recoil not only acts to unload the diaphragm but also to unload the inspiratory muscles, permitting larger pressure generation with little risk of developing inspiratory muscle fatigue. In conclusion, it does not appear as if either diaphragmatic fatigue or inspiratory muscle fatigue played a limiting role in exercise.

10.7 CONCLUSIONS

These results indicate that dyspnea perception is directly related to respiratory muscle function. The hypothesis that dyspnea is a perception of the outgoing motor command to the inspiratory muscles (intercostals/accessories) is one which would provide an explanation for our results.

Exercise imposes demands upon the respiratory system, demands which the inspiratory muscles must meet by increasing their force generation, in order to generate the required ventilation. Thus, as exercise intensity increases, the inspiratory muscles are faced with increasing ventilatory loads. Both athletes and normals demonstrate normal respiratory muscle strength and function, whereas the COPD patient is faced with severe limitations in both endurance and function. However, when examining pressure generation and relative RC and AB motion, it is surprising to see the similarity between LDR patients and athletes.

The importance of passive abdominal muscle relaxation upon early inspiration has not been directly investigated in the past. This study, though, demonstrates that this mechanism indeed appears to play a very important role mechanically. It appears to not only unload both the diaphragm and the inspiratory intercostal/accessory muscles, preventing fatigue at high workloads, but also to provide a very strong accelerative force for pleural pressure generation. Subjects who presented with a coordinated pattern of abdominal muscle relaxation presented with lower dyspnea responses to changes in pleural pressure. It also can be hypothesized that in some COPD patients, the lower dyspnea response to changes in pleural pressures is a direct result of their adopted breathing patterns and not due to perceptual blunting of sensation. In conclusion, passive abdominal muscle relaxation appears to play a significant role in limiting the perception of dyspnea during progressive exercise.

SUGGESTIONS FOR FUTURE RESEARCH

1. Perform an epidemiological study, examining the prevalence of early inspiratory abdominal muscle relaxation during exercise in a population of athletes, normal subjects, chronic obstructive pulmonary diseased patients

2. Study the effects on dyspnea perception of abdominal muscle relaxation during another form of exercise (ie. treadmill exercise)

3. Investigate the mechanics of breathing during expiratory resistive loading in normal subjects and the relationship to perceived dyspnea

4 Investigate the effects that early inspiratory abdominal muscle relaxation has on both the timing of and electromyographic activity of the diaphragm

5 Investigate if inspiratory abdominal muscle relaxation has a similar role in the reduction of dyspnea perception during exercise in a group of patients suffering from restrictive lung disease.

REFERENCES

1 De Troyer A Respiratory Muscles In The Lung, R.G Crystal, J B. West et al (eds), New York Raven Press Ltd, 1991.

2. De Troyer A, Estenne M Functional Anatomy of the Respiratory Muscles Clinics in Chest Med 1988,9(2) 175-193

3 Jordanoglou J Vector analysis of rib movement Respir Physiol 1970;10 109-120

4 Wilson TA, Rehder K, Krayer S, Hoffman EA, Whitney CG, Rodarte JR Geometry and respiratory displacement of human ribs J Appl Physiol 1987,62 1872-1877

5 De Troyer A, Decramer M. Mechanical coupling between ribs and sternum in the dog. Respir. Physiol. 1985,59 27-34

6 De Troyer A, Estenne M, Vincken W. Rib cage motion and muscle use in high tetraplegics Am Rev Respir Dis 1986,133:1115-1119.

7 Roussos C Function and fatigue of respiratory muscles Chest 1985, 88(2):124S-132S.

8 Mead J Functional significance of the area of apposition of diaphragm to rib cage Am Rev Respir Dis 1979,119(2) 31-32

9 Mead J, Loring S Analysis of volume displacement and length changes of the diaphragm during breathing J Appl Physiol 1982,53 750-755

10 Kim MJ, Druz WS, Danon J, Machnach W, Sharp JT Mechanics of the canine diaphragm J Appl Physiol 1976;41 369-382

11 Loring SH, Mead J Action of the diaphragm on the rib cage inferred from a force balance analysis J Appl Physiol 1982,53 756-60

12 De Troyer A, Sampson M, Sigrist S, Macklem PT Action of costal and crural parts of the diaphragm on the rib cage in dog J Appl Physiol 1982,53[,]30-39.

13 Duchenne GB Physiologie des mouvements Paris Ballilliere 1857.

14 Mortola JP, Sant'Ambrogio G Motion of the rib cage and the abdomen in tetraplegic patients. Clin Sci Mol Med 1978,54 25-32

15 Estenne M, De Troyer A Relationship between respiratory muscle electromyogram and the rib cage motion in tetraplegia Am Rev Respir Dis 1985,132 53-59

16. Goldman MD, Mead J Mechanical interaction between the diaphragm and rib cage J. Appl Physiol 1973;35 197-204

17. Campbell EJM The role of the scalene and sternomastoid muscle in breathing in normal subjects. An electromyographic study J Anat 1955,89 378-386

18 Raper AJ, Thompson WT Jr, Shapiro W, Patterson JL Jr Scalene and sternomastoid muscle function J Appl Physiol 1966,21497-502

19 De Troyer A, Estenne M. Coordination between rib cage muscles and draphragm during quiet breathing in humans. J. Appl Physiol 1984,57.899-906

20 De Troyer A, Kelly S Chest wall mechanics in dogs with acute diaphragm paralysis J Appl Physiol 1982,53 373

21 Hamberger GE De respirationis mechanismo lena, 1727

22. De Troyer A, Sampson M, Sigrist S, Kelly S. How the abdominal muscles act on the rib cage. J. Appl. Physiol. 1983;54:465-469

23. Taylor A The contribution of the intercostal muscles to the effort of respiration in man J Physiol (Lond) 1960,151 390-402

24 Duran B Intercostal and diaphragmatic muscle endings and afferents. In Regulation of breathing, Part 1. Hornbein TF (ed), New York Marcel Dekker, 1981 473-540

25 De Troyer A, Sampson MG. Activation of the parasternal intercostals during breathing efforts in human subjects. J Appl. Physiol 1982,52 524-529

26. De Troyer A, Kelly S, Macklem PT, Zin WA Mechanics of intercostal space and actions of external and internal intercostal muscles. J. Clin. Invest. 1985,75.850-857

27 Decramer M, Kelly S, De Troyer A. Respiratory and postural changes in intercostal muscle length in supine dogs. J Appl. Physiol. 1986,60 1686-1691

28 De Troyer A, Ninane V Triangularis sterni a primary muscle of breathing in the dog J Appl Physiol 1986,60 14-21

29 De Troyer A, Ninane V, Gilmartin JJ, Lemerre C, Estenne M. Triangularis sterni muscle use in supine humans. J. Appl. Physiol. 1987,62 919-925

30 Estenne M, Ninane V, De Troyer A. Triangularis sterni muscle use during eupnea in humans effect of posture. Respir. Physiol. 1988,74 151-162

31 Mier A, Brophy C, Estenne M, Moxham J, Green M, De Troyer A. Action of abdominal muscles on rib cage in humans J Appl Physiol 1985,58.1438-1443

32 De Troyer A, Loring SH Action of the respiratory muscles In: Handbook of Physiology-The Respiratory System III Ch 26 1984 443-461

33 Koterba AM, Kosch PC, Beech J, Whitlock T Breathing strategy of the adult horse (Equus caballus) at rest J Appl Physiol 1988,64 337-346

34 Ninane V, Gilmartin JJ, De Troyer A Changes in abdominal muscle length during breathing in supine dogs. Respir Physiol 1988,73 31-42

35 Arnold JS, Haxhiu MA, Cherniack NS, Van Lunteren E. Transverse abdominis length changes during eupnea, hypercaphia, and airway obstruction. J Appl Physiol 1988,64 658-665

36 Farkas GA, Estenne M, De Troyer A Expiratory muscle contribution to tidal volume in head-up dogs J Appl Physiol 1989,67 1438-1442

37 De Troyer A Mechanical role of the abdominal muscles in relation to posture. Respir. Physiol 1983,53 341-353

38 Strohl KP, Mead J, Banzett RB, Loring SH, Kosch PC. Regional differences in abdominal muscle activity during various manoeuvres in humans. J. Appl. Physiol. 1981;51 1471-1476

39 Martin JG, De Troyer A The behaviour of the abdominal muscles during inspiratory mechanical loading Respir Physiol 1982;50 63-73

40 De Troyer A, Estenne M, Ninane V, Van Gansbeke D, Gorini M. Transversus abdominis muscle function in humans J Appl Physiol. 1990,68 1010-1016

41 Mahler DA, Barlow PB, Matthay RA Chronic Obstructive Pulmonary Disease Clinics in Genatric Med 1986,2(2) 285-312

42 Higgins M Epidemiology of COPD State of the art Chest 1984;85(Suppl) 3S-8S

43 Petty TL Definition, identification, assessment, and risk factors. In Chronic Obstructive Pulmonary Disease. Vol 9, Lung Biology in Health and Disease. Petty TL (ed), New York. Marcel. Dekker, Inc., 1978-1-22.

44 Tisi GM Pulmonary Physiology in Clinical Medicine, 2nd Edition Baltimore' Williams & Wilkins, 1983 1-287

45 Pratter MR, Curley FJ, Dubois J Cause and evaluation of chronic dyspnea in a pulmonary disease clinic Arch Intern Med 1989,49 2277-2282

46. Gandevia B, Hugh-Jones P Terminology for measurements of ventilatory capacity A report to the Thoracic Society Thorax 1957,12 290-292

47 Snider GL, Kory RC, Lyons HA Grading of pulmonary function impairment by means of pulmonary function tests Chest 1967;52 270-271

48 Dosman JF, Bode F, Urbanetti J The use of helium-oxygen mixture during maximum expiratory flow to demonstrate obstruction in small airways in smoker J Clin Invest 1975;55 1090-1099

49. Dunnill MS The contribution of morphology to the study of chronic obstructive lung disease Am. J Med 1974,57:506-519

50. Hyatt RE The interrelationship of pressure, flow and volume during various respiratory manoeuvres in normal and emphysematous subjects Am Rev Respir Dis 1961,83 676-683

51. Baldwin E DEF, Cournand A, Richards Jr DW Pulmonary insufficiency, III A study of 122 cases of chronic pulmonary emphysema Medicine Baltimore 1949, 28 201-237

52. Hurtado A, Kaltreider NL, Fray WW, Brooks WDW, McCann WS Studies of total pulmonary capacity and its subdivisions VI. Observations on cases of obstructive pulmonary emphysema. J. Clin. Invest. 1934;13:1027-1051

53. Pride NB, Macklem PT Lung mechanics in disease. In Handbook of Physiology. The Respiratory System III. 1986 659-691

54. O'Connell JM, Campbell AH Respiratory mechanics in airways obstruction associated with inspiratory dysphoea Thorax 1976,31 669-677

55. Marini JJ Ventilatory Management of COPD In Chemiack NS, Chronic Obstructive Pulmonary Disease, Ch 53, W B Saunders Co , Montreal 1991 495

56 Rochester DF, Arora NS Respiratory Muscle Fatigue In The Medical Clinics of North America. Symposium on Respiratory Failure Vol 63(3) Bone RC (ed), Toronto WB Saunders Co., 1983.

57. Martin J, Powell E, Shore S, Emrich J, Engel LA The role of the respiratory muscles in the hyperinflation of bronchial asthma. Am Rev Respir Dis 1980,121 441-447

58 Martin JG, Habib M, Roussos Ch, Engel LA Inspiratory muscle activity during induced hyperinflation Physiologist 1978,21:77

59 Cherniack RM The oxygen consumption and efficiency of the respiratory muscles in health and in emphysema J Clin Invest 1959,38 494-499.

60 Campbell EJM Physical signs of diffuse airways obstruction and lung distension. Thorax 1969,24 1-3

61 Braun NMT, Arora NS, Rochester DF. Force-length relationship of the normal human diaphragm J Appl Physiol 1982;53:405-412

62 Farkas GA, Roussos Ch. Adaptability of the hamster diaphragm to emphysema and/or exercise. J. Appl. Physiol. 1982,53 1263-1271

63 Farkas G, Roussos Ch. The diaphragm in emphysematous hamsters sarcomere adaptability. J Appl. Physiol. 1983,54:1635-40

64 Decramer M, De Troyer A Respiratory changes in parasternal intercostal length J. Appl Physiol 1984,57 1254-1260

65 Tobin MJ Respiratory muscles in disease Clin. Chest Med. 1988,9(2).263-286.

66 Macklem PT Respiratory muscles The vital pump Chest 1980;78:753-758

67 Roussos C, Macklem PT The respiratory muscles N Engl J Med 1982;307:786-797

68 Pardy RL, Hussain SNA, Macklem PT The ventilatory pump in exercise. Clinics in Chest Med 1984,5(1) 35-49

69 Asmussen E. Similarities and dissimilarities between static and dynamic exercise. Circ Res 1981,48 3-10

70 Shephard RJ, Blomqvist GG, Lind AR Static (isometric) exercise. Circ Res. 1981,48 179-188

71 Campbell EJM, Westlake EK, Cherniack RM Simple methods of estimating oxygen consumption and efficiency of the muscles of breathing. J. Appl. Physiol. 1957,11 303-308

72 Milic-Emili G, Petit JM Mechanical efficiency of breathing. J. Appl. Physiol 1960,15 359-362

73 Otis AB The work of breathing. Physiol. Rev 1954,34.449-458

74. Rochester DF, Arora NS, Braun NMT The respiratory muscles in chronic obstructive pulmonary disease Bull Eur Physiopathol Respir 1979,15 951.

75 Pearce DH, Milhorn HT Dynamic and steady-state respiratory responses to bicycle exercise J Appl Physiol 1977,42 959-967

76 Wasserman K, Whipp BJ, Davis JA Respiratory physiology of exercise Metabolism, gas exchange, and ventilatory control Respir Physiol 1981,23 149-211

77 Wasserman K, Whipp BJ, Koyal SN Anaerobic threshold and respiratory gas exchange during exercise J Appl Physiol 1973,35.236-243

78 Cotes J, Johnson G, McDonald A In Breathing, Porter R (ed.), Churchill London 1970:297-314

79 Lind F, Hesser CM Breathing pattern and lung volumes during exercise Acta Physiol. Scand 1984,120 123-129.

80. Hey EN, Lloyd BB, Cunningham DJC, Jukes MGM, Bolton DPG Effects of various respiratory stimuli on the depth and frequency of breathing in man Resp Physiol 1966,1:193-205

81 Bechbache RR, Duffin J. The entrainment of breathing frequency by exercise rhythm J. Physiol 1977,272 553-561

82. Folinsbee LJ, Wallace ES, Bedi JF, Horvath SM Respiratory pattern in trained athletes. In: Modelling and Control of Breathing, Whipp BJ, Wiberg DM (eds). Elsevier Science Publishing Co., Inc., 1983.

83 Jensen JI, Lyager S, Pedersen OF The relationship between maximal ventilation, breathing pattern and mechanical limitation of ventilation J Physiol 1980,309 521-532

84 Dempsey JA, Vidruk EH, Mastenbrook SM Pulmonary controlled systems in exercise Fed. Proc. 1980,39 1498-1505

85 Ogilvie CM, Stone RW, Marshall R The mechanics of breathing during the maximum breathing capacity test Clin Sci 1955,14 101-107

86 Olafsson S, Hyatt RE Ventilatory mechanics and expiratory flow limitation during exercise in normal subjects J Clin Invest 1969,48 564-573

87. Otis AB, Fenn WO, Rahn H. Mechanics of breathing in man. J. Appl. Physiol. 1950,2:592-607

88 Stubbing DG, Pengelly LD, Morse JLC, Jones NL. Pulmonary mechanics during exercise in normal males J. Appl Physiol Respirat Environ Exercise Physiol. 1980,40(3) 506-510

89 Stubbing DG, Pengelly LD, Morse JLC, Jones NL. Pulmonary mechanics during exercise in subjects with chronic airflow obstruction J. Appl Physiol Respirat Environ Exercise Physiol 1980,49(3) 511-515

90 Shephard RJ The oxygen cost of breathing during vigorous exercise. Q. J. Exp. Physiol 1966,51 336-350

91 Potter WA, Olafsson S, Hyatt RE Ventilatory mechanics and expiratory flow limitation during exercise in patients with obstructive lung disease J Clin Invest 1971,50 910-919

92 Aldrich TK, Arora NS, Rochester DF The influence of airway obstruction and respiratory muscle strength on maximal voluntary ventilation in lung disease Am. Rev. Respir Dis 1982,126 195

93 Clark TJH, Freedman S, Campbell EJM, Winn RR The ventilatory capacity of patients with chronic airway obstruction. Clin. Sci. 1969;36:307-316

94 Cherniack RM The oxygen consumption and efficiency of respiratory muscles in health and emphysema J Clin Invest 1959,38 494-499

95 Brown HE, Wasserman K Exercise performance in chronic obstructive pulmonary diseases. Med Clin North Am 1981,65 525-547

96 Jones NL, Jones G, Edwards RHT Exercise tolerance in chronic airway obstruction. Am Rev Respir Dis 1971,103 477-491

97 Grimby G, Stiska J Flow-volume curves and breathing patterns during exercise in patients with obstructive lung disease Scand J Clin. Lab Invest 1970,25303-313.

98 Dodd DS, Brancatisano T, Engel LA Chest wall mechanics during exercise in patients with severe chronic air-flow obstruction Am Rev Respir Dis 1984,129 33-38

99 Sergysels R, Degre S, Garcia-Herreros P. The ventilatory pattern during exercise in chronic obstructive pulmonary disease Bull Eur Pathophysiol Respir 1979,15 57-70

100 Leaver DG, Pride NB Flow-volume curves and expiratory pressures during exercise in patients with chronic airway obstruction. Scand J Respir Dis [Suppl] 1971,77 23-27

101. Levison H, Cherniack RM Ventilatory cost of exercise in chronic obstructive pulmonary disease J Appl Physiol 1968,25 21-27

102. Dempsey JA, Hanson PE, Mastenbrook SM Arterial hypoxemia during exercise in highly trained runners. Fed Proc 1981,40 396(Abstr)

103. Dempsey JA, Hanson P, Pegelow D, Claremont A Limitations to exercise capacity and endurance pulmonary system. Can J Appl Sport Sci 1982;7.4-13

104 Rowell LB, Taylor HL, Wang Y, Carlson WS Saturation of arterial blood with oxygen during maximal exercise J Appl Physiol 1964,19 284-286

105 Grimby G, Saltin B, Wilhelmsen L Pulmonary flow-volume and pressure-volume relationship during submaximal and maximal exercise in young well-trained men Bull Physiopathol Respir 1971,7 157-168

106 Roussos C, Fixley M, Gross D, Macklem PT Fatigue c inspiratory muscles and their synergic behaviour J Appl Physiol Respirat Environ Exercise Physiol 1979,46(5) 897-904

107 Nattie EE, Tenney SM The ventilatory response to resistance unloading during muscular exercise Respir Physiol 1970,10 249-262

108. Otis AB, Bembower WC Effect of gas density on resistance to respiratory gas flow in man. J Appl Physiol 1949,2 300-306

109. Ward SA, Poon C, Whipp BJ Does turbulent airflow constrain compensatory hypernea during metabolic acidosis in exercise? Med Sci Sports Exer 1980;12.123(Abstr)

110. Bye PTP, Farkas GA, Roussos Ch. Respiratory factors limiting exercise. Ann. Rev. Physiol. 1983,45.439-451

111. Agostini E, Fenn WO Velocity of muscle shortening as a limiting factor in respiratory airflow J Appl Physiol 1960,15 349-353

112 Kim MJ, Druz WS, Danon J. Mechanics of the canine diaphragm. J. Appl. Physiol. 1976;41:369

113. Jones NL, Killian KJ Limitation of Exercise in Chronic Airway Obstruction In The Lung, R G Crystal, J B West et al (eds), New York Raven Press Ltd 1991, Ch23 196-206

114. National Heart, Lung and Blood Institute Workshop In Grassino A, Clanton T Respiratory muscle fatigue Seminars in Respiratory Medicine, 1991, 12(4) 305-321

115 Pardy RL, Rivington RN, Despas PJ, Macklem PT Inspiratory muscle training compared with physiotherapy in patients with chronic airflow limitations. Am Rev Respir

Dis 1981,123 421-425

116 Pardy RL, Rivington RN, Despas PJ, Macklem PT. The effects of inspiratory muscle training on exercise performance in chronic airflow limitation Am. Rev Respir Dis. 1981,123 426-433

117 Grassino A, Gross D, Macklem PT, Roussos C, Zagelbaum G. Inspiratory muscle fatigue as a factor limiting exercise. Bull Eur. Physiopath. Respire: 1979,15.105-111.

118 Bye PTP, Esau SA, Levy R, Shiner RJ, Macklem PT, Pardy RL. The effect of oxygen breathing on inspiratory muscle function during exercise in patients with chronic airflow limitation. Am Rev Respir Dis 1982,125 207 [Abst]

119 Levine S, Weiser P, Gillen J. Evaluation of a ventilatory muscle endurance training program in the rehabilitation of patients with chronic obstructive pulmonary disease. Am. Rev Respir. Dis. 1986,133:400-406

120 Gallagher CG, Younes MK Breathing pattern during and following exercise in patients with chronic obstructive lung disease, interstitial lung disease and cardiac disease and in normal subjects. Am Rev Respir Dis 1986,133 581-586.

121 Grassino A, Bellemare F, Laporta D. Diaphragm fatigue and the strategy of breathing in COPD. Chest 1984,855 515-545

122 Grimby G, Elgefois G, Oxhoj H Ventilatory levels and chest wall mechanics during exercise in obstructive lung disease. Scand J Respir Dis 1973,54 45-52

123 Tzelepis G, McCool FD, Leith DE, Hoppin FG Jr Increased lung volume limits endurance of inspiratory muscles. J Appl. Physiol. 1988,64 1796-1802

124 Farkas GA, Roussos C Acute diaphragmatic shortening in vitro mechanics and fatigue Am Rev Respir Dis 1984,130,434-438

125 Grassino AE, Goldman MD Respiratory muscle coordination. In Handbook of Physiology-The Respiratory System, Ch 27 1986

126 Rohrer F Der Zusammenhang der AtemKrafte und Ihre Abhanaligigkeit vom Dehnungszustand der Atmungsorgane Arch Gesamte Physiol 1916,165 419-444

127 Rahn H, Otis AB, Chadwich LE, Fenn WO The pressure-volume diagram of the thorax and lung Am J Physiol 1946,146 161-178.

128 Buytendijk HJ Oesophagusdruck en longelasticiteit. Disertation Universitat Groningen, 1943 **129**. Agostoni E, Rahn H. Abdominal and thoracic pressures at different lung volumes J Appl. Physiol 1960;15:1087-1092

130 Wade O Movements of thoracic cage and diaphragm in respiration J Physiol (Lond) 1954,124 193-212

131. Bergofsky E Relative contributions of the rib cage and the diaphragm to ventilation in man. J Appl Physiol 1964,19698-706

132. Agostoni E, Mognoni P, Torri G, Saracino F. Relation between changes on rib cage circumference and lung volume. J. Appl. Physiol. 1965;20:1179-1186

133 Konno K, Mead J. Measurement of the separate volume changes of rib cage and abdomen during breathing. J. Appl. Physiol. 1967,22(3) 407-422

134 Smith J, Loring S. Passive mechanical properties of the chest wall. In: Mead J,
Macklem PT (eds) Handbook of Physiology, Section 3, The Respiratory System Vol3, Part
2 Washington, DC. American Physiological Society, 1986,429-442

135 Grimby G, Goldman M, Mead J. Rib-cage and abdominal volume partitioning during exercise and induced hyperventilation. Scand. J. Respirat Diseases (Suppl.), 1971,774-7

136 Grimby G, Bunn J, Mead J Relative contribution of rib cage and abdomen to ventilation during exercise J Appl Physiol 1968,24(2) 159-166

137. Konno K, Mead J Static volume-pressure characteristics of the rib cage and abdomen J Appl Physiol 1968,24(4) 544-548

138. Mead J, Peterson N, Grimby G, Mead J. Pulmonary ventilation as measured from body surface movements. Science 1967,156 1383-1384.

139. Sackner JD, Nixon A, Davis B, Atkins N, Sackner M. Non-invasive measurement of ventilation during exercise using a respiratory inductive plethysmograph. Am Rev Respir Dis 1980, 122 867-871

140. Cohn MA, Watson H, Weisshaut R. Stott F, Sackner MA A transducer for noninvasive monitoring of respiration. In Stott FD, Raftery EB, Sleight P, Goulding L (eds), ISAM 1977 Proceedings of the Second International Symposium on Ambulatory Monitoring London Academic Press, 1978 119-128

141. Cohn MA, Rao ASV, Broudy M, Birch S, Watson HL, Atkins N, Davis B, Stott FD, Sackner MA. The respiratory inductive plethysmographila new non-invasive monitor of respiration. Bull Eur Physiopathol. Respire 1982,18,643-658

142 Chadha TS, Watson H, Birch S, Jenouri GA, Schneider AW, Cohn MA, Sackner MA Validation of respiratory inductive plethysmography using different calibration procedures. Am Rev Respir Dis 1982,125 644-649

143 Loveridge B, West P, Anthonisen NR, Kryger MH Single-position calibration of the respiratory inductance plethysmograph - J. Appl. Physiol : Respirat Environ Exercise Physiol 1983,55(3) 1031-1034

144 Sackner MA, Watson H, Belsito AS, Feinerman D, Suarez M, Gonzalez G, Bizousky F, Krieger B. Calibration of respiratory inductive plethysmograph during natural breathing J. Appl. Physiol. 1989;66(1):410-420

145 Grimby G, Goldman M, Mead J. Respiratory muscle action inferred from rib cage and abdominal V-P partitioning. J. Appl. Physiol. 1976;41(5):739-751

146 Goldman M, Grassino A, Mead J, Sears T Pressure-volume characteristics of the diaphragm during voluntary contraction. Fed. Proc. 1972;31:321(Abst)

147 Macklem PT, Gross D, Grassino A, Roussos C Partitioning of inspiratory pressure swings between diaphragmand intercostal/accessory muscles. J Appl. Physiol. Respirat Environ. Exercise Physiol. 1978;44(2):200-208

148 Macklem PT, Macklem DM, De Troyer A. A model of inspiratory muscle mechanics. J. Appl. Physiol. Respirat Environ Exercise Physiol. 1983,55(2):547-557

149 Loring SH, Mead J Action of the diaphragm on the rib cage inferred from a forcebalance analysis J Appl Physiol 1982,53 756-760

150 Schmidt, RF Fundamentals of Sensory Physiology, New York: Springer-Verlag, 1978 15-29

151 Boring EG A History of Experimental Psychology, 2nd Edition New York, Appleton-Century-Crofts, 1950

152 Fechner, GT Elements of Psychophysics, Leipzig, 1860, in, The Handbook of Perception. Volume II, Psychophysical Judgement and Measurement, Carterette C and Friedman MP, Academic Press, New York, 1974, preface

153 Stevens SS The Psychophysics of Sensory Function Sensory Communication, Contributions to the Symposium on Principles of Sensory Communication, MIT Press, 1959 1-33

154 Stevens SS The psychophysics of sensory function Am Sci 1960;48:226-253.

155. Stevens, SS Issues in Psychophysical Measurement, Psychological Review 1971;78,5:426-450.

156. Stevens SS, Galanter EH Ratio Scales and Category Scales for a Dozen Perceptual Continua 1957,54(6) 377-411

157 Stevens SS On the Theory of Scales of Measurement Science 1946,103,(2684) 677-680

158. Borg G J Physiol 1967,192.13-20

159. Mahler, DA Dyspnea New York, Futura Publishing Company Inc , 1990

160 Baird JC, Noma E Fundamentals of Scaling and Psychophysics New York, J Wiley & Sons, 1978

161 Stevens SS The measurement of loudness Journal of the Acoustical Society of America 1955,27 815-820

162. Aitken RCB Measurement of feelings using visual analogue scales Proc R Soc Med 1962,62 989-993

163. Price DD, McGrath PA, Rofii A. The validation of visual analogue scales as ratio scale measures for chronic and experimental pain. Pain. 1983,17:45-56

164 Gift AC Validation of a vertical visual analogue scale as a measure of clinical dyspnea Rehab Nurs 1989,14 323-324

165 Adams L, Chronos N, Lane R, Guz A The measurement of breathlessness induced in normal subjects validity of two scaling techniques. Clin. Sci. 1985;69:7-16

166. Muza SR, Silverman MT, Gilmore GC, Hellerstein HK, Kelsen SG Comparison of scales used to quantitate the sense of effort to breathe in patients with chronic obstructive pulmonary disease. Am Rev Respir Dis 1990,141 909-913

167. Borg G Psychophysical Bases of Perceived Exertion Med Sci Sports Exer 1982,14(5) 377-381

168. Borg GAV., Noble BJ Perceived Exertion In Exercise and Sports Sciences Review, J H Wilmore (ed.) New York Academic Press 1974 131-153

169. Mihevic, P.M. Sensory cues for perceived exertion. A review. Med. Sci. Sports. Exercise 1981,13 150-163.

170 Borg G Simple rating methods for estimation of perceived exertion. In Borg G (ed) Physical Work and Effort. Oxford. Permagon Press, 1977-39-47.

171 Borg G A category scale with ratio properties for intermodal and interindividal comparisons in Geissler H, Petzold P (eds) Psychophysical Judgment and the Process of Perception Amsterdam North-Holland 1982 25-34

172 Stevens J C and Mach J D Scales of apparent force J Exp Psychol 1959;58:405-413

173 Noble G, Borg G Jacobs, I Validation of a category-ratio perceived exertion scale blood and muscle lactates and fibre types. Int J Sports Med 1981,2 279-283

174 Zechman, FW Wiley RL Afferent inputs to breathing respiratory sensation. In Cherniack NS, Widdicombe JG (eds). Handbook of Physiology, Section 3. The Respiratory System. Bethesda American Physiological Society, 1986. 449-474.

175 Campbell EJM, Freedman S, Smith PS, Taylor ME. The ability of man to detect added elastic loads to breathing. Clin. Sci. 1961,20 223-231

176 Bennett ED, Jayson MIV, Rubenstein D, Campbell EJM The ability of man to detect non-elastic loads to breathing. Clin. Sci. 1962;23:155-162

177 Wiley RL, Zechman FW Perception of added airflow resistance in humans. Respir Physiol 1966,2 73-87

178 Killian KJ, Mahutte CK, Campbell EJM Effect of timing, flow, lung volume, and threshold pressures on resistive load detection J Appl Physiol 1980,49(6) 958-963

179 Zechman FW, Davenport PW Temporal differences in the detection of resistive and elastic loads to breathing Respir Physiol 1978,34 267-277

180 Zechman FW, Wiley RL, Davenport PW, Burki NK Load detection latencies and temporal patterns of added load sensation. Am Rev Respir Dis 1979,119 73-75

181 Gottfried SB, Altose MD, Kelsen SG, Fogarty CM, Cherniack NS The perception of changes in airflow resistance in normal subjects and patients with chronic airways obstruction. Chest 1978,73 286-288

182 Gottfried SB, Altose MD, Kelsen SG, Cherniack NS Perception of changes in airflow resistance in obstructive pulmonary disorder. Am Rev Respir Dis 1981,123 566-570

183 Guz A, Noble MIM, Widdicombe JG, Trenchard D, Mushin WW, Makey ARE The role of vagal and glossopharyngeal afferent nerves in respiratory sensation, control of breathing

and arterial pressure regulation in conscious man Clin Sci 1966,30 161-170

184 Newsom-Davis J Contribution of somatic receptors in the chest wall to detection of added inspiratory airway resistance. Clin. Sci. 1967;33:249-260

185 Noble MIM, Frankel HL, Else W, Guz A The ability of man to detect added resistive loads to breathing. Clin. Sci. 1971,41 285-287

186 Chaudhary BA, Burki NK Effects of airway anaesthesia on the ability to detect added inspiratory resistive loads. Clin. Sci. 1978,54 621-626

187 Chaudhary BA, Burki NK The effects of airway anaesthesia on detection of added inspiratory elastic loads Arri Rev Respir Dis 1980,122 635-639

188 Zechman FW, O'Neil R, Shannon R Effect of low cervical spinal cord lesions on detection of increased airflow resistance in man. Physiologist 1967,10 356

189 Eisele J, Trenchard D, Burki N, Guz A. The effect of chest wall block on respiratory sensation and control in man. Clin. Sci. 1968;35 23-33

190 Campbell EJM, Gandevia SC, Killian KJ, Mahutte CK, Rigg JRA Changes in the perception of inspiratory resistive loads during partial curarization J Physiol London 1980;309 93-100

191 Killian KJ, Mahutte CK, Campbell EJM Resistive load detection during passive ventilation. Clin. Sci. 1980,50:493-495

192 Killian KJ, Campbell EJM, Howell JBL The effect of increased ventilation on resistive load discrimination. Am Rev Respir Dis 1979,120 1233-1238

193 Noble MIM, Frankel HL, Else W, Guz A The ability of man to detect added resistive loads to breathing. Clin. Sci. 1971,41 285-287

194 West DWM, Ellis CG, Campbell EJM Ability of man to detect increases in his breathing J Appl Physiol 1975,39 372-376

195 Gandevia SC, Killian KJ, Campbell EJM The contribution of upper airway and inspiratory muscle mechanisms to the detection of pressure changes at the mouth in normal subjects. Clin Sci 1981,60 513-518

196 Bakers JH, Tenney SM The perception of some sensations associated with breathing Respir Physiol 1970,10 85-92

197 Stubbing DG Killian KJ, Campbell EJM The quantification of respiratory sensations by normal subjects Respir Physiol 1981,44 251-260

198 Wolkove N, Altose MD, Kelsen SG, Konapalli PG, Cherniack NS Perception of changes in breathing in normal human subjects J Appl Physiol 1981,50(1).78-83

199 Gliner JA, Folinsebe LJ, Horvath SM Accuracy and precision of matching inspired lung volume Percept Psychophys 1981,29 511-515

200 Halttunen PK The voluntary control in human breathing Acta Physiol. Scan. Suppl. 1974,419 1-45

201 Killian KJ, Campbell EJM Dyspnea In Roussos C, Macklem PT (eds). The Thorax, Part B. New York. Marcel Dekker, 1985-787-828

202 Killian KJ, Mahutte CK, Campbell EJM Magnitude scaling of externally added loads to breathing Am Rev Respir Dis 1981,123 12-15

203 Altose MD, Di Marco AF, Gottfried SB, Strohl KP. The sensation of respiratory muscle force. Am Rev. Respir. Dis. 1982, 126-807-811

204 Killian KJ, Bucens DD, Campbell EJM Effect of breathing patterns on the perceived magnitude of added loads to breathing J Appl Physiol Respirat Environ Exercise Physiol 1982,52(3) 578-584

205 Burdon JGW, Killian KJ, Campbell EJM Effect of ventilatory drive on the perceived magnitude of added loads to breathing J Appl Physiol Respirat Environ Exercise Physiol 1982,53(4) 901-907

206 Stubbing DG, Ramsdale EH, Killian KJ, Campbell EJM Psychophysics of inspiratory muscle force J Appl Physiol Respirat Environ Exercise Physiol 1983,54(5):1216-1221.

207 Revelette WR, Wiley RL Plasticity of the mechanism subserving inspiratory load perception J Appl Physiol 1987,62(5) 1901-1906

208 Tack M, Altose MD, Cherniack NS Effect of aging on respiratory sensations produced by elastic loads J Appl Physiol Respirat Environ Exercise Physiol 1981,50.844-850

209 Tack M, Altose MD, Cherniack NS Effect of aging on the perception of resistive ventilatory loads. Am Rev Respir Dis 1982,126 463-467

210 Ward ME Stubbing DG Effect of chronic lung disease on the perception of added inspiratory loads Am Rev Resr - Dis 1985,132 652-656.

211. Tack M, Altose MD, Cherniack NS Effects of aging on sensation of respiratory force and displacement J Appl. Physiol Respirat Environ Exercise Physiol 1983,55(5) 1433-1440

212 Gottfried SB, Redline S, Altose MD Respiratory sensation in chronic obstructive pulmonary disease Am Rev Respir Dis 1985,132 954-959

213. Gottfried SB, Leech I, DiMarco AF, Zaccardelli, Altose MD Sensation of respiratory force following low cervical cord transection J Appl Physiol Respirat Environ Exercise Physiol. 1984,57(4) 989-994

214 Gandevia SC, Killian KJ, Campbell EJM The effect of respiratory muscle fatigue on respiratory sensetion. Clin. Sci. 1981,60:463-466

215 Revelette WR, Zechman FW Jr, Parker DE, Wiley RL Effect of background loading on perception of inspiratory loads J Appl Physiol Respirat Environ Exercise Physiol 1984,56(2) 404-410

216. Burdon JGW, Killian KJ, Stubbing DG. Campbell EJM Effect of background loads on the perception of added loads to breathing J Appl Physiol Respirat Environ Exercise Physiol 1983,54 1222-1228

217. Killian KJ, Gandevia SC, Summers E, Campbell EJM Effect of increased lung volume on perception of breathlessness, effort, and tension J. Appl. Physiol. Respirat Environ. Exercise Physiol. 1984,57(3) 686-691.

218 Campbell EJM, Guz A Breathlessness In Hornbein TF (ed) Regulation of breathing, Part 2 New York Marcel Dekker, 1981 1181-1192

219 Mahler, DA, Faryniarz, Lentine T, Ward J, Olmstead EM, O'Connor GT. Measurement of breathlessness during exercise in asthmatics, predictor variables, reliability, and responsiveness. Am Rev Respir Dis 1991,144-39-44

220 Meakins J The Cause and Treatment of Dysphoea in Cardio-Vascular Disease The Br. Med J 1923,1043-1045

221. Jones NL Dyspnea in exercise Med Sci Sports Exerc 1984,16 14-19

222 Altose MD Assessment and Management of Breathlessness Chest 1985,88 77S-83S.

223 Altose MD Mechanisms of Dyspnea In Chronic Obstructive Pulmonary Disease, Montenegro HD (ed), Churchill Livingstone 1984 117-140 224 Burki NK Dyspnea Clin Chest Med. 1980,1 47-55

225 Killian KJ, Jones NL The use of exercise testing and other methods in the investigation of dyspnea. Clin Chest Med 1984,5 99-108

226 Killian KJ Assessment of dysphoea Eur Respir J 1988, 1.195-197.

227 Mahler DA, Weinberg DH, Wells CK, Feinstein AR The measurement of dyspnea: contents, interobserver agreement, and physiologic correlates of two new clinical indexes. Chest 1984,85 751-758

228 Medical Research Council Committee on research into chronic bronchitis. Instructions for use of the questionnaire on respiratory symptoms Devon, W J Holman, 1966

229 Dudley DL, Sitzman J, Rugg M Psychiatric aspects of patients with chronic obstructive pulmonary disease. In Thompson WL, Thompson TL II (eds). Advances in Psychosomatic Medicine (Vol 14). Basel, Switzerland. S Karger, 1985.64-77.

230 Mahler DA, Rosiello RA, Harver A, Lentine T, McGovern JF, Daubenspeck JA Comparison of clinical dyspnea ratings and psychophysical measurements of respiratory sensation in obstructive airway disease Am Rev Respir Dis 1987,135 1229-1233

231 Altose, MD Psychophysics-an approach to the study of respiratory sensation and the assessment of dyspnea (ed) Am Rev Respir Dis 1987,135 1227-1228

232 Wilson RC, Jones PW A comparison of the visual analogue scale and modified Borg scale for the measurement of dysphoea during exercise. Clin. Sci. 1989,76 277-282

233 El-Manshawi A, Killian KJ, Summers E, Jones NL Breathlessness during exercise with and without resistive loading J Appl Physiol 1986,61(3) 896-905

234 Adams L, Lane R, Shea SA, Cockcroft A, Guz A Breathlessness during different forms of ventilatory stimulation a study of mechanisms in normal subjects and respiratory patients. Clin Sci 1985;69:663-672

235 Adams L, Chronos N, Lane R, Guz A The measurement of breathlessness induced in normal subjects individual differences. Clin. Sci. 1986,70 131-140

236 Gambles SA, Stark RD Breathlessness during exercise and during rebreathing carbon dioxide J Physiol (Lond), 1981,316 29P

237 O'Neill PA, Morton PB. Sharman P, Percival G, Stark RD Validation of hypercaphia as a model of breathlessness. Clin. Sci. 1983,64 10P-11P

238. Castele RJ, Altose MD Chemical and non-chemical influences on respiratory sensation and control of breathing Fed Proc. 1983,42 742 [Abstr]

239. Swinburn CR, Wakefield JM, Jones PW Relationship between ventilation and breathlessness during exercise in chronic obstructive airways disease is not altered by prevention of hypoxaemia. Clin Sci 1984,67 515-519

240. Lane R, Cockcroft A, Adams L, Guz A. Arterial oxygen saturation and breathlessness in patients with chronic obstructive airways disease. Clin. Sci. 1987,72:693:698

241 Cockcroft A, Beaumont A, Adams L, Guz A Ventilatory and subjective effects of supplemental oxygen during exercise in COPD patients. Clin. Sci. 1984,66 7P

242 Winning AJ, Hamilton RD, Shea SA, Knott C, Guz A. The effect of airway anaesthesia on the control of breathing and the sensation of breathlessness in man. Clin. Sci 1985,68 215-225

243 Davis SF, McQuaid KR, Iber C Extreme dyspnea from unilateral pulmonary venous obstruction Am Rev Respir Dis. 1987,136 184-188

244 Berglund E, Furhoff AK, Loefstroem B, Oequist L A study of the effect of unilateral vagus nerve block in a dysphoeic patient. Scand J Respir Dis 1971,52 34-38

245 Kimoff RJ, Cheong TH, Cosio MG, Guerraty A, Levy RD Pulmonary denervation in humans effects on dyspnea and ventilatory pattern during exercise. Am Rev Respir Dis 1990;142 1034-1040

246 Killian KJ, Campbell EJM Dyspnea and exercise Ann Rev Physiol 1983,45465-479.

247. Leblanc P, Summers E, Inman MD, Jones NL, Campbell EJM, KJ Inspiratory muscles during exercise. a problem of supply and demand J Appl Physiol 1988;64(6) 2482-2489

248 Leblanc P, Bowie DM, Summers E, Jones NL, Killian KJ Breathlessness and exercise in patients with cardiorespiratory disease. Arn Rev Respir Dis 1986,133 21-25

249. Stark RD, Gambles SA, Chatterjee SS An exercise test to assess clinical dyspnea estimation of reproducibility and sensitivity Br J Dis Chest 1982,76 269-278

250 Mazza FG, Zuti RA, Smith RL, Altose MD Determinants of the sensation of breathlessness Am Rev Respir Dis 1983,127 234 [Abstr]

251 Stark RD, Gambles SA, Lewis JA Methods to assess breathlessness in healthy subjects a critical evaluation and application to analyze the acute effects of diazepam and promethazine on breathlessness induced by exercise or by exposure to raised levels of carbon dioxide. Clin. Sci. 1981,61:429-439

252 Silverman M, Barry J, Hellerstein H, Janos J, Kelsen S. Variability of the perceived sense of effort in breathing during exercise in patients with chronic obstructive pulmonary disease. Am Rev Respir Dis 1988,137 206-209

253 Muza SR, Silverman MT, Gilmore GC, Hellerstein HK, Kelsen SG Comparison of scales used to quantitate the sense of effort to breathe in patients with chronic obstructive pulmonary disease Am Rev Respir Dis 1990,141 909-913

254 Swinburn CR, Wakefield JM, Jones PW Relationship between ventilation and breathlessness during exercise in chronic obstructive airways disease is not altered by prevention of hypoxemia. Clin. Sci. 1984,67 515-519

255 Chonan T, Mulholland MB, Altose MD, Cherniack NS Effects of changes in level and pattern of breathing on the sensation of dyspnea J Appl Physiol 1990,69(4) 1290-1295

256 Jones NL Clinical exercise testing 3rd ed Toronto WB Saunders, 1986

257 Gallagher CG, Younes M. Breathing pattern during and after maximal exercise in patients with chronic obstructive lung disease, interstitial lung disease, and cardiac disease, and in normal subjects. Am Rev Respir Dis 1986,133 581-586

258 Harber Ph, SooHoo K, Tashkin DP Is the MVV FEV, Ratio Useful for Assessing Spirometry Validity? Chest 1985,88(1) 52-57

259 O'Neill PA, Stark RD, Allen SC, Stretton TB The relationship between breathlessness and ventilation during steady-state exercise Bull Eur Physiopath Respir 1986,22(3).247-250

260 Wilson RC, Jones PW Influence of prior ventilatory experience on the estimation of breathlessness during exercise Clin Sci 1990,78.149-153

261 Lane R, Adams L, Guz A Is low-level resistive loading during exercise perceived as breathlessness? Clin Sci 1987,73 627-634

262 Jones GL, Killian KJ, Summers E, Jones NL Inspiratory muscle forces and endurance in maximum resistive loading J Appl Physiol 1985,58 1608-1615.

263 Milic-Emili J, Mead J, Turner JM, Glauser EM Improved technique for estimating pleural pressure from esophageal balloons J Appl Physiol 1964a,21.207-211

264 Bradley TD, Chartrand DA, Fitting JW, Killian KJ, Grassino A The relation of inspiratory effort sensation to fatiguing patterns of the diaphragm Am. Rev Respir Dis 1986;134 1119-1124

265. Fitting JW, Chartrand DA, Bradley TD, Killian KJ, Grassino A Effect of thoracoabdominal breathing patterns on inspiratory effort sensation J Appl Physiol 1987,62(4) 1665-1670

266 Rahn H, Otis AB, Chadwick LE, Fenn WO The pressure-volume diagram of the thorax and lung Am J Physiol 1946,146 161-178

267 Agostoni E, Fenn WO Velocity of muscle shortening as a limiting factor in respiratory air flow J Appl Physiol 1960,16 349-353

268 Fry DL, Hyatt RE Pulmonary mechanics A unified analysis of the relationship between pressure, volume and gas flow in the lungs of normal and diseased human subjects Am J Med 1960,29 672-689

269 Hyatt RE, Flath RE Relationship of air flow to pressure during maximal respiratory effort in man J Appl Physiol 1966,21 477-482

270 Jones NL, Killian KJ. Limitation of exercise in chronic airway obstruction. In: The Lung, R.G. Crystal, J.B. West et al (Eds), New York. Raven Press Ltd., 1991.

271 Ward ME, Eidelman D, Stubbing DG, Bellemare F, Macklem PT Respiratory sensation and pattern of respiratory muscle activation during diaphragm fatigue J. Appl Physiol 1988,65(5) 2181-2189

272 Bellemare F, Grassino A Effect of pressure and timing of contraction o human diaphragm fatigue J Appl Physiol 1982,53 1190-1195

273 Garidevia SC, Killian KJ, Campbell EJM. The effect of respiratory muscle fatigue on respiratory sensations. Clin. Sci. 1981,60:463-466

274 Supinski GS, Clary SJ, Bark J, Kelsen SG Effect of inspiratory muscle fatigue on perception of effort during loaded breathing J Appl Physiol 1987,62(1) 300-307

275 Chonan T, Mulholland MB, Altose MD, Cherniack NS Effects of changes in level and pattern of breathing on the sensation of dyspnea J Appl Paysiol 1990,69(4) 1290-1295

276 Grassino A, Bellemare F, Laporta D. Diaphragm fatigue and the strategy of breathing in COPD. Chest 1984,85(Suppl) 51-54 277 Cooke NT, Wilson SH, Freedman S Blood lactate and respiratory muscle fatigue in patients with chronic airways obstruction. Thorax: 1983,38:184-187

278 Smidt U, Worth H, Petro W. Ventilatory limitation of exercise in patients with chronic bronchitis and/or emphysema. Bull Eur. Physiopathol. Respir. 1979,15.95-101

279 Wilson SH, Cooke NT, Moxham J, Spiro SG Sternomastoid muscle function and fatigue in normal subjects and in patients with chronic obstructive pulmonary disease. Am Rev Respir Dis 1984,129 460-464

280 Criner GJ, Celli BR Ventilatory muscle recruitment in exercise with O2 in obstructed patients with millid hypoxemia. J. Appl. Physiol. 1987;63:195-200

281 Bye PTP, Esau SA, Levy RD Ventilatory muscle function during exercise in air and oxygen in patients with chronic airflow limitation. Am Rev Respir Dis 1985,132 236-240

282 Effects of COPD on the Respiratory Muscles In The Lung, R G Crystal, J B West et al (eds), New York Raven Press Ltd , 1991

283 McFadden ER Jr., Kiser R, de Groot WJ. Acute bronchial asthma. Relations between clinical and physiologic manifestations. N. Eng. J. Med. 1973,288 221-225.

284 Rochester DF, Braun NMT The diaphragm and dyspnea Evidence from inhibiting diaphragmatic activity with respirators Am Rev Respir Dis 1979,119 77-80

285 Dodd DS, Brancatisano TP, Engel LA Effect of abdominal strapping on chest wall mechanics during exercise in patients with severe chronic air-flow obstruction. Am Rev Respir Dis 1985,131 816-821

286 Sharp JT, Drutz WS, Moisan T, Foster J, Machnach W. Postural relief of dyspnea in severe chronic obstructive pulmonary disease. Am Rev Respir Dis 1980,122:201-211

287 McCool FD, Mead J Dyspnea on immersion Mechanisms in patients with bilateral diaphragm paralysis. Am Rev Respir Dis 1989,139 275-276

288 Goldman HI, Becklake MR Respiratory function tests Normal values at median altitudes and the prediction of normal results Am Rev Tuberc 1959,79 457-467

289 Crapo RO, Morris AH, Gardner RM Reference spirometric values using techniques and equipment that meet ATS recommendations. Am Rev Respir Dis 1981,123 659-664

290 Buytendijk HJ Oesophagusdruck en longelasticiteit Disertation Universitat Groningen, 1949

291 Campbell EJM. The respiratory muscles and the mechanics of breathing Chicago Year Book Publishers, 1958

292. Sackner JD, Nixon AJ, Davis B, Atkins N, Sackner MA – Non-invasive Measurement of Ventilation During Exercise Using a Respiratory Inductive Plethysmograph – Am Rev Respir Dis 1980,122 867-871

293. Laporta D, Grassino A Assessment of transdiaphragmatic pressure in humans J Appl Physiol 1985,58(5) 1469-1476

294. Levison H, Cherniack RM Ventilatory cost of exercise in chronic obstructive pulmonary disease J Appl Physiol 1968,25 21-27

295 Agostoni E, Torri G. An analysis of chest wall motions at high values of ventilation Respir Physiol 1967,3 318-332

296 Grassino AE, Derenne JP, Almirall J, Milic-Emili J, Whitelaw W. Configuration of the chest wall and occlusion pressures in awake humans. J. Appl. Physiol. 1981,50:134-142.

297 Gilmartin JJ, Gibson GJ. Mechanisms of paradoxical rib cage motion in patients with Chronic Obstructive Pulmonary Disease. Am. Rev. Respir. Dis. 1986,134-683-687

298 Grassino AE, Lewinson GE, Tyler JM Effects of hyperinflation of the thorax on the mechanics of breathing J Appl Physiol 1973,35(3) 336-342

299. Henke KG, Sharratt M, Pegelow D, Dempsey JA Regulation of end-expiratory lung volume during exercise J Appl Physiol 1988,64 135-146

300 Kiers A, VanderMark TW, Woldring MG, Peset R Determination of the functional residual capacity during exercise Ergonomics 1980,23 955-959

301 Sackner JD, Nixon AJ, Davis B, Atkins N, Sackner MA Effects of breathing through external dead space on ventilation at rest and during exercise, II Am Rev Respir Dis 1980,122 933-940

302 Abbrecht PT, Rajagopal KR, Kyle RR Expiratory muscle recruitment during inspiratory flow-resistive loading and exercise Am Rev Respir Dis 1991,144 113-120

303 Pengelly LD, Alderson AM, Milic-Emili J Mechanics of the diaphragm J Appl Physiol 1971,30 797-805

304 Gilmartin JJ, Gibson GJ Abnormalities of chest wall motion in patients with chronic air-flow obstruction Thorax 1984,39 264-271

305 Delgado HR, Braun SR, Skatrud JB, Reddan WG, Pegelow DF Chest wall and abdominal motion during exercise in patients with Chronic Obstructive Pulmonary Disease. Am Rev Respir Dis 1982,126 200-205

306 Cavagna GA, Dusman B, Margaria R Positive work done by a previously stretched muscle J Appl Physiol 1968,24 21-32